KU Leuven Biomedical Sciences Group Faculty of Medicine Department of Oncology



A geriatric assessment-based approach for the older patient with cancer and the impact of ageing and cancer on well-being

Abdelbari Baitar

Jury:

Promotor:	Prof. Dr. Hans Wildiers
Co-promotors:	Dr. Dirk Schrijvers
	Prof. Dr. Frank Buntinx
Chair:	Prof. Dr. Maarten Naesens
Secretary:	Prof. Dr. Paul Clement
Jury members:	Prof. Dr. Jan De Lepeleire
	Prof. Dr. Paul Clement

Dissertation presented in partial fulfilment of the requirements for the degree of Doctor in Biomedical Sciences

December, 2017

Copyright ©2018 by Abdelbari Baitar

All rights reserved. No part of this publication may be reproduced in any form by print, photoprint, microfilm, or other means, without prior permission from the authors.

Alle rechten voorbehouden. Niets in deze uitgave mag worden openbaar gemaakt en/of verveelvoudigd door middel van druk, fotokopie, microfilm of op welke manier dan ook, zonder voorafgaande toestemming van de auteurs.

ISBN 9789090125398

Dankwoord

Mijn dank gaat naar Dirk Schrijvers die de weg vrij maakte voor dit werk, naar Hans Wildiers die de me de kans gaf om te werken aan een doctoraat en naar Frank Buntinx die me de mogelijkheid gaf om mee te werken aan de Klimop studie.

De afgelopen jaren heb ik gewerkt aan verschillende projecten die hebben geleid tot deze thesis. Hierbij kwam ik in contact met veel mensen die geconfronteerd werden met kanker. Zowel in het ziekenhuis als bij hen thuis tijdens interviews, mensen met verschillende type kankers en verschillende vooruitzichten. Hen ben ik enorm dankbaar om me in vertrouwen te nemen in een moeilijke periode in hun leven. Bij velen had ik het gevoel dat ze onze doelstellingen konden waarderen maar dat ze vooral nood hadden aan een goed gesprek.

Voor mijn werk in ZNA Middelheim ben ik mijn co-promotor Dirk Schrijvers, collega oncologen en Riet De Vogel dankbaar voor hun inzet bij de implementatie van geriatrische evaluaties op de afdeling oncologie. Mijn dank gaat uit naar Anita Boumans, waar ik dagdagelijks mee samen werk. Zij zorgt ervoor dat het steeds aangenaam werken is en zonder haar zou ik nog steeds verloren lopen in het ziekenhuis. Ook vermeld ik mijn andere collega's waaronder Jan, Micheline, Veerle, Titia, Samina, Hanne, Luc, Dianne en Corinne die steeds begaan waren met mij en mijn project.

Voor mijn werk dat uitging van UZ Leuven onder leiding van mijn promotor Hans Wildiers, ben ik dankbaar voor de goede samenwerking met Cindy Kenis. Ik denk dat we vanuit onze eigen ervaring elkaar goed konden aanvullen. Ik dank hierbij ook Jean Pierre Lobelle voor zijn ondersteuning.

Voor mijn werk onder leiding van mijn co-promotor Frank Buntinx dat uitging van het academisch centrum van huisartsgeneeskunde te Leuven, ben ik dankbaar voor de samenwerking met Marjan Van den Akker en Tine De Burghgraeve. Dit verliep altijd zeer vlot en aangenaam. Graag zou ik ook mijn collega's van vzw Farmaka en nog vele anderen uit mijn familie - en vriendenkring willen bedanken voor hun interesse en regelmatige consideratie.

Tenslotte, dit werk was enkel mogelijk dankzij de jarenlange en onvoorwaardelijke steun van mijn ouders Abderrachman Baitar en Malika Ghciel, en dankzij de steun van mijn broers en zussen Nordin, Redouane, Nasia, Assia, Mohammed en Rachid.

> December 2017, Abdelbari Baitar

Table of contents

Chapter 1. General introduction	1
The implementation of Geriatric Assessment (GA) in oncology The impact of aging and cancer on well-being: the Klimop study Aims and main research questions Overview of the dissertation	
Chapter 2	9
Evaluation of the G8 and GFI as screening tools for GA	
Chapter 3	25
The implementation of GA-based recommendations	
Chapter 4	43
The predictive value of G8 and GFI for severe treatment toxicity	
Chapter 5	55
The added prognostic value of geriatric screening and GA for overall survival	
Chapter 6	79
The added prognostic value of biomarkers and GA for overall survival	
Chapter 7.	107
The Klimop study: The utilization of formal - and informal care by older patients with cancer	
Chapter 8	127
The Klimop study: Coping shortly after a cancer diagnosis and subsequent well-being	
Chapter 9. General discussion	143
Discussion The implementation of Geriatric Assessment in oncology The impact of aging and cancer on well-being: the Klimop study Future perspectives	
Chapter 10. Summary (English & Nederlands)	159

Chapter 1. General introduction

The implementation of geriatric assessment in oncology

The time has arrived *now* that large numbers of older people are diagnosed with cancer who are in need of subsequent care and treatment. In a population of about 11 million inhabitants, 67 087 new diagnoses of invasive tumors (excl. non-melanoma skin cancer) were registered in Belgium in 2015 with 30122 (45%) of these patients being older than 70 years (1). The yearly number of patients with a new cancer diagnosis is expected to surpass 79 000 by 2025 with the highest incidence rates in the elderly. These numbers do not even consider older patients with a relapse of their cancer or with progressive disease who also present to the clinic for cancer treatment and care. Given this epidemiologic trend of cancer in older adults, healthcare providers need to be aware of this public health concern. In this dissertation, we focus on some clinical, societal, and psychosocial aspects of ageing and cancer.

Health care providers and policy makers should be aware that older patients with cancer require special attention regarding treatment decisions and care. Many of these patients present with complex clinical presentations for example in terms of comorbidity, functional status, and cognition reflecting differences in biological and chronological age. Oncologists and other specialists who care for patients with cancer face the difficulty in identifying optimal treatment options for older adults. The treating physician has to consider some difficult questions: Is the patient going to die of or with cancer?, Is the patient able to tolerate treatment?, Will any treatment improve quality of life? Is there enough social support during treatment? (2,3). This together with the lack of evidence-based data makes treatment of this population complex. The Comprehensive Geriatric Assessment (CGA), developed by geriatricians, is currently considered the most appropriate way to obtain a better view on the global health and reserve capacity of the older patient. It consists of a set of screening tools and includes the assessment of functional status, co-morbidity, nutritional status, cognition, and psychological status. Several guidelines have recommended a CGA-based approach in oncology to guide treatment decision-making (3-5). The term geriatric assessment (GA) is more and more used in oncology instead of CGA because studies on CGA in this discipline are less 'comprehensive' than in geriatrics (4). Both terms are used in this dissertation and a definition of the (C)GA is provided in each relevant chapter. We note that the GA domains that were assessed vary from chapter to chapter. This discrepancy is due to the different studies this work was based on (see below) and the lack of consensus on which geriatric domains to include in a GA. Furthermore, older patients were defined as older than 65 or older than 70 depending which chapter you read, again because of the different studies. It is somewhat contradictory to select a cut-off age for older patients while we are trying to assess their physiological reserve capacity or biological age. However, for the implementation of GA in routine practice, a selection of a cut-off age might be necessary.

Despite recommendations, the GA-based approach is currently not widely implemented in routine oncology practice, in part because of limited resources. There is a need for a short tool that can quickly identify patients who can benefit from a GA (7-8). Two geriatric screening tools will be validated in this dissertation in reference to a GA: the G8 questionnaire and the Groningen Frailty Indicator (GFI). Another topic concerning the implementation of GA in routine practice that will be studied in this dissertation involves geriatric recommendations that follow GA. To what extent are identified geriatric problems addressed in oncology? Which health care professionals or services are the most referred to and for which geriatric problems? Although important for the effectiveness of the GA-based approach, this step following GA has received little research attention.

A major issue confronting oncologists is to find the right balance between providing effective cancer treatment and minimizing treatment toxicity in older patients with cancer, given the increased risk for treatment-related toxicity in this population (9). There are no validated methods to identify older patients fit enough to receive the same treatment as younger patients. Various studies have looked at the predictive value of GA for treatment toxicity showing inconsistent results (10-12). In this dissertation, we will look at the predictive value of the two aforementioned geriatric screening tools, the GFI and G8, for treatment toxicity.

The GA-based approach is recommended to improve the estimation of treatment tolerance but also to improve the estimation of life-expectancy, which is sometimes a difficult task though very important in the treatment decision-making process. Several studies have shown that multiple GA components have some predictive value for overall survival. However, studies that evaluate the prognostic value of GA generally focus on the most prognostic individual GA components or on the best set of GA components and not on the GA as a whole (10-12). In this dissertation, we will quantify the added prognostic value of screening and GA beyond clinical information by comparing a baseline model of clinical information for overall survival with models extended with geriatric information. This analysis approach allows us to quantify and compare the added prognostic value of the GA as a whole (here the 10-item GA), the individual GA components, and screening tools. A recently developed GA summary score, the Leuven Oncogeriatric Screening Tool (LOFS), will be included in this analysis as well. In a separate chapter, we will also include the three commonly measured laboratory parameters hemoglobin, albumin, and Creactive protein and associated composite scores. Their prognostic value for overall survival has been shown in a range of tumor types (13-15). We will evaluate how much prognostic information these biomarkers add to the baseline model of clinical information and compare this with the added value of geriatric information.

Acknowledging the challenges that come with the growing population of older patients with cancer, several efforts have been made over the past years to improve the care of the older patient with cancer in Belgium. A cancer plan was launched in 2008; one of the initiatives in this plan was to provide support

for pilot projects in the field of clinical geriatric oncology. This led to the implementation of GA in numerous academic and non-academic hospitals between 2009 and 2012 and between 2012 and 2015. In this period, the 'Vlaamse Liga tegen Kanker' (VLK) also provided financial support for projects that aimed to improve care in this population. The knowledge gaps raised above will be addressed based on the aforementioned efforts.

The impact of ageing and cancer on well-being: the Klimop study

The second part of this dissertation will be based on the ongoing Klimop study (16). The primary aims of this longitudinal study are to assess the impact of cancer and ageing on subsequent well-being and to identify factors that predict well-being in older patients with cancer. For this purpose, older patients with cancer (OCP) are compared with two control groups: middle-aged patients with cancer (MCP) and older primary care patients without cancer (ONC). Two analyses are included in this dissertation, as part of this umbrella study.

In the first analysis, we will focus on home care. With the ageing population come societal changes such as declining family size, increased participation of women in the labor market, increased retirement age, and more single-living elderly. This evolution has led to an increased demand for formal care and changes in the balance between formal and informal care. The resulting changes in patterns of care are a challenge for the organization and financing of home care. Over the past decades, several studies have analyzed the use of home care considering different national contexts to prepare for these challenges (17-19). These studies focused on the general population and have informed policy decisions ever since. In this dissertation, we will focus on patients with a recent cancer diagnosis. A diagnosis of cancer and subsequent treatment can have a substantial effect on the well-being of patients and their need for home care. Therefore, we aim to gain a better insight on the utilization of formal and informal care shortly after a cancer diagnosis and one year later in OCP and the two control groups. We will also examine predictors of transitions towards formal care one year after a cancer diagnosis.

In the second analysis, we will focus on coping. The psychosocial care of the older patient with cancer has begun to receive more research attention, acknowledging differences between older patients and their younger counterparts. Although coping is not a stand-alone phenomenon, a better understanding of coping mechanisms used in cancer patients is important because unlike other concepts, like personality, coping is amenable to cognitive-behavioural intervention (20). The large majority of published studies that assess the relationship between coping and well-being are based on cross-sectional data. In this analysis, we will evaluate the predictive value of coping strategies shortly after a cancer diagnosis for subsequent well-being in OCP while disentangling ageing – and diagnosis effects.

Overview of the dissertation

This dissertation is divided in ten chapters. After a general introduction, chapter 2 until chapter 6 will cover several aspects concerning the implementation of geriatric screening and GA in oncology. Chapter 7 and chapter 8 are based on the Klimop study. Chapter 9 entails the general discussion. Chapter 10 provides a summary of the dissertation.

Chapter 2 describes the validation of two geriatric screening tools in reference to a GA. The effects of GA by itself are limited, unless followed by targeted geriatric recommendations and interventions. Chapter 3 describes the geriatric recommendations based on the results of the GA and the implementation of these recommendations in routine practice.

A major research topic in geriatric oncology is the value of geriatric screening and GA for outcome prediction. This could aid physicians in the treatment decision-making process. Chapter 4 describes the predictive value of two geriatric screening tools for severe treatment toxicity in patients who received one cycle of (radio)chemotherapy. Chapter 5 focuses on the added prognostic value of geriatric information beyond a basic prognostic model of clinical information for overall survival. Chapter 6 goes a step further by including three commonly measured laboratory parameters and associated composite scores in the analysis allowing a comparison with geriatric information.

Chapter 7 examines the utilization of formal and informal care shortly after a cancer diagnosis and after one year in OCP and the two control groups. Chapter 8 investigates if baseline coping strategies predict changes in different dimensions of well-being after one year in OCP while disentangling ageing - and cancer effects.

Aims and main research questions

The first part of this dissertation focuses on the GA-based approach in routine oncology practice. Our analysis was based on data from three implementation studies. Two studies implemented GA in oncology from October 2009 until July/December 2011. The third study implemented GA in routine practice between August 2011 and July 2012 and also focused on GA-based recommendations.

The second part of the dissertation focuses on the impact of ageing and cancer on well-being and was based on data from the ongoing Klimop study which started inclusion of patients from 2010. Additional details about each study can be found in each chapter and on page 163 in the 'personal contribution' section.

In this section, we provide an overview of the main research questions that each chapter aims to answer.

Chapter 2

There is a need for a short and accurate screening tool to discriminate fit patients and patients who can benefit from a CGA. Preferably, this screening tool can be used routinely in a busy practice regardless of tumor characteristics.

- What is the diagnostic accuracy of the G8 and the GFI in reference to a CGA and how to they compare?

Chapter 3

The implementation of GA in routine oncology practice should be followed by appropriate geriatric recommendations and interventions. This step following GA is important for the effectiveness of the GA-based approach but has received little research attention.

- What is the proportion of patients that receives geriatric recommendations based on GA results? Which geriatric recommendations and for which geriatric problems?
- How many of the geriatric recommendations are performed after one month? Which geriatric recommendations and for which geriatric problems?

Chapter 4

There are no validated methods to identify older patients fit enough to receive the same cancer treatment as younger patients. Predictive tools for severe treatment toxicity have the potential to minimize undertreatment and overtreatment.

- Do the G8 and the GFI predict severe treatment toxicity in patients who receive one cycle of (radio)chemotherapy?

Chapter 5

One of the aims of the GA-based approach is to improve the estimation of life-expectancy. Studies that evaluate the prognostic value of GA for mortality generally focus on the most prognostic individual GA components or on the best set of GA components. However, these studies do not focus on the GA as a whole or on the additional prognostic value of geriatric information beyond clinical information.

- Does GA improve the estimation of life-expectancy beyond clinical information? How much? What is the relative importance of the individual GA components and the GA as a whole?

Chapter 6

Several studies have shown that multiple routinely measured laboratory parameters and associated composite scores are independent prognostic factors for overall survival in patients with cancer. Very few studies have evaluated the prognostic value of both GA and biomarkers for overall survival.

- Do laboratory parameters improve the estimation of life expectancy beyond clinical information? How does this compare with the added prognostic value of the GA?
- Are laboratory parameters able to optimize the estimation of life expectancy beyond clinical information and information from the GA as a whole?

Chapter 7

The home care literature focuses on the general population and not specifically on patients with cancer. A diagnosis of cancer and subsequent treatment can have a substantial effect on the well-being of patients and their need for home care.

- Is the use of formal care and informal care shortly after a cancer diagnosis different in older patients with cancer compared with the two control groups (MCP and ONC)?
- How does formal care and informal care change after one year in older patients with cancer and the two control groups?
- Which factors predict a transition towards formal care one year after a cancer diagnosis?

Chapter 8

Much of what we know about coping is based on cross-sectional research in younger patients. More psychosocial research is needed that focuses on older patients.

- Do dispositional coping strategies shortly after a cancer diagnosis predict changes in well-being after one year in older patients with cancer?
- What is the impact of ageing and cancer on the relationship between baseline coping and subsequent well-being after one year?

References

1. Cancer Burden in Belgium via the Belgian Cancer Registry: <u>www.kankerregister.org</u>

2 Balducci L, Extermann M. Management of cancer in the older person: a practical approach. The Oncologist. 2000;5(3):224-237.

3. Wildiers H, Kenis C. Comprehensive geriatric assessment (CGA) in older oncological patients: Why and how? J Geriatr Oncol. 2012;3(2):174-176.

4. Wildiers H, Heeren P, Puts M, Topinkova E, Janssen-Heijnen ML, Extermann M, et al. International Society of Geriatric Oncology Consensus on Geriatric Assessment in older patients with cancer. J Clin Oncol 2014;32(24):2595-603.

5. Hurria A, Wildes T, Blair SL, Browner IS, Cohen HJ, Deshazo M, et al. Senior adult oncology, version 2.2014: clinical practice guidelines in oncology. J Natl Compr Cancer Netw 2014;12(1):82-126.

6. Pallis AG, Fortpied C, Wedding U, Van Nes MC, Penninckx B, Ring A, et al. EORTC elderly task force position paper: approach to the older cancer patient. Eur J Cancer 2010;46(9):1502–1513.

7. Hamaker ME, Jonker JM, de Rooij SE, Vos AG, Smorenburg CH, van Munster BC. Frailty screening methods for predicting outcome of a comprehensive geriatric assessment in elderly patients with cancer: a systematic review. Lancet Oncol. 2012;13(10):e437-44.

8. Decoster L, Van Puyvelde K, Mohile S, Wedding U, Basso U, Colloca G, et al. Screening tools for multidimensional health problems warranting a geriatric assessment in older cancer patients: an update on SIOG recommendations. Ann Oncol 2015;26(2):288-300.

9. Walko CM, Mcleod HL. Personalizing medicine in geriatric oncology. J Clin Oncol. 2014;32(24):2581-6

10. Puts MT, Santos B, Hardt J, et al. An update on a systematic review on the use of geriatric assessment for older adults in oncology. Ann Oncol. 2014;25(2):307-15.

11. Hamaker ME, Vos AG, Smorenburg CH, de Rooij SE, van Munster BC. The value of geriatric assessments in predicting treatment tolerance and all-cause mortality in older patients with cancer. Oncologist 2012;17(11):1439-49.

12. Versteeg KS, Konings IR, Lagaay AM, van de Loosdrecht AA, Verheul HM. Prediction of treatmentrelated toxicity and outcome with geriatric assessment in elderly patients with solid malignancies treated with chemotherapy: a systematic review. Ann Oncol. 2014;25(10):1914-8.

13. Caro JJ, Salas M, Ward A, Goss G. Anemia as an independent prognostic factor for survival in patients with cancer: a systematic, quantitative review. Cancer. 2001;91(12):2214-21.

14. Gupta D, Lis CG. Pretreatment serum albumin as a predictor of cancer survival: a systematic review of the epidemiological literature. Nutr J. 2010;9:69.

15. Roxburgh CS, McMillan DC. Role of systemic inflammatory response in predicting survival in patients with primary operable cancer. Future Oncol. 2010;6(1):149-63.

16. Deckx L, van Abbema D, Nelissen K, Daniels L, Stinissen P, Bulens P, et al. Study protocol of KLIMOP: a cohort study on the wellbeing of older cancer patients in Belgium and the Netherlands. BMC Public Health 2011;11:825.

17. Bolin K, Lindgren B, Lundborg P. Informal and formal care among single-living elderly in Europe. Health Econ. 2008;17(3):393-409.

18. Geerts J, Van den Bosch K. Transitions in formal and informal care utilization amongst older Europeans: The impact of national contexts. European Journal of Ageing 2012;9(1):27-37.

19. Suanet B, Broese Van Groenou M, Van Tilburg T. Informal and formal home-care use among older adults in Europe: can cross-national differences be explained by societal context and composition? Ageing and Society 2012;32(3):491-515.

20. Folkman S, Moskowits JT. Coping: pitfalls and promise. Annu Rev Psychol. 2004;55:745-74.

Chapter 2.

Evaluation of the Groningen Frailty Indicator and the G8 questionnaire as screening tools for frailty in older patients with cancer

Baitar A. Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D.

This chapter has been published in the Journal of Geriatric Oncology 2013;4(1):32-38.

Abstract

Objective

In this study, we evaluated the Groningen Frailty Indicator (GFI) and the G8 questionnaire as screening tools for Comprehensive Geriatric Assessment (CGA) in older cancer patients.

Patients and Methods

Eligible patients with various types and stages of cancer were evaluated for frailty before treatment. Patients were categorized as patients with a normal CGA and abnormal CGA (≥ 2 impaired tests). The diagnostic performance of the screening tools was evaluated against the CGA with Receiver Operating Characteristic analysis.

Results

In total, 170 patients (79 women) with median age 77 years old (range 66-97 years) were included. Sixty-four percent of patients had an abnormal CGA while according to the GFI (GFI \geq 4) and G8 questionnaire (G8 \leq 14) 47% and 76% of patients had an abnormal screening test. Overall, there was no significant difference (p=0.97) in diagnostic performance between the two screening tools. The Area Under the Curve was for both 0.87. For the GFI and G8 questionnaire the sensitivity was respectively 66% (95%CI: 56-75%), 92% (95%CI: 85-96%); the negative predictive value: 59% (95CI%: 49-69%), 78% (95%CI: 63-88%); and the specificity: 87% (95%CI: 76-94%), 52% (95%CI: 39-65%).

Conclusion

In this study, we showed that overall both the GFI and the G8 questionnaire were able to separate older cancer patients with a normal and abnormal CGA. For the G8 questionnaire, an adequate sensitivity and NPV were demonstrated, however at the expense of the specificity. For the GFI, we suggest lowering the threshold with one point to GFI \geq 3 to screen patients for a CGA.

Introduction

Cancer is a disease of the elderly and more than 50% of cancer patients are older than 65 years old.¹ The changing demographics will result in an even bigger number of elderly patients facing cancer and who are in need of subsequent care and treatment.

Older cancer patients have been underrepresented in cancer-treatment trials.² Because of this, there is a lack of scientific knowledge how to select the elderly cancer patient for different anticancer strategies. The Comprehensive Geriatric Assessment (CGA) is recommended by the International Society of Geriatric Oncology (SIOG) to improve the detection of problems and guide the oncologist in treatment decision-making.³ The CGA consists of a set of screening tools and includes assessment of functional status, physical performance, co-morbidity and medication use, nutritional status, social support, cognitive and psychological status. This assessment method adds information to Eastern Cooperative Oncology Group (ECOG) performance status⁴ and detects more older cancer patients as being unfit for chemotherapy than physicians' judgement⁵. However, its completion is time consuming for the patient and the health care professional and this limits its use in everyday practice.

A two-step approach has been proposed to integrate the CGA in oncology practice. First, patients are screened. Several short screening tools have been developed to quickly identify potentially frail patients.⁶ Patients who screen negative (normal test, considered 'fit') are likely to benefit most from standard treatment. Secondly, those patients who screen positive (abnormal test, considered 'unfit' or vulnerable) need a CGA linked to appropriate interventions. In order to follow a two-step approach, a screening tool should have a sufficient predictive ability to distinguish patients that may or may not benefit from a CGA. In this study, we compared the Groningen Frailty Indicator (GFI) and the G8 questionnaire as screening tools for frailty in older cancer patients in reference to the CGA.

Patients and methods

Study design and participants

This was a cross-sectional cohort study. Inclusion criteria were patients with all types and stages of cancer; age ≥ 65 years; an adequate understanding of the Dutch language; and ability to give informed consent. Exclusion criteria included severe known dementia; symptomatic brain metastases; and pre-existing major neurological or psychiatric problems.

Patients were simultaneously screened with the GFI and the G8 questionnaire followed by a CGA in Dutch, before treatment. Patients were seen and evaluated for frailty by a registered nurse or researcher in two non-academic general hospitals in Belgium (ZiekenhuisNetwerkAntwerpen Middelheim, Antwerp (central site) and Jessa hospital, Hasselt). When communication with the patient was difficult, a family member was allowed to help answering the questions. The study protocol was approved by the appropriate Ethical Committee and each patient provided written informed consent prior to study entry.

The Comprehensive Geriatric Assessment

There is no consensus about which tools to include in a CGA. The CGA in this study consisted of eight questionnaires that are widely used in the literature.^{7,8} For analysis purposes dichotomous outcomes were used for each questionnaire. For questionnaires that normally don't have a dichotomous outcome in their scoring system, i.e. for nutrition and cognition, we selected a cut-off value so that not only patients who were malnourished or with a severe cognitive impairment were identified but also patients who were at risk of malnourishment and with a mild cognitive impairment. See Table 1 for an overview of the selected cut-off values for the individual questionnaires that comprise the CGA. Consistent with previous studies, we defined patients at risk for frailty if more than one of the individual tests of the CGA scored above the cut-off score for impairment.⁹⁻¹¹ Patients were categorized as patients with a normal CGA (≤ 1 deficit out of eight tests within the CGA) and patients with an abnormal CGA (≥ 2 deficits in the CGA).

Function. Function was evaluated by the Katz Activity of Daily Living (ADL)¹² and Lawton-Brody Instrumental Activity of Daily Living (IADL)¹³. Dependence in one or more domains was defined as having an impaired test.⁷

Mobility/Falls. Various cut-off points are found in the literature for the Timed Up and Go test (TUG).¹⁴ A cut-off score of \geq 13.5 seconds has been shown to indicate a high risk of falls and was adopted in this study.¹⁵

Nutrition. Malnourished patients and patients at risk of malnutrition were identified with the full Mini Nutritional Assessment questionnaire (MNA).¹⁶

Co-morbidity. The Cumulative Illness Rating Scale for Geriatrics (CIRS-G) was used to assess comorbidity.^{17,18} Guidelines for scoring the CIRS-G are described by Miller and Towers.¹⁹

Cognition. The Folstein Mini-Mental State Examination (MMSE) is widely used.²⁰ Patients with a score <24 have an abnormal test.

Depression. For the 30-item Geriatric Depression Scale (GDS-30), we adopted a cut-off value of >10 in line with the original authors.²¹

Social support. Support from family and friends was measured by the Medical Outcomes Study Social Support Scale (MOS-SSS).²² The 'overall support index' was calculated by averaging the scores of the 19 items.

Geriatric domain	Measure	N° items	Score range	Cut-off value
Function	ADL	6	0-6	≤ 5
	IADL	8	0-8	≤ 7
Mobility	TUG	-	0-90	≥ 13.5 sec
Nutrition	MNA	18	0-30	≤ 23.5
Co-morbidity	CIRS-G	14	-	≥ 1 category at level 3/4 severity
Cognition	MMSE	5	0-30	< 24
Depression	GDS-30	30	0-30	> 10
Social support	MOS-SSS	19	0-5	< 4

Table 1. Comprehensive geriatric assessment measures

ADL: Activity of Daily Living; IADL: Instrumental Activity of Daily Living; TUG: Timed Up and Go test; MNA: Mini Nutritional Assessment; CIRS-G: Cumulative Illness Rating Scale for Geriatrics; MMSE: Mini-Mental State Examination; GDS-30: 30-item Geriatric Depression Scale; MOS-SSS: MOS Social Support Survey

The Groningen Frailty Indicator

The GFI was developed as a screening instrument for the level of frailty in geriatrics that specifically also included psycho-social components (Appendix 1).^{23,24} The GFI screens for loss of functions and resources in four domains: physical, cognitive, social, and psychological; and was found to be a one-dimensional concept. The original authors tested the GFI in a sample of 275 older people that included hospital inpatients, nursing home residents, and community-dwelling elderly. Scores range from zero (not frail) to fifteen (very frail). A score of GFI \geq 4 was regarded as moderately frail according to a panel of geriatric experts. We adopted this cut-off value in our study to define an abnormal screening test. This tool has already been used in various patient groups, oncogeriatric patients included.^{11,25-29}

The G8 questionnaire

The G8 questionnaire was recently developed as a screening tool for CGA for elderly patients in oncology. The G8 questionnaire (eight questions) is an easy-to use screening tool with a total score ranging from zero to seventeen (Appendix 2). This tool was developed based on the MNA and includes seven items and age. The G8 screening tool was evaluated in 364 cancer patients older than 70 years prior to chemotherapy.³⁰ Results suggested 14 as threshold, equivalent to 90% sensitivity, and 60% specificity. Based on these findings, we used in our study a G8 cut-off score of \leq 14 to define an abnormal screening test.

Statistical analysis

The ability of the GFI and the G8 questionnaire to differentiate patients with a normal and an abnormal CGA was evaluated. Receiver Operating Characteristic (ROC) curve analysis was used to evaluate the diagnostic performance of both screening tools against the CGA. The area under the curve (AUC) was calculated to reflect and to compare the predictive value of both screening tools to discriminate patients with a normal and an abnormal CGA. A statistical (nonparametric) comparison of both ROC curves was

carried out with the method of Delong et al.³¹ The sensitivity, specificity, Negative Predictive Value (NPV), and Positive Predictive Value (PPV) of the screening tests were determined. This was a pilot study without calculation of a specific sample size. ROC analysis was done with MedCalc for Windows, Version 12.2.1.0 (Medcalc software, Mariakerke, Belgium).

Results

Sample description

A total of 170 eligible cancer patients (79 women) were recruited at two sites in Belgium. From October 2009 to December 2011, the central site included 150 eligible patients. Only four other patients refused participation. The second site included 20 patients for this study.

Patient characteristics are summarized in Table 2. The median age of patients in the study was 77 years with a range of 66-97 years; 89% of patients were \geq 70 years old, and 9% of patients were \geq 85 years old. From 170 patients, 36% of the patients were hospitalized at the time of frailty assessment. Sixty percent had a newly diagnosed cancer while the other patients had recurrent or progressive disease. Metastatic disease was present in 58% of patients and the treatments received by the patients after frailty assessment are summarized in Table 3.

Table 2. Patient characteristics

Characteristic	Patients (%)
Gender	
Female	79 (46%)
Male	91 (54%)
Median Age (range)	77 (66-97)
Outpatients	109 (64%)
Newly diagnosed	102 (60%)
Cancer type	
Urological	49 (29%)
Digestive system	32 (19%)
Head and neck	28 (16%)
Breast	25 (15%)
Lung	18 (11%)
Other	18 (11%)
Metastatic disease	
Yes	98 (58%)
No	67 (39%)
Unknown	7 (4%)

Table 3. Treatment received

Treatment received	Treatment intent					
	<u>Palliative</u>	<u>Curative</u>	Total			
Chemotherapy	60	35	95 (56%)			
CRT	1	13	14 (8%)			
Radiotherapy	5	5	10 (6%)			
Surgery	0	5	5 (3%)			
Targeted therapy	6	0	6 (4%)			
Hormonal therapy	14	0	14 (8%)			
Other*	-	-	25 (15%)			

* Wait-and-see policy, patients received best supportive care, refused therapy, or were lost to follow up.

CGA results

All questionnaires were completed for each patient at inclusion. Sixty- one patients (36%) had a normal CGA. Two or three deficits in the CGA were detected in 55 patients (32%) and four or more deficits in 54 patients (32%). In total, 109 patients (64%) had an abnormal CGA. The percentage of patients that had an impairment on the individual CGA components according to the cut-off values described in Table 1 were as follows: ADL: 33%; IADL: 52%; mobility: 35%; nutrition: 53%; co-morbidity: 35%; cognition: 9%; depression: 24%; and social support: 20%. For mobility, 17 patients (10%) were not able to do the TUG because they were hospitalized and/or too weak and were given a predefined maximum score of 90 seconds.

Predictive performance of the GFI

The GFI screened 80 patients (47%) as positive (GFI \geq 4). The sensitivity was 66% (95% Confidence Interval [CI]: 56-75%) and the specificity 87% (95% CI: 76-94%) for identifying two or more deficits in the CGA. The PPV and the NPV were respectively 90% (95% CI: 82-95%) and 59% (95% CI: 49-69%). The group of patients with an abnormal CGA had a mean GFI score of 4.7 (Standard Deviation [SD] 2.4) and patients with a normal CGA had a mean score of 1.8 (SD 1.3).

The AUC for the GFI was 0.87 (95%CI: 0.81-0.92; Standard Error [SE]: 0.03). The optimal cut-off point to identify patients with an abnormal CGA was estimated at GFI score \geq 3. The sensitivity and the specificity for the different cut-off values are given in Table 4.

Predictive performance of the G8 questionnaire

Out of 170 patients, 129 patients (76%) had an abnormal screening test according to the G8 questionnaire. For cut-off value G8 score \leq 14, the sensitivity was 92% (95%CI: 85-96%) and the specificity 52% (95%CI: 39-65%). The PPV and NPV were respectively 78% (95%CI: 70-84%) and 78% (95%CI: 63-88%). The mean G8 score was 10.0 (SD 2.9) for patients with an abnormal CGA. Patients with a normal CGA had a mean G8 score of 14.1 (SD 2.0).

The G8 questionnaire had an AUC of 0.87 (95%CI: 0.81-0.92; SE 0.03) and the optimal cut-off value for identifying patients with an abnormal CGA was estimated at G8 score ≤ 12.5 . The results for sensitivity and specificity for different cut-off values are shown in Table 5.

Cut-off value	%pts	Sensitivity	Specificity
Cut-off ≥2	80%	96%	49%
Cut-off ≥3**	66%	87%	70%
Cut-off ≥4*	47%	66%	87%
Cut-off ≥5	25%	39%	98%

Table 4. The GFI: Sensitivity/specificity for different cut-off values

*Cut-off value used in previous studies and in this study. **Cut-off value that corresponded with the highest Youden index in

ROC analysis. Abbreviations: % pts: percentage of patients

Table 5. The G8 questionnaire: Sensitivity/specificity for different cut-off values

Cut-off value	%pts	Sensitivity	Specificity
Cut-off ≤11	48%	67%	85%
Cut-off ≤12	53%	74%	85%
Cut-off ≤12.5**	54%	76%	85%
Cut-off ≤13	65%	83%	67%
Cut-off ≤14*	76%	92%	52%

^{*}Cut-off value used in previous studies and in this study. **Cut-off value that corresponded with the highest Youden index in ROC analysis. Abbreviations: % pts: percentage of patients

Comparison predictive performance

The ROC curves of the GFI (AUC= 0.87) and the G8 questionnaire (AUC= 0.87) are shown in Figure 1. A pair wise comparison of the ROC curves of both screening tools did not give a significant difference (p=0.97) for the ability to differentiate patients with a normal CGA and patients with an abnormal CGA.

Figure 1. ROC curves of the GFI and the G8 questionnaire



ROC curve comparison for the GFI and the G8 questionnaire. AUC= 0.87 for both screening tools. The significance level of a pairwise comparison of the ROC curves was p=0.97.

Discussion

Frailty is a useful yet debatable concept.^{24,32} The core feature of frailty is increased vulnerability to stressors due to impairments in multiple, inter-related systems that lead to decline in homeostatic reserve and resiliency. The main consequence is an increased risk for multiple adverse health-related outcomes. Therefore, frailty is an important factor to be considered in the senior cancer patient when a treatment decision (e.g. chemotherapy or surgery) needs to be made. The integration of CGA in oncology practice could provide valuable guidance for the oncologist in this decision-making process by taking the global health status of the patient into account. To avoid costs and the unnecessary in-depth assessment of 'fit' patients, a two-step approach with screening is recommended. In this study, we showed that overall both the GFI and the G8 questionnaire (no significant difference, p= 0.97) were able to differentiate patients who may or may not benefit from a CGA.

As there is a need for a simple and accurate screening tool that can be used routinely in a busy practice regardless of tumour specific factors, we studied a heterogenic sample of cancer patients. A large proportion of our sample had advanced disease and a smaller proportion was hospitalized at the time of frailty assessment. All patients were assessed at a time when a treatment decision had to be made. Whether the information obtained by the CGA influenced the treatment decision could not be determined in this study.

In our sample, 64% had more than one impaired test in the CGA or had an 'abnormal CGA'. For nutrition, approximately half of the patients had an impaired test. Malnutrition is a common problem among cancer patients especially in patients with cancer of the digestive system and head and neck which together constituted 35% of our sample. For IADL, also approximately half of the patients had an impaired test, i.e. impairment in at least one domain. See Table 1 for an overview of the cut-off values for the individual questionnaires that comprise the CGA. It could be argued whether having one impaired IADL domain (for example doing the laundry) should be weighed equally as being malnourished. This question could be extended to all geriatric domains in a CGA. For the purpose of the study, i.e. to identify at least two impaired tests in the CGA, we believe that this is not necessary. It is also not clear what the best approach would be for such an analysis. Obviously when the results of a CGA are interpreted for decisions concerning treatment or interventions, the different domains are weighed differently. This 'third step' (e.g. interpretation and discussion of results with oncologist/geriatrician, multidisciplinary meeting) after the two-step approach of screening followed by CGA was not investigated in this study.

Two screening tools were evaluated in this study to screen for CGA in older cancer patients. For the evaluation of a screening test, the sensitivity and the NPV are considered the most important characteristics. The sensitivity (92% versus 66%) and the NPV (78% versus 59%) of the G8 questionnaire were superior to the GFI. This was at the expense of the specificity (52% versus 87%) of the G8 questionnaire leading to a high number of false positives. ROC analysis of our sample estimated

G8 score ≤ 12.5 as the optimal cut-off value for an abnormal screening test. This however resulted in a decrease of the NPV from 78% to 67%. ROC analysis for the GFI estimated GFI ≥ 3 as the optimal cut-off value with a sensitivity (87%) and NPV (75%) similar to that of the G8 questionnaire. It should be noted that while the G8 questionnaire was designed to screen for CGA in geriatric oncology, the GFI was not. The GFI was developed to screen for frailty in geriatrics where a score of ≥ 4 was regarded as 'moderately frail'. However, based on our results we suggest selecting cut-off value GFI ≥ 3 instead of GFI ≥ 4 when this tool is used in geriatric oncology to screen patients for CGA.

An adequate sensitivity and NPV are important characteristics for a screening tool. A too sensitive tool, however, might make the screening step in this CGA-based approach redundant by excluding only a small proportion of patients for CGA. The percentage of patients with a positive screening test are shown in Table 4, 5 and 6. The G8 questionnaire in our study, for example, would have excluded only 24% patients from a CGA. However, if one considers the time needed for a CGA (30 to 45 minutes) in contrast to the time needed for a screening test (approximately 5 minutes), then this two-step approach might be interesting in a setting where older cancer patients are screened systematically.

Similar studies validating the GFI or the G8 questionnaire are summarized in Table 6. Here, the results of other screening tools (VES-13 and TRST) used for comparison were added. We found two reports validating the GFI. Kellen et al.¹¹ validated the GFI in reference to a CGA and Kenis et al.²⁵ in reference to the 'geriatric profile' determined by consensus by a multidisciplinary team. In both studies the patient population consisted of patients with various types of cancer. Both groups reported a lower sensitivity (39% and 57%) and NPV (36% and 40%) than our findings (sensitivity: 66% and NPV 59%). Results for other cut-off values were not explored in these studies. As mentioned above we suggest lowering the threshold with one point for the GFI when this screening tool is used in this setting.

The G8 questionnaire has only recently been developed. More recently an abstract of the Oncodage study, a large prospective multicentre study validating the G8 screening tool, was published.³³ The study population consisted of patients with various types of cancer, approximately half with breast cancer. Corresponding with our findings, an adequate sensitivity and NPV was reported. An adequate reproducibility of the G8 screening tool was also demonstrated in the Oncodage study. This screening tool is also being used in several ongoing studies.

Research in this field is challenging. One major issue is the lack of a gold standard. Usually, frailty screening tools are compared against the CGA. However, there is no unique standardized CGA. The definition for an abnormal CGA, scales, and thresholds vary across studies. For example, the abnormal CGA definition used in the Oncodage study (≥ 1 impaired CGA test) would result, in this study, to 86% patients with an abnormal CGA.

	N° pts	% abnormal CGA	Screening tool	Screened positive (%)	Sens	Spec	NPV
Soubeyran et al. 2008 [abstract]	363	-	G8	-	90%	60%	-
Kenis et al. 2009	140	NA*	G8	-	80%	40%	35%
[abstract]			GFI	-	57%	87%	36%
[สมริกิสตร]			TRST**	-	92%	50%	63%
Kellen et al. 2010	113	68%***	GFI	31%	39%	86%	40%
			VES13	49%	61%	78%	48%
Soubeyran et al. 2011	1425	80%	G8	68%	77%	64%	-
Oncodago study			VES13	60%	69%	74%	-
Oncouage study			G8+VES13	-	87%	53%	-
[abstract]							
This study	170	64%	G8	76%	92%	52%	78%
			GFI	47%	66%	87%	59%
			GFI cut-off ≥3	66%	87%	70%	75%

Table 6. Overview: studies validating the GFI or G8

Results were rounded off; * No CGA was done, but a 'geriatric profile' was present in 79% of the patients as determined by consensus; ** Cut-off value 1; *** Patients had an abnormal CGA or were cognitively impaired. Abbreviations: Pts: Patients; NA: Not Applicable; Sens: sensitivity; Spec: Specificity; NPV: Negative Predictive Value; TRST: Triage Risk Screening Tool; VES13: Vulnerable Elders Survey

As more and more centers are gaining experience in integrating frailty assessment in oncology practice, research should also focus to the subsequent interventions. Prospective studies are also warranted to determine the predictive validity of frailty screening tools for treatment outcome e.g. severe chemotherapy toxicity.

Conclusion

In this study, we showed that overall both the GFI and the G8 questionnaire were able to separate older cancer patients with a normal and abnormal CGA. For the G8 questionnaire, an adequate sensitivity and NPV were demonstrated, however at the expense of the specificity. For the GFI, we suggest lowering the threshold with one point to GFI \geq 3 to screen patients for a CGA.

Author Contributions

Concepts and design: Dirk Schrijvers, Abdelbari Baitar.

Data collection: All authors.

Analysis and interpretation: Abdelbari Baitar, Dirk Schrijvers.

Manuscript preparation: Abdelbari Baitar.

Manuscript review: All authors.

References

1) Yanick R. Cancer burden in the aged: an epidemiologic and demographic overview. Cancer 1997; 80(7):1273-1283.

2) Hutchins LF, Unger JM, Crowley JJ, Coltman Jr CA, Albain KS. Underrepresentation of patients 65 years of age or older in cancer-treatment trials. N Engl J Med 1999; 341(27):2061-7.

3) Extermann M, Aapro M, Bernabei R, Cohen HJ, Droz JP, Lichtman S, et al. Use of comprehensive geriatric assessment in older cancer patients: recommendations from the task force on CGA of the International Society of Geriatric Oncology (SIOG). Crit Rev Oncol Hematol 2005; 55(3):241-52.

4) Repetto L, Fratino L, Audisio RA, Venturino A, Gianni W, Vercelli M, et al. Comprehensive geriatric assessment adds information to Eastern Cooperative Oncology Group performance status in elderly cancer patients: an Italian Group for Geriatric Oncology Study. J Clin Oncol 2002; 20(2):494-502.

5) Wedding U, Ködding D, Pientka L, Steinmetz HT, Schmitz S. Physicians' judgment and comprehensive geriatric assessment (CGA) select different patients as fit for chemotherapy. Crit Rev Oncol Hematol 2007; 64(1):1-9.

6) Extermann M. Basic assessment of the older cancer patient. Curr Treat Options Oncol 2011; 12(3):276-85. doi: 10.1007/s11864-011-0161-5.

7) Balducci L, Extermann M. Management of cancer in the older person: a practical approach. Oncologist 2000; 5(3):224-37.

8) Rodin MB, Mohile SG. A practical approach to geriatric assessment in oncology. J Clin Oncol 2007; 25(14):1936-44.

9) Mohile SG, Bylow K, Dale W, Dignam J, Martin K, Petrylak DP, et al. A pilot study of the Vulnerable Elders Suvey-13 compared with the comprehensive geriatric assessment for identifying disability in older patients with prostate cancer who receive androgen ablation. Cancer 2007; 109(4):802-10.

10) Molina-Garrido MJ, Guillen-Ponce C. Comparison of two frailty screening tools in older women with early breast cancer. Crit Rev Oncol Hematol 2011; 79(1):51-64. doi: 10.1016/j.critrevonc.2010.06.004.

11) Kellen E, Bulens P, Deckx L, et al. Identifying an accurate pre-screening tool in geriatric oncology. Crit Rev Oncol Hematol 2010; 75(3):243-8.

12) Katz S, Downs TD, Cash HR, Grotz RC. Progress in development of the index of ADL. Gerontologist 1970; 10(1):20-30.

13) Lawton MP, Brody EM. Assessment of older people: self-maintaining and instrumental activities of daily living. Gerontologist 1969; 9(3):179-86.

14) Podsiadlo D, Richardson S. The timed "Up & Go": a test of basic functional mobility for frail elderly persons. J Am Geriatr Soc 1991; 39(2):142-8.

15) Shumway-Cook A, Brauer S, Woollacott M. Predicting the probability for falls in community-dwelling older adults using the timed up & go test. Phys Ther 2000; 80(9):896-903.

16) Guigoz Y, Vellas B, Garry PJ. Assessing the nutritional status of the elderly: The Mini Nutritional Assessment as part of the geriatric evaluation. Nutr Rev 1996; 54(1 Pt 2):S59-65.

17) Linn BS, Linn MW, Gurel L. Cumulative illness rating scale. J Am Geriatr Soc 1968; 16(5):622-6.

18) Miller MD, Paradis CF, Houck PR, Mazumdar S, Stack JA, Rifai AH. Rating chronic medical illness burden in geropsychiatric practice and research: application of the Cumulative Illness Rating Scale. Psychiatry Res 1992; 41(3):237-48.

19) Miller MD, Towers A. A manual of guidelines for scoring the cumulative illness rating scale for geriatrics (CIRS-G). University of Pittsburgh 1991. Available at: http://catalogue.iugm.qc.ca/GEIDEFile/CIRSG.PDF?Archive=102955792013&File=CIRSG_PDF 20) Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. J Psychiatr Res 1975; 12(3):189-98.

21) Yesavage JA, Brink TL, Rose TL, Lum O, Huang V, Adey M, et al. Development and validation of a geriatric depression screening scale: a preliminary report. J Psychiatr Res 1982-1983; 17(1):37-49.

22) Sherbourne CD, Stewart AL. The MOS social support survey. Soc Sci Med 1991; 32(6):705-14.

23) Steverink N, Slaets JPJ, Schuurmans H, Van Lis M. Measuring Frailty. Development and testing of the Groningen Frailty Indicator (GFI). Gerontologist 2001; 41(1):236-237.

24) Schuurmans H, Steverink N, Lindenberg S, Frieswijk N, Slaets JP. Old or frail: what tells us more? J Gerontol A Biol Sci Med Sci 2004; 59(9):M965-5.

25) Kenis C, Schuermans H, Van Cutsem E, Verhoef G, Vansteenkiste J, Vergote I, et al. Screening for a geriatric risk profile in older cancer patients: a comparitive study of the predictive validity of three screening tools. Crit Rev Oncol 2009; 72: pp. S22-S22. doi:10.1016/S1040-8428(09)70046-0.

26) Andela RM, Dijkstra A, Slaets JP, Sanderman R. Prevalence of frailty on clinical wards: description and implications. Int J Nurs Pract 2010; 16(1):14-9.

27) Metzelthin SF, Daniëls R, van Rossum R, de Witte L, van den Heuvel WJ, Kempen GI. The psychometric properties of three self-report screening instruments for identifying frail older people in the community. BMC Public Health 2010; 10:176.

28) Aaldriks AA, Maartense E, le Cessie S, et al. Predictive value of geriatric assessment for patients older than 70 years, treated with chemotherapy. Crit Rev Oncol Hematol 2011; 79(2):205-12.

29) Pol RA, van Leeuwen BL, Visser L, Izaks GJ, van den Dungen JJ, Tielliu IF et al. Standardized Frailty Indicator as Predictor for Postoperative Delirium after Vascular Surgery: A Prospective Cohort Study. Eur J Vasc Endovasc Surg 2011; In press. doi 10.1016/j.ejvs.2011.07.006.

30) Soubeyran P, Bellera CA, Gregoire F, Blanc J, Ceccaldi J, Blanc-Bisson C, et al. Validation of a screening test for elderly patients in oncology. J Clin Oncol (Meeting Abstracts) May 2008 vol. 26 no. 15_suppl 20568.

31) DeLong ER, DeLong DM, Clarke-Pearson DL. Comparing the areas under two or more correlated receiver operating characteristic curves: a nonparametric approach. Biometrics 1988; 44(3):837-45.

32) Bergman H, Ferrucci L, Guralnik J, Hogan DB, Hummel S, Karunananthan S, et al. Frailty: an emerging research and clinical paradigm--issues and controversies. J Gerontol A Biol Sci Med Sci 2007; 62(7):731-7.

33) Soubeyran P, Bellera C, Goyard J, Heitz D, Cure H, Rousselot H, et al. Validation of the G8 screening tool in geriatric oncology: The ONCODAGE project. J Clin Oncol (Meeting Abstracts) May 2011 vol. 29 no. 15_suppl 9001.

Appendix A. The Groningen Frailty Indicator (GFI)

Are you able to carry out these tasks single-handed without any help? (The use of aids such as a walking stick, walking frame, wheelchair, is considered as independent)

- 1 Shopping,
- 2 Walking around outside (around the house or to the neighbors)
- 3 Dressing and undressing
- 4 Going to the toilet
- 5 What score do you give yourself for physical fitness? (scale 0 to 10)
- 6 Do you experience problems in daily life due to poor vision?
- 7 Do you experience problems in daily life due to poor hearing?
- 8 During the last 6 months (6 kg) have you lost a lot of weight unwillingly? (or 3 kg in 1 month)
- 9 Do you take 4 or more different types of medicine?
- 10 Do you have any complaints about your memory?
- 11 Do you sometimes experience an emptiness around you?
- 12 Do you sometimes miss people around you?
- 13 Do you sometimes have the feeling of being left alone?
- 14 Have you recently felt downhearted or sad?
- 15 Have you recently felt nervous or anxious?

Scoring: Questions 1-4: Independent=0; dependent=1 Question 5: 0-6=1; 7-10=0 Question 6-9: No=0; Yes=1 Question 10: No or sometimes=0; Yes=1 Questions 11-15: No=0; sometimes or yes=1

	The G8 questionnaire							
	Items	Possible answers	Score					
Α	Loss of appetite? Has food intake declined	0: severe anorexia						
	over the past 3 months due to loss of appetite,	1: moderate anorexia						
	digestive problems, chewing or swallowing	2: no anorexia						
	difficulties?							
B	Loss of weight during the last months	0: weight loss >3 kg						
		1: does not know						
		2: weight loss between 1 and 3 kg						
		3: no weight loss						
С	Mobility	0: bed or chair bound						
		1: able to get out bed/chair but not to go out						
		2: goes out						
Е	Neuropsychological problems	0: severe dementia or depression						
		1: moderate dementia or depression						
		2: no psychological problem						
F	Body Mass Index	0: BMI < 18.5						
		1: BMI between 18.5 and <21						
		2: BMI 21 to <23						
		3: BMI ≥23						
Н	Takes >3 prescription drugs per day	0: yes						
		1: no						
Р	In comparison with other people of the same	0: not as good						
	age, how do they consider their health status	0.5: does not know						
		1: as good						
		2: better						
	Age	0:>85						
		1: 80-85						
		2: <80						
	Total score	0-17						

Appendix B. The G8 questionnaire

г

Chapter 3.

Implementation of geriatric assessment-based recommendations in older patients with cancer: a multicentre prospective study

Baitar A*, Kenis C*, Moor R, Decoster L, Luce S, Bron D, Van Rijswijk R, Rasschaert M, Langenaeken C, Jerusalem G, Lobelle JP, Flamaing J, Milisen K, Wildiers H

This chapter has been published in the Journal of Geriatric Oncology 2015;6(5):401-10. * equal contribution

Abstract

Purpose

The main objective of this study was to describe geriatric recommendations based on geriatric assessment (GA) and to evaluate the implementation of these recommendations.

Patients and methods

A two-step approach of screening followed by GA was implemented in nine hospitals in Belgium. Patients \geq 70 years were included at diagnosis or at disease progression/relapse. Concrete geriatric recommendations were systematically documented and reported to the treating physicians and consisted of referrals to professional health care workers. Patient charts were reviewed after one month to verify which geriatric recommendations have been performed.

Results

From August 2011 to July 2012, 1550 patients were included for analysis. The median age was 77 (range: 70-97) and 57% were female. A solid tumour was diagnosed in 91.4% and a haematological malignancy in 8.6%. Geriatric screening with the G8 identified 63.6% of the patients for GA (n=986). A median of two geriatric recommendations (range: 1-6) were given for 76.2% (95%CI: 73.4-78.8) of the evaluable patients (n=710). A median of one geriatric recommendation (range: 1-5) was performed in 52.1% (95%CI: 48.4-55.8) of the evaluable patients (n=689). In general, 460 or 35.3% (95%CI: 32.8-38.0) of all the geriatric recommendations were performed. Geriatric recommendations most frequently consisted of referrals to the dietician (60.4%), social worker (40.3%), and psychologist (28.9%).

Conclusion

This implementation study provides insight into GA-based recommendations/interventions in daily oncology practice. Geriatric recommendations were given in about three fourths of patients. About one third of all geriatric recommendations were performed in approximately half of these patients.

Introduction

Cancer and its treatment might precipitate classic geriatric syndromes such as frailty, falls, malnutrition, delirium, or urinary incontinence [1]. It is therefore crucial to have better insights on the impact of 'non-cancer' related health issues when cancer treatment is considered. Comprehensive Geriatric Assessment (CGA), taking patients medical, functional, cognitive and social needs into account, is considered the most appropriate way to obtain a better view on the global health of the older patient. Several guidelines have recommended a CGA-based approach in oncology to guide treatment decision-making and to implement geriatric interventions [2-4].

In Belgium, a Cancer Plan was launched in 2008 in accordance with guidelines of the World Health Organization [5]. One of the initiatives in this plan was to provide support for pilot projects in the field of clinical geriatric oncology which led to the implementation of CGA in numerous academic and non-academic hospitals. Different models for implementation of CGA (e.g. geriatric oncology units) are possible, each with their advantages and disadvantages [2]. The best model of care cannot be determined due to lack of evidence and depends of local preferences and available resources. In the current implementation study, geriatric expertise was brought to both inpatients and outpatients who remained under the supervision of their treating physicians. This model was implemented from 2009 and resulted in various papers showing the feasibility and relevance of systematic geriatric screening followed by geriatric assessment in daily oncology practice [6-8].

From a practical point of view, the implementation of CGA can be divided into five consecutive steps: (i) identifying patients who can benefit from CGA; (ii) assessing these patients; (iii) developing recommendations; (iv) implementing the recommendations; and (v) provision of follow-up and adjustment of the care plan with repeated CGA. Most studies on the implementation of CGA in oncology have mainly focused on the first two steps. For this reason, the recent update of the consensus paper of the International Society of Geriatric Oncology (SIOG) preferred the use of the term Geriatric Assessment (GA) over Comprehensive Geriatric Assessment (CGA) in the approach of the older cancer patient [2]. However, the effects of GA by itself are limited, unless followed by targeted geriatric recommendations and interventions. Few studies have focused on these geriatric interventions in oncology, despite their importance for GA effectiveness. Therefore, this study aims to describe the geriatric recommendations based on the results of the GA and to evaluate the implementation of these recommendations.

Patients and methods

Patient population

Patients with all tumour types (except non-melanoma skin cancer) and haematologic malignancies were assessed in a uniform manner across six academic and three non-academic hospitals spread all over Belgium. Patients 70 years and older were included at diagnosis or at disease progression/relapse, when

a change in therapeutic strategy was considered. Both inpatients and outpatients were recruited from August 2011 until July 2012. This study cohort was preceded in the same hospitals by a GA implementation study between October 2009 and July 2011 that included only six tumour types and only focused on GA and not on GA-based recommendations [7]. The study was approved by the Ethical Committee of each participating hospital.

Geriatric screening and assessment

In each center, a trained health care worker (THCW) (i.e. a (para-)medical graduate attached to the oncology and/or geriatric department) was appointed to identify eligible patients and to perform a geriatric screening with the G8 (range:0-17) [6,9,10]. In patients with an abnormal G8 score (cut-off \leq 14), the geriatric screening was followed by a GA, also performed by a THCW. Patients were assessed for the following GA domains: functional status (FS), fall history, fatigue, cognition, depression, nutrition, polypharmacy, and comorbidities. A detailed description of the content of the GA can be found in the supplementary data section of ref 7. In contrast to the preceding study that included patients between October 2009 and July 2011, concrete geriatric recommendations based on GA were now systematically documented and reported to the treating physicians. Patient charts were reviewed after one month to verify which geriatric recommendations were actually performed. Informed consent was obtained from all patients or their caregiver.

Geriatric recommendations based on GA results

Depending on the hospital and its specific process of care, case-specific geriatric recommendations were made by the THCW based on standard protocols approved by the geriatric team, by the THCW in collaboration with geriatrician and/or treating physician and/or internal geriatric liaison, by a geriatrician or by a geriatric oncology team at multidisciplinary team meetings. Geriatric recommendations consisted of referrals to the geriatrician, geriatric liaison team, social worker, occupational therapist, physiotherapist, geronto-psychiatrist, psychologist, dietician, geriatric day clinic, fall clinic, another physician, and other. For each patient, geriatric recommendations were documented in two different ways. Firstly, the frequency of the different types of referrals was documented to get an overview of the different geriatric problems led to which geriatric recommendations, we documented for each patient which geriatric problems, at the level of the GA domains, led to which referrals. For example, a referral to a social worker for a particular patient might be recommended for problems concerning the GA domain 'social status' and/or the GA domain 'functional status'.

Statistical analysis

A descriptive analysis (frequencies, median, and percentages) was performed using SPSS 17.0 software (Chicago, IL). Percentages were associated with a 95% confidence interval calculated in accordance with Wilson's method where appropriate [11].

Results

Patient population

From a total of 1672 patients that were approached, 88 patients refused participation and 34 patients were excluded from the study for not meeting the inclusion criteria. The data of 1550 patients were used for analysis (see Figure 1).

The median age was 77 years (range: 70-97) and 57% were female. A solid tumour was diagnosed in 91.4% of patients and 8.6% of patients were diagnosed with a haematological malignancy. The most frequent diagnoses were cancer of the digestive system (31.4%), breast (26.1%), and genitourinary sites (12.7%). At the moment of evaluation, 69.5% had a newly diagnosed cancer, whereas 30.5% had disease progression or relapse. The most common comorbidities were peripheral vascular disease (21.2%), diabetes without complications (14.7%), congestive heart failure (13.5%), ulcer disease (12.9%), secondary malignancy (12.5%), and chronic pulmonary disease (11.4%). Details on the Charlson comorbidity index and other patient characteristics are summarized in Table 1.

Geriatric screening and assessment

The G8 identified 986 patients (63.6%) with a geriatric risk profile (G8 \leq 14) who could benefit from a full GA. A full GA was performed in 979 patients. Geriatric recommendations data based on GA were available for 932 patients (see Figure 1). Patient characteristics and GA results for this patient population are separately summarized in Table 1 and Table 2, respectively. For FS, respectively 57.7% and 63.5% of the patients had impairments for activities of daily living (ADL) and instrumental ADL (IADL). At least one fall incident during the past year was reported by 38.3% of the patients. Fatigue was experienced by 64.4% of the patients; mild/severe cognitive decline was detected in 17.5% of the patients; and 59.6% of the patients were at risk for depression. Screening for nutritional status identified 77.6% of the patients, of which 17.6% turned out to have a normal nutritional status and 69.0% and 13.4% respectively were at risk for malnutrition and malnourished.

Figure 1. Patient Flowchart



Table 1. Patient and clinical characteristics

	Total of patients included		Patients with recommendati baseline av	n geriatric ons data at vailable	Patients with da implementation recommendations a	ata on the of geriatric after 1 month
	N° ptc	0/	N° ptc	%	availab N° ptc	%
Total N° nts	1550	100	932	100	689	100
Age	1550	100	552	100	005	100
Median (range)	77 (7	0-97)	79 (70-	.97)	79 (70-9	(7)
Gender		,				-)
Female	884	57,0	519	55,7	390	56,6
Male	666	43,0	413	44,3	299	43,4
Tumour type		-,-		, -		- /
Carcinoma	1416	91,4	841	90,2	627	91,0
Digestive system	487	31,4	335	35,9	252	36,6
Breast	405	26,1	187	20,1	134	19,4
Genitourinary sites	197	12,7	107	11,5	69	10,0
Gynecologic sites	116	7,5	72	7,7	56	8,1
Thorax	95	6,1	67	7,2	51	7,4
Skin	44	2,8	22	2,4	19	2,8
Head and neck	36	2,3	25	2,7	24	3,5
Musculoskeletal sites	25	1,6	19	2,0	16	2,3
CUP	7	0,5	6	0,6	5	0,7
CNS (brain and spinal cord)	4	0,3	1	0,1	1	0,1
Haematological malignancies	134	8,6	91	9,8	62	9,0
Time point of assessment						
New diagnosis	1078	69,5	633	67,9	463	67,2
At progression	472	30,5	299	32,1	226	32,8
Carcinoma						
Stage	n= 1	.349	n=81	1	n=609)
I	183	13,6	83	10,2	53	8,7
Ш	284	21,1	144	17,8	96	15,8
III	267	19,8	157	19,4	120	19,7
IV	617	45,7	427	52,7	340	55,8
Treatment received	n=1	416	n=84	1	n= 627	7
Surgery	693	48,9	363	43,2	252	40,2
Chemotherapy	638	45,1	405	48,2	306	48,8
Radiotherapy	443	31,3	207	24,6	158	25,2
Hormonal therapy	344	24,3	147	17,5	106	16,9
Haematological malignancies						
Setting	n=:	L34	n=92	1	n=62	_
Curative	54	40,3	34	37,4	32	51,6
Palliative	80	59,7	57	62,6	39	62,9
Treatment received	_ n=:	134	n=93	1	n=62	
Surgery	7	5,2	3	3,3	1	1,6
Chemotherapy	104	//,6	68	/4,/	48	//,4
Radiotherapy	48	35,8	9	9,9	/	11,3
Comorbidity	n=1	547	n=93	1	n=688	aa -
	563	36,4	297	31,9	225	32,7
	372	24,0	223	24,0	167	24,3
ULI 22	612	39,6	411	44,1	296	43,0
	n=1	51/	n=93	1	n=689	44 7
	1/3	51,0	413	44,4	287	41,/
VAS 4 10	202	1/,3	200	18,0	107 107	15,5
VAS 4-10	482	31,8 F1C	350	37,6	295	42,8
	n=1	DTC	n=91	42.2	n=6/9	42.4
n>5	7/5	780 780	300 570	42,3 57 7	200	42,1 57 0
iicJ	/41	40,9	529	57,7	222	51,9

Legend: N° pts: Number of patients; CUP: Cancer of Unknown Primary origin; CNS: Central Nervous System; CCI: Charlson Comorbidity

Index; ECOG-PS: Eastern Cooperative Oncology Group - Performance Status; VAS: Visual Analogue Scale.

Table 2. GA results

		Patients geriatric baseline a	ts with Patients with data on the c data at implementation of geriatric available recommendations after 1 month available		
		N° nts	%	N° nts	%
			100	680	100
Demographic data Living situation		322	100	665	100
Home: alone		314	33,7	264	38,3
Home: with partner		514	55,2	357	51,8
Home: with family mem	iher	57	6.1	42	6.1
Service flat		14	1.5	5	0.7
Institution		- 25	2.7	14	2.0
Other		8	0.9	7	1.0
Marital status		C	0,0	•	±,•
Single		47	5 0	37	54
Marriad		507	54.4	258	5,4 52 0
Divorced		37	بب 1 0	330 27	32,0
Midow or		221	4,0 25 5	27	3,3 28 7
Widow-er		331 7	33,5 0 0	205	30,2 0.2
Legally conaditing		/	0,0	2	0,3
Other		3	0,3	Z	0.3
		00	0.4	C 2	0.1
Primary education	at	88	9,4 40 7	03	9,1 42 F
Lower secondary educat		398	42,7	293	42,5
Higher secondary educa	tion	276	29,0	217	31,5
Higher education		88	9,4	63	9,1
University education		62	6,7	42	6,1
Other		20	2,1	10	1,5
Functional status	· · ·	~		~	
ADL (6-24)	Independent	394	42,3	257	37,3
	Dependent (≥7)	538	57,7	432	62,7
IADL	Independent	340	36,5	218	31,6
(0-5/8 for male/female)	Dependent (<5 or <8)	592	63,5	471	68,4
Falls		n=93	30	n=	-687
	No falls	574	61,7	412	60,0
	Fall present	356	38,3	275	40,0
Fatigue		n=92	25	n=	-686
MOB-T (0-6)	No fatigue	329	35,6	230	33,53
	Presence of fatigue (<6)	596	64,4	456	66,47
Cognition		n=90	06	n=	-671
MMSE (0-30)	Normal cognition (≥24)	747	82,5	529	78,8
	Mild decline (18-23)	118	13,0	104	15,5
	Severe decline (≤17)	41	4,5	38	5,7
Depression		n=91	19	n=	-682
GDS-4 (0-4)	Normal At risk for depression	371	40,4	255	37,4
	(≥1)	548	59 <i>,</i> 6	427	62,6
Nutrition		n=9:	29	n=	-687
MNA-SF (0-14)	Normal At risk for malnutrition	208	22,4	118	17,2
	(≤11)	721	77.6	569	82.8
	· · ·	n=7	17	n=	-567
Full MNA (0-30)**	Normal	126	17,6	69	12,2
	At risk for malnutrition				
	(17 - ≤23,5)	495	69,0	415	73,2
1	Malnourished (<17)	96	13,4	83	14,6

Legend: ADL: Activities of Daily Living; IADL: Instrumental IADL; MOB-T: Mobility-Tiredness Test; MMSE: Mini Mental State Examination; GDS: Geriatric Depression Scale; MNA(-SF): Mini Nutritional Assessment (-Short Form)

*Defined as the presence of at least 1 item scored as 'tired' on the MOB-T; **A full MNA was conducted in patients with a positive MNA-SF
Geriatric recommendations based on GA results

Frequencies

Concrete geriatric recommendations were given for 710 patients (76.2%, 95% CI: 73.3-78.8%) from the 932 patients with available geriatric recommendations data. A median of two different geriatric recommendations (range: 1-6) were given for each patient. At chart review after one month, 689 patients (see Figure 1) with a total of 1302 given geriatric recommendations were evaluable to determine whether or not geriatric recommendations were performed. Patient characteristics and GA results for this specific patient population are also separately summarized in Table 1 and Table 2, respectively. In Table 3, the numbers of patients are shown according to the number of different (performed) geriatric recommendations per patient. The performed geriatric recommendations can be viewed stratified to the number of geriatric recommendations per patient at baseline. At least one geriatric recommendation was performed in 359 patients (52.1%, 95% CI: 48.4-55.8%) with a median of one performed recommendation (range: 1-5) per patient. A total of 460 different geriatric recommendations (35.3%, 95% CI: 32.8-38.0%) were performed after one month.

Referrals to implement geriatric recommendations

Most patients were referred to the dietician (60.4%), social worker (40.3%), psychologist (28.9%), or geriatric day clinic (16.1%). Referrals to other healthcare professionals or services were recommended in less than 10% of the patients.

The following referrals were the least performed when recommended: geronto-psychiatrist (0%), fall clinic (10.3%), geriatric day clinic (10.8%), physiotherapist (22.5%), another physician (22.6%), and psychologist (25.1%).

More details on referrals are provided in Table 4.

Geriatric recommendations on geriatric domain level

Problems in a specific GA domain led to referrals to various disciplines. An overview of all (performed) geriatric recommendations at the level of the GA domain is presented in Table 5. Problems in one GA domain always corresponded with one referral. Most referrals were recommended to address problems concerning nutrition (66.2%), social status (34.8%), FS (31.8%), or depression (31.6%).

Referrals for the following GA domains were the least performed when recommended: cognition (19.1%), falls (19.8%), and depression (25.7%).

Geriatric			N°pts with data on the implementation	N° pts with at least one		Performed geriatric		
per patient	N° pts	% ^a	recommendations	recommendation	%	per patient	N° pts	% ^b
0	222	23,8						
1	325	34,9	313	146	46,6	1	146	46,6
						0	167	53,4
2	212	22,7	208	110	52,9	2	29	13,9
						1	81	38,9
						0	98	47,1
3	118	12,7	114	75	65,8	3	9	7,9
						2	24	21,1
						1	42	36,8
						0	39	34,2
4	41	4,4	40	20	50,0	4	1	2,5
						3	3	7,5
						2	7	17,5
						1	9	22,5
						0	20	50,0
5	13	1,4	13	7	53,8	5	1	7,7
						4	2	15,4
						3	1	7,7
						2	2	15,4
						1	1	7,7
						0	6	46,2
6	1	0,1	1	1	100,0	6-2	0	0,0
						1	1	100,0
						0	0	0,0
Total	932		689	359				

Table 3. Frequency of geriatric recommendations based on GA results

^a calculated from total N° pts with geriatric recommendations data (n=932 pts)

 $^{\rm b}$ calculated from the 'N° pts with data on implementation of geriatric recommendations'

N° pts with recommended				N° pts with performed			
Referral	referral*	%	95%CI	referral*	%	95%CI	Ratio (%)
Dietician	416	60,4	56.7-64.0	180	26,1	23.0-29.5	43,3
Social worker	278	40,3	36.7-44.1	127	18,4	15.7-21.5	45,7
Psychologist	199	28,9	25.6-32.4	50	7,3	5.5-9.4	25,1
Geriatric day clinic	111	16,1	13.6-19.0	12	1,7	1.0-3.0	10,8
Other physician	62	9,0	7.1-11.4	14	2,0	1.2-3.4	22,6
Geriatrician	50	7,3	5.5-9.4	22	3,2	2.1-4.8	44,0
Other	44	6,4	4.8-8.5	16	2,3	1.4-3.7	36,4
Geriatric Liaison team	42	6,1	4.5-8.1	19	2,8	1.8-4.3	45,2
Physiotherapist	40	5,8	4.3-7.8	9	1,3	0.7-2.5	22,5
Fall clinic	29	4,2	2.9-6.0	3	0,4	0.1-1.3	10,3
Occupational therapist	25	3,6	2.5-5.3	8	1,2	0.6-2.3	32,0
Geronto-psychiatrist	6	0,9	0.4-1.9	0	0,0	0.0-0.6	0,0

 Table 4. Recommended and performed referrals to implement geriatric recommendations

*Data were calculated from the total N° pts with data on implementation of geriatric recommendation at chart review after one month (n=689 pts)

		GA domain							
		Social status	FS	Falls	Fatigue	Cognition	Depression	Nutrition	Other
		% (95%CI)	% (95%CI)	% (95%CI)	% (95%CI)	% (95%CI)	% (95%CI)	% (95%CI)	% (95%CI)
Dietician	Recommendation given							60.5 (56.8-64.1)	
	Recommendation performed							26.1 (23.0-29.5)	
Cocial worker	Recommendation given	31.3 (28.0-34.9)	17.1 (14.5-20.1)	0.3 (0.1-1.1)		1.2 (0.6-2.3)			0.1 (0.0-0.8)
Social worker	Recommendation performed	15.1 (12.6-18.0)	6.8 (5.2-9.0)	0.0 (0.0-0.6)		0.6 (0.2-1.5)			0.0 (0.0-0.6)
Develo a la sist	Recommendation given	0.6 (0.2-1.5)				1.2 (0.6-2.3)	27.1 (24.0-30.6)		
Psychologist	Recommendation performed	0.1 (0.0-0.8)				0.4 (0.1-1.3)	7.0 (5.3-9.1)		
Coriotria dav aliaia	Recommendation given	1.3 (0.7-2.5)	5.4 (3.9-7.3)	2.5 (1.5-3.9)	1.7 (1.0-3.0)	12.2 (10.0-14.8)	1.5 (0.8-2.7)	1.3 (0.7-2.5)	
Genatric day clinic	Recommendation performed	0.4 (0.1-1.3)	0.7 (0.3-1.7)	0.4 (0.1-1.3)	0.6 (0.2-1.5)	1.5 (0.8-2.7)	0.3 (0.1-1.1)	0.4 (0.1-1.3)	
Othernebussien	Recommendation given	0.4 (0.1-1.3)	0.7 (0.3-1.7)	1.0 (0.5-2.1)	5.1 (3.7-7.0)	0.9 (0.4-1.9)	0.6 (0.2-1.5)	0.4 (0.1-1.3)	1.6 (0.9-2.8)
Other physician	Recommendation performed	0.3 (0.1-1.1)	0.3 (0.1-1.1)	0.3 (0.1-1.1)	1.0 (0.5-2.1)	0.4 (0.1-1.3)	0.1 (0.0-0.8)	0.3 (0.1-1.1)	0.6 (0.2-1.5)
Conistrision	Recommendation given		1.6 (0.9-2.8)	1.2 (0.6-2.3)	2.0 (1.2-3.4)	2.9 (1.9-4.4)	0.7 (0.3-1.7)	1.6 (0.9-2.8)	1.0 (0.5-2.1)
Genatrician	Recommendation performed		0.3 (0.1-1.1)	0.1 (0.0-0.8)	0.3 (0.1-1.1)	0.7 (0.3-1.7)	0.1 (0.0-0.8)	1.0 (0.5-2.1)	0.4 (0.1-1.3)
Geriatric liaison	Recommendation given	0.9 (0.4-1.9)	1.9 (1.1-3.2)	1.2 (0.6-2.3)	1.3 (0.7-2.5)	1.5 (0.8-2.7)	0.6 (0.2-1.5)	1.7 (1.0-3.0)	0.9 (0.4-1.9)
team	Recommendation performed	0.4 (0.1-1.3)	0.7 (0.3-1.7)	0.6 (0.2-1.5)	0.7 (0.3-1.7)	0.3 (0.1-1.1)	0.3 (0.1-1.1)	0.9 (0.4-1.9)	0.3 (0.1-1.1)
Dhuaiath ana niat	Recommendation given		3.3 (2.2-5.0)	3.2 (2.1-4.8)	0.3 (0.1-1.1)				
Physiotherapist	Recommendation performed		1.0 (0.5-2.1)	0.4 (0.1-1.3)	0.0 (0.0-0.6)				
E all'all'alla	Recommendation given			4.2 (2.9-6.0)					
Fall clinic	Recommendation performed			0.4 (0.1-1.3)					
Occupational	Recommendation given		0.7 (0.3-1.7)	2.5 (1.5-3.9)					0.3 (0.1-1.1)
therapist	Recommendation performed		0.1 (0.0-0.8)	0.9 (0.4-1.9)					0.1 (0.0-0.8)
Geronto-	Recommendation given					0.4 (0.1-1.3)	0.7 (0.3-1.7)		
psychiatrist	Recommendation performed					0.0 (0.0-0.6)	0.0 (0.0-0.6)		
Other	Recommendation given	0.3 (0.1-1.1)	1.0 (0.5-2.1)	0.3 (0.1-1.1)	0.7 (0.3-1.7)	0.3 (0.1-1.1)	0.4 (0.1-1.3)	0.6 (0.2-1.5)	1.7 (1.0-3.0)
Other	Recommendation performed	0.1 (0.0-0.8)	0.4 (0.1-1.3)	0.1 (0.0-0.8)	0.4 (0.1-1.3)	0.0 (0.0-0.6)	0.3 (0.1-1.1)	0.1 (0.0-0.8)	1.0 (0.5-2.1)
-	Recommendation given	34.8 (31.4-38.5)	31.8 (28.4-35.4)	16.8 (14.2-19.8)	11.2 (9.0-13.7)	20.5 (17.6-23.6)	31.6 (28.3-35.2)	66.2 (62.6-69.6)	5.7 (4.2-7.6)
Iotal	Recommendation performed	16.5 (14.0-19.5)	10.4 (8.4-13.0)	3.3 (2.2-5.0)	3.0 (2.0-4.6)	3.9 (2.7-5.6)	8.1 (6.3-10.4)	28.9 (25.6-32.4)	2.5 (1.5-3.9)
	ratio %	47,5	32,9	19,8	27,3	19,1	25,7	43,6	43,6

Table 5. Overview of given and performed geriatric recommendations on geriatric domain level

Legend: FS = Functional status

Data were calculated from the total N° pts with data on implementation of geriatric recommendations at chart review after one month (n=689 pts)

Discussion

Geriatric assessment can identify problems that are not always captured by standard anamnesis or physical examination [2]. This information might influence the treatment decision-making process, especially in vulnerable patients for whom intensive cancer treatment is considered [12,13]. However, to what extent are these identified geriatric problems addressed? Which health care professionals or services are the most referred to and for which geriatric problems? To our knowledge, this is the first study to describe the frequency and type of targeted recommendations/interventions based on GA in daily oncology practice on a large scale.

The added value of CGA has been shown over the past two decades in many studies in the geriatric literature, though the benefits seem to arise more from trials with discrete geriatric wards rather than from trials with inpatient geriatric consultation teams [14-18]. Various authors have argued that poor implementation of geriatric recommendations may have accounted in part for the lack of CGA effectiveness in some randomized controlled trials in the general geriatric population [19]. In oncology, many observational studies have been conducted on the value of GA for predicting oncological outcomes [20-22]. None of these studies described in depth recommendations/interventions based on GA and it could be questioned whether in some studies GA was probably used more as a treatment decision-making tool rather than an approach to implement geriatric recommendations [21]. As such, it would be advisable to report information about (non)adherence to GA recommendations in future cancer studies evaluating relevant outcomes including survival, functional decline and quality of life.

Currently, limited data exist regarding interventions following GA in cancer patients. As shown in our results, not all geriatric problems lead to geriatric recommendations and not all geriatric recommendations are carried out. Geriatric recommendations were given in about three fourths of patients (76.2%). This high percentage is not entirely unexpected when we consider the percentage of patients with an impaired test for one of the GA domains shown in Table 2. A third of all the geriatric recommendations (35.3%) were performed in about half (52.1%) of the evaluable patients corresponding to 460 performed recommendations on a total of 1302 given recommendations. The quality and success of implementation of geriatric interventions is dependent on the degree of implementation [23]. In this setting there are no data to compare but our results indicate that there are opportunities for improvement, both in the number of patients that should receive geriatric interventions as in the number of geriatric interventions per patient. However, we have not assessed the reasons for not implementing certain recommendations in this study, and there might be sound and less sound explanations for this. The THCW who performs the GA mostly has an advisory role only with no direct or limited control over patient care [24]. Adherence to a geriatric recommendation is dependent on its importance and anticipated benefit as judged by the treating physician as well as by the patient [25]. It might indeed be that health care workers and physicians do not pay enough attention to the implementation of recommendations. This is a phenomenon that certainly will be present in busy clinics where also

experienced and motivated staff might not be sufficiently available for all patients. Patients can also refuse referrals because they are too busy with oncological appointments and treatments and too exhausted for extra consultations. Or there could be logistical issues, such as too long waiting lists for the geriatric day clinic or other referrals. But on the other hand, treating physicians, teams (and patients) may decide that some of the identified problems are important to know for other reasons (e.g. tolerance of therapy, or life expectancy), but are less priority in terms of approach. In the oncology setting, it makes sense to focus more initially on remediable problems that could interfere with cancer treatment. It should also be acknowledged that 460 performed recommendations in 359 patients is not a small number. Oncological teams may select those recommendations that are most relevant for the patient at that time point. But as we mentioned already, this study does not allow an in-depth evaluation of the reasons for (not) implementing recommendations and further research in this domain is absolutely needed.

When we look at the type of recommendations, it is clear that many patients are referred to the dietician, social worker, and psychologist at the start of their cancer treatment (Table 4). It might be that oncological teams evaluated these referrals as most important for immediate implementation, but it could also be that these health care workers are most easily accessible. Furthermore, it was noteworthy that not many patients (7.3%) were directly referred to the geriatrician. One important reason for this is the shortage of physicians specialized in this discipline. From a health perspective, it is probably most cost-effective to involve geriatricians for a selected group of more complex patients, while the majority of problems can be initially approached by standard paths of care. It should be emphasized that geriatricians could/should be involved in the establishment of geriatric recommendations or care paths for the entire older population with cancer, without necessarily seeing all the patients personally. Referrals to a psychologist, geriatric day clinic, fall clinic and physiotherapist were not often performed when recommended. This should require further attention.

In order to have a better understanding of which geriatric problems led to which geriatric recommendations and whether or not the geriatric recommendations were performed for these problems, we also looked at referrals at the level of the GA domain (Table 5) as described in the methods section. Patients with problems concerning cognitive function were often referred to the geriatric day clinic (12.2%) and as previously mentioned these referrals were not often performed, irrespective of the type of problem. Next to cognitive problems, follow-up of other problems concerning fall risk, depression, and fatigue should get more attention, if not immediately then certainly on the longer term if relevant.

When interpreting our results, it should be noted that the nine participating hospitals had experience with the GA-based approach in oncology from a previous project (2009-2011). Centres that are at the stage of establishing geriatric oncology programs might have more influence of a learning curve [8]. Furthermore, differences between the nine sites may exist with each having their own processes and

paths of care. We evaluated this aspect, and indeed found differences in the number of performed geriatric recommendations between sites (data not shown). However, the main purpose of our study was to describe the global approach in the whole study cohort allowing to have a view on clinical practice. The lack of information for reasons of (non) implementation of recommendations is a weakness of our study, but is extremely difficult to evaluate, since in the Belgian setting, hundreds of care providers are involved at different levels. Within 'geriatric oncology' units, this might be easier to achieve, but the major disadvantage of this setting is that only limited amounts of older persons can be approached, while our concept of 'mobile' GA brought to the treating oncology teams reached more than 1500 patients in one year.

In conclusion, a better understanding of the GA-based approach will allow improving its implementation in order to optimize its effectiveness; next to this our data are also relevant for policymaking with the eventual goal to deliver high quality care in the growing population of older patients with cancer. This implementation study provides more insight into GA-driven interventions in daily oncology practice. Geriatric recommendations were given in about three fourths of patients. At least one geriatric recommendation was performed in approximately half of these patients and about one third of all geriatric recommendations were performed after one month.

Author Contributions

Study concept: C Kenis, J Flamaing, KM, H Wildiers Study design: A Baitar, C Kenis, R Moor, L Decoster, S Luce, D Bron, R Van Rijswijk, M Rasschaert, C Langenaeken, G Jerusalem, JP Lobelle, J Flamaing, K Milisen, H Wildiers Data acquisition: A Baitar, C Kenis

Quality control of data and algorithms: A Baitar, C Kenis Data analysis and interpretation: A Baitar, C Kenis, JP Lobelle, H Wildiers

Statistical analysis: JP Lobelle, A Baitar

Manuscript preparation: A Baitar, C Kenis, JP Lobelle, H Wildiers

Manuscript editing: A Baitar, C Kenis, R Moor, L Decoster, S Luce, D Bron, R Van Rijswijk, M Rasschaert, C Langenaeken, G Jerusalem, JP Lobelle, J Flamaing, K Milisen, H Wildiers

Manuscript review: A Baitar, C Kenis, R Moor, L Decoster, S Luce, D Bron, R Van Rijswijk, M Rasschaert, C Langenaeken, G Jerusalem, JP Lobelle, J Flamaing, K Milisen, H Wildiers

References

1. Naeim A, Reuven D. Geriatric syndromes and assessment in older cancer patients. Oncology 2001;15:1567-1577.

2. Wildiers H, Heeren P, Puts M, Topinkova E, Janssen-Heijnen ML, Extermann M, et al. International Society of Geriatric Oncology Consensus on Geriatric Assessment in older patients with cancer. J Clin Oncol 2014;32(24):2595-603.

3. Hurria A, Wildes T, Blair SL, Browner IS, Cohen HJ, Deshazo M, et al. Senior adult oncology, version 2.2014: clinical practice guidelines in oncology. J Natl Compr Cancer Netw 2014;12(1):82-126.

4. Pallis AG, Fortpied C, Wedding U, Van Nes MC, Penninckx B, Ring A, et al. EORTC elderly task force position paper: approach to the older cancer patient. Eur J Cancer 2010;46(9):1502–1513.

5. World Health Organization. National cancer control programmes. Policies and managerial guidelines, 2nd edition. Geneva: WHO, 2002

6. Kenis C, Decoster L, Van Puyvelde K, De Grève J, Conings G, Milisen K, et al. Performance of two geriatric screening tools in older patients with cancer. J Clin Oncol 2014;32(1):19-26.

7. Kenis C, Bron D, Libert Y, Decoster L, Van Puyvelde K, Scalliet P, et al. Relevance of a systematic geriatric screening and assessment in older patients with cancer: results of a prospective multicentric study. Ann Oncol 2013;24(5):1306-12.

8. Kenis C, Heeren P, Bron D, Decoster L, Moor R, Pepersack T, et al. Multicenter implementation of geriatric assessment in older cancer patients: a survey on treating physicians' general experiences and expectations. J Geriatr Oncol 2014;5(4):431-8.

9. Bellera CA, Rainfray M, Mathoulin-Pelissier S, Mertens C, Delva F, Fonck M, et al. Screening older cancer patients: first evaluation of the G-8 geriatric screening tool. Ann Oncol 2012;23(8):2166-72.

10. Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, et al. Evaluation of the Groningen frailty indicator and the G8 questionnaire as screening tools for frailty in older patients with cancer. J Geriatr Oncol 2013;4(1):32-8.

11. Julious SA. Two-sided confidence intervals for the single proportion: comparison of seven methods by Robert G. Newcombe, Statistics in Medicine 1998; 17:857-872. Statistics in medicine. 2005;24(21):3383-4.

12. Hamaker ME, Schiphorst AH, ten Bokkel Huininck D, Schaar C, van Munster BC. The effect of a geriatric evaluation on treatment decisions for older cancer patients: a systematic review. Acta Oncol 2014;53(3):289-96.

13. Decoster L, Kenis C, Van Puyvelde K, Flamaing J, Conings G, De Grève J, et al. The influence of clinical assessment (including age) and geriatric assessment on treatment decisions in older patients with cancer. J Geriatr Oncol 2013;4(3):235-41.

14. Stuck AE, Siu AL, Wieland GD, Adams J, Rubenstein LZ. Comprehensive geriatric assessment: a meta-analysis of controlled trials. Lancet 1993;342(8878):1032-6.

15. Ellis G, Langhorne P. Comprehensive geriatric assessment for older hospital patients. Br Med Bull 2005;71:45-59.

16. Van Craen K, Braes T, Wellens N, Denhaerynck K, Flamaing J, Moons P, et al. The effectiveness of inpatient geriatric evaluation and management units: a systematic review and meta-analysis. J Am Geriatr Soc 2010;58(1):83-92.

17. Ellis G, Whitehead MA, Robinson D, O'Neill D, Langhorne P. Comprehensive geriatric assessment for older adults admitted to hospital: meta-analysis of randomized controlled trials. BMJ 2011;343:d6553.

18. Deschodt M, Flamaing J, Haentjens P, Boonen S, Milisen K. Impact of geriatric consultation teams on clinical outcome in acute hospitals: a systematic review and meta-analysis. BMC Med 2013;11:48.

19. Allen CM, Becker PM, McVey LJ, Saltz C, Feussner JR, Cohen HJ. A randomized, controlled clinical trial of a geriatric consultation team. Compliance with recommendations. JAMA 1986;16;255(19):2617-21.

20. Puts MT, Hardt J, Monette J, Girre V, Springall E, Alihbai SM. Use of geriatric assessment for older adults in the oncology setting: a systematic review. J Natl Cancer Inst 2012;104(15):1133-63.

21. Hamaker ME, Vos AG, Smorenburg CH, de Rooij SE, van Munster BC. The value of geriatric assessments in predicting treatment tolerance and all-cause mortality in older patients with cancer. Oncologist 2012;17(11):1439-49.

22. Puts MT, Santos B, Hardt J, Monette J, Girre V, Atenafu EG, et al. An update on a systematic review of the use of geriatric assessment for older adults in oncology. Ann Oncol 2014;25(2):307-15.

23. Carroll C, Patterson M, Wood S, Booth A, Rick J, Balain S. A conceptual framework for implementation fidelity. Implement Sci 2007;2:40.

24. Deschodt M, Flamaing J, Rock G, Boland B, Boonen S, Milisen K. Implementation of inpatient geriatric consultation teams and geriatric resource nurses in acute hospitals: a national survey study. Int J Nurs Stud 2012;49(7):842–9.

25. Reuben DB, Maly RC, Hirsch SH, Frank JC, Oakes AM, Siu AL, et al. Physician implementation of and patient adherence to recommendations from comprehensive geriatric assessment. Am J Med 1996;100(4):444-51.

Chapter 4.

Geriatric screening results and the association with severe treatment toxicity after the first cycle of (radio)chemotherapy.

Baitar A. Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D.

This chapter has been published in the Journal of Geriatric Oncology 2014;5(2):179-84.

Abstract

Background

Screening tools are used in geriatric oncology to determine who should receive a Comprehensive Geriatric Assessment (CGA). However, in this prospective study, we evaluated the association between geriatric screening results, measured with the G8 and Groningen Frailty Indicator (GFI), and severe treatment toxicity.

Methods

Patients over 65 years with various types and stages of cancer were screened with the G8 and the GFI prior to start of treatment. The association between geriatric screening results and Serious Adverse Events (SAE) after the first cycle of (radio)chemotherapy was studied with bivariate analysis (normal versus abnormal screening test) and logistic regression analysis.

Results

From 170 screened patients, 85 patients were eligible for this study. The median age was 76 years (range: 66-88 years). The treatment intent was curative in 46% and palliative in 54%. A SAE occurred in 15 patients (18%) of which three resulted in death. There was no significant association between the G8 as a dichotomous predictor (p=0.376) or as a continuous predictor (p=0.298) and the risk for SAE. We also found no significant association for the GFI analysed as a dichotomous predictor (cut-off \geq 4: p=0.384; cut-off \geq 3: p= 0.773), nor as a continuous predictor (p= 0.734). All associations remained insignificant when adjusted for treatment type and comorbidity.

Conclusion

The G8 and the GFI can be used to select patients for CGA but they do not seem to be predictive for short-term severe treatment toxicity.

Introduction

Cancer commonly occurs in older patients. Treatment decision-making is complex in this population. A major issue confronting oncologists is how to effectively select older patients for therapies with significant potential toxicity. It is clear that a distinction should be made between chronological age and biological age, as older patients have a very variable health status. The Comprehensive Geriatric Assessment (CGA) is a well-established approach in geriatrics to evaluate the elderly patients' functional and global health status. The CGA has been recommended by several guidelines in the approach of the older patient with cancer for different purposes of which one is to estimate this risk for severe treatment toxicity in order to guide treatment decision-making (1-3). The most important domains that are evaluated in CGA include functional status, nutrition, comorbidity, cognition, mobility, social support, and depression (4-6). It is also well-known that CGA is time-consuming in routine practice and it is currently not reimbursed by health systems. Therefore, a two-step approach of screening followed by CGA when needed has been proposed. Several geriatric instruments that screen for patients who should receive a CGA, have been or are being validated in an oncology setting (7). We previously studied the G8 and the Groningen Frailty Indicator (GFI) as screening tools for CGA in older patients with cancer (8). In this study, we looked at their association with severe treatment toxicity. It is possible that these geriatric screening tools can be used to estimate the risk for treatment toxicity while at the same time can be used to identify patients who should receive a CGA.

Patients and methods

Study design and patients

Patients with various types and stages of cancer, aged \geq 65 years, with an adequate understanding of the Dutch language were included. Exclusion criteria included severe known dementia, symptomatic brain metastases, and pre-existing major neurological or psychiatric problems. Patients with newly diagnosed cancer or with recurrent disease were assessed before the start of treatment with two geriatric screening tools and a CGA as part of a validation study described in a previous work published from our group (8). In this prospective study, only patients who received chemotherapy or concomitant radiochemotherapy (CRT) were included and this no later than three weeks after geriatric assessment. In this period, radiotherapy or major surgery was not allowed. Patients who deteriorated in the period between geriatric assessment and start of treatment were excluded (e.g. hospitalisation). Serious adverse events (SAE) were recorded during the first cycle or month of treatment. A SAE was defined as any untoward medical occurrence that resulted in death, was life-threatening, required hospitalization or prolongation of existing hospitalization; or resulted in persistent or significant disability/incapacity.

Geriatric screening tools

The G8 questionnaire is a screening tool for CGA in geriatric oncology (9). This tool consists of 8 questions and its development was based on the Mini Nutritional Assessment. A score of ≤ 14 (score range: 0-17) corresponds with an abnormal screening test (8-10).

The Groningen Frailty Indicator (GFI) (15 questions) is a screening instrument for the level of frailty (8,11,12). A GFI score of \geq 4 (score range: 0-15) was regarded as moderately frail according to a panel of geriatric experts. This cut-off value is used in various studies. In our previous study, we suggested to use cut-off GFI \geq 3 to define an abnormal screening test if the GFI is used in the context of screening for CGA. (8). Both cut-off values were evaluated here.

Data analysis

The G8 and the GFI were both evaluated as a dichotomous predictor (normal screening test versus abnormal screening test) and as a continuous predictor for the risk of SAE. The Fisher's exact test was used in bivariate analysis. Continuous predictors were analysed with univariate logistic regression. A multivariate analysis was conducted to adjust for treatment type and comorbidity. The treatments received by the patients were categorized into five types described below. Comorbidity was evaluated with the Cumulative Illness Rating Scale for Geriatrics (CIRS-G) and all analyses were conducted with the total CIRS-G score. All p values reported are two-sided, and p values of <0.05 were considered statistically significant. All analyses were performed using the SPSS 17.0 statistical package (Chicago, IL, USA). Ethics approval for the study was obtained from the Ethical Committee and each patient provided written informed consent prior to study entry.

Figure 1. Study recruitment



Results

Patient characteristics and treatment received

From October 2009 to December 2011, 170 patients were screened of which 85 patients were included in this study (Figure 1). Patient characteristics are summarized in Table 1. The median age of patients was 76 years with a range of 66-88 years. Fifty-four patients (63.5%) had newly diagnosed cancer while 31 patients (36.5%) had progressive disease or relapse. Twenty-nine patients (34%) were hospitalized at the time of geriatric assessment. The results from the CGA are depicted in Table 2. All patients were screened and received a CGA prior to treatment which revealed that the tests for the CGA components nutrition (52%) and IADL (44%) were most frequently impaired. There were no missing data. After geriatric assessment, CRT (1) was started in nine patients (10.6%), 37 patients (43.5%) received platinum-containing regimens (2), nine patients (10.6%) were treated with taxanes (3), 20 patients (23.5%) with antimetabolites (4), and 10 patients (11.8%) with topoisomerase inhibitors (5). The treatment intent was curative in 39 patients (46%) and palliative in 46 patients (54%).

Table 1. Patient characteristics

N° patients	85	100%
Gender		
Male	41	48%
Female	44	52%
Age (years)		
Median	76	
Range	66-88	
Type of cancer		
Uro-genital	21	25%
Digestive tract	19	22%
Breast	18	21%
Head and neck	11	13%
Lung	5	6%
Gynaecological	4	5%
Unknown primary site	4	5%
Bone and soft tissue	2	2%
Non-Hodgkin lymphoma	1	1%

Serious Adverse Events

Fifteen SAE (18%) were reported in 15 different patients (seven women). Twelve had newly diagnosed cancer and eight were treated with a palliative intent. Furthermore, 12 patients received chemotherapy and three patients received CRT. Six patients were hospitalized due to haematological toxicity of which one patient (G8=15; GFI=4) died. Two patients with head and neck cancer were hospitalized due to dysphagia and mucositis, three patients due to symptoms that included dizziness and fatigue, one patient due to vomiting, and one patient due to dyspnea. One patient (G8=14; GFI=6) was hospitalized for renal

insufficiency that resulted in death. Another patient (G8=13; GFI=3) developed pleural effusion resulting in respiratory insufficiency and eventually died from respiratory failure.

	Test result				
	Normal (%)	Abnormal (%)			
Geriatric screening					
G8	31	69			
GFI (cut-off ≥4)	60	40			
GFI (cut-off ≥3)	39	61			
CGA					
ADL	81	19			
IADL	56	44			
Nutrition	48	52			
Cognition	94	6			
Depression	78	22			
Social Support	80	20			
Mobility	72	28			
Comorbidity	71	29			

Table 2. Results for geriatric screening and CGA

Abbreviations: ADL: Activities of Daily Living; IADL: Instrumental Activities of Daily Living; GFI: Groningen Frailty Indicator. ADL was evaluated with the Katz scale, IADL with Lawton-Brody Instrumental scale, nutrition with the Mini Nutritional Assessment, cognition with the Mini-Mental State Examination, depression with the 30-item Geriatric Depression Scale, and social support with the MOS social support survey. The scales are described more in detail elsewhere together with the associated cut-off values (8).

Geriatric screening results and the association with SAE

The GFI

The mean GFI score in this sample was 3.2 (SD 2.1). Fifty-one patients (60%) had a normal screening test. The group of patients without a SAE had a mean GFI score of 3.3 (SD 2.3) while the group of patients with a SAE had a mean GFI score of 3.1 (SD 1.3) prior to treatment. For the alternative cut-off value (GFI score \geq 3), 33 patients (39%) had a normal screening test. For both cut-off values, there was no significant association for SAE in bivariate analysis, p= 0.384 and p=0.773 respectively. We also found no significant association when the GFI was analysed as a continuous variable in univariate logistic regression(p=0.734). The associations remained insignificant when adjusted for treatment type and comorbidity (Table 3).

The G8

The mean G8 score for the whole sample was 11.9 (SD 3.2). Twenty-six patients (30.6%) had a normal screening test. The mean G8 score for the group of patients with and without a SAE was respectively 12.7 (SD 3.2) and 11.7 (SD 3.2) prior to treatment. There was no significant association between the G8 and the occurrence of SAE in bivariate analysis (p=0.376), in univariate logistic regression analysis (p=0.298), and when adjusted for treatment type and comorbidity (Table 3).

	OR** (95%CI)	p-value	OR*** (95%CI)	p-value
G8 screening tool				
Continuous score	0,91 (0,76-1,09)	0,30	0,86 (0,70-1,06)	0,16
G8 ≤14 vs normal score	0,60 (0,19-1,91)	0,376*	0,52 (0,14-1,89)	0,32
Groningen Frailty Indicator (GFI)				
Continuous score	1,05 (0,80-1,37)	0,73	1,14 (0,84-1,55)	0,41
GFI ≥4 vs normal score	0,49 (0,14-1,67)	0,384*	0,38 (0,10-1,42)	0,15
GFI ≥3 vs normal score	1,33 (0,41-4,32)	0,773*	1,30 (0,36-4,73)	0,69

Table 3. Association between geriatric screening results and severe treatment toxicity

*Results from bivariate analysis are shown (Fisher's Exact test).

Unadjusted. *Adjusted to treatment type and comorbidity.

Discussion

In oncology practice, classical performance status measures (e.g. Karnofsky Performance Status, ECOG) are regarded to be of limited value to estimate whether a patient is likely to tolerate a certain chemotherapy regimen (13,14). Older patients enrolled in clinical trials derive the same treatment benefits as younger patients (2). However, data from clinical trial participants are not representative for all older patients. There are no validated methods to identify older patients fit enough to receive the same treatment as younger patients. A CGA-based approach has been recommended in the approach of the older patient with cancer. In our previous study, we showed that with geriatric screening tools, it is possible to identify most of the patients who should receive a CGA but also that a significant number of false positives should be expected. The completion of the G8 or the GFI only takes a few minutes. However, the benefit of the screening step of the two-step approach has been raised into question by a recent systematic review of available frailty screening methods (7,15). Whereas studies validating these screening tools against the CGA are useful, prospective studies evaluating the association of screening results and treatment tolerance could give us valuable information directly relevant for the clinician.

The G8 was developed to screen patients who should receive a CGA (normal screening test versus abnormal screening test). The GFI, on the other hand, was developed as a (continuous) measure for frailty. This was the rationale to evaluate both screening tools as a dichotomous and continuous variable. It would be interesting to know if there is a significant association with treatment tolerance, even considering that both screening tools were not developed for this purpose. Moreover, results for one of these screening tools might be available anyway in a setting where older patients are screened systematically for CGA. However, our results showed no significant association for both screening tools and SAE after the first cycle of (radio)chemotherapy, also when adjusted for treatment type and comorbidity. Furthermore, it didn't matter if the screening can be used to determine who should receive a CGA but the GFI and the G8 do not seem to be predictive for severe treatment toxicity. It should be noted that, although we did adjust for treatment type, this adjustment was not at the level of

the individual chemotherapy regimen, as is done with the MAX2 index (16). A confirmation of our results in a more homogeneous and larger sample of patients is warranted. We also mention that recently two promising instruments have been developed specifically to estimate the risk for severe chemotherapy toxicity (14,17). However, further validation is still needed before use in routine practice and such validation is underway.

In this study, we primarily looked at tools that might be used in the first step of the two-step approach of screening followed by CGA. Various studies have evaluated the predictive value of CGA for chemotherapy toxicity, without the screening step (18-21). Over different studies, depression, functional status, general mental health, and social support for example have been identified as independent predictors for toxicity. On the other hand, in other studies weak or no associations were reported (19,21). A recent systematic review also found little consistency for the association of CGA with other relevant clinical outcomes besides chemotherapy toxicity and concluded that the available evidence is currently to inconsistent to guide clinical decision making (22). An explorative analysis on our data showed that when all CGA variables were entered simultaneously, only cognition was borderline significant (p=0.049) as being predictive for SAE after the first cycle of therapy (data not shown). Comorbidity tended towards significance (p=0.062). Unexpectedly, patients with a better cognitive function had a higher risk for SAE. This association remained significant after adjustment for treatment type. It should be noted that confidence in the robustness of this analysis is limited by the small sample size combined with the high number of predictor variables entered into the model.

Some other considerations need to be made as well when interpreting our study results. Firstly, the lack of a relationship between the studied screening tools and treatment tolerance might be explained by confounding by indication. Vulnerable patients might have been treated with less toxic regimens than fit patients. This might as well explain the observed relationship between cognitive function and the risk for SAE. Also, the treating oncologists had access to the screening and CGA results and this could have influenced the treatment choice which in turn could have influenced treatment tolerance. However, we believe this was unlikely in our study because the oncologists were still getting familiar with the results of a multidimensional CGA and its interpretation. Secondly, for our outcome severe treatment toxicity, we evaluated SAE after the first cycle of chemotherapy. We looked at short-term SAE because patients who develop severe toxicity often do so during the first cycle of chemotherapy (17). SAE were not uncommon (18%) in our sample. But it would be useful to have information for other time-points as well. In this regard, one study showed that the GFI was related to mortality but not to the probability to complete chemotherapy (23). Thirdly, for severe chemotherapy toxicity we evaluated SAE while other studies, but not all, evaluate grade 3 and grade 4 chemotherapy toxicity. The endpoint SAE has the advantage that it relates more to the most severe and possibly life-threatening toxicity. This is the type of toxicity the clinician especially wants to avoid. Then again, other important toxicity that does not fall under the definition of SAE might be missed. It should be noted that SAE in our study could have been related to the given chemotherapy and/or radiotherapy but also to progressive disease. It would be perhaps interesting to evaluate different definitions of severe treatment toxicity in future studies.

Author Contributions

Study concept and design: D. Schrijvers, A. Baitar Data collection: A. Baitar, F. Van Fraeyenhove, A. Vandebroek, E. De Droogh, D. Galdermans, J. Mebis, D. Schrijvers Analysis and interpretation: A. Baitar

Manuscript preparation: A. Baitar, F. Van Fraeyenhove, A. Vandebroek, E. De Droogh, D. Galdermans, J. Mebis, D. Schrijvers

Manuscript Editing: A. Baitar, F. Van Fraeyenhove, D. Schrijvers

Manuscript review and approval: A. Baitar, F. Van Fraeyenhove, A. Vandebroek, E. De Droogh, D. Galdermans, J. Mebis, D. Schrijvers

References

1) Extermann M, Aapro M, Bernabei R, Cohen HJ, Droz JP, Lichtman S, et al. Use of comprehensive geriatric assessment in older cancer patients: recommendations from the task force on CGA of the International Society of Geriatric Oncology (SIOG). Crit Rev Oncol Hematol 2005;55(3):241-52.

2) Hurria A, Browner IS, Cohen HJ, Denlinger CS, deShazo M, Extermann M, et al. Senior adult oncology. J Natl Compr Canc Netw 2012;10(2):162-209.

3) Pallis AG, Fortpied C, Wedding U, Van Nes MC, Penninckx B, Ring A et al. EORTC elderly task force position paper: approach to the older cancer patient. Eur J Cancer 2010;46(9):1502-13.

4) Puts MT, Hardt J, Monette J, Girre V, Springall E, Alibhai SM. Use of geriatric assessment for older adults in the oncology setting: a systematic review. J Natl Cancer Inst. 2012;104(15):1133-63.

5) Rodin MB, Mohile SG. A practical approach to geriatric assessment in oncology. J Clin Oncol 2007;25(14):1936-44.

6) Maas HA, Janssen-Heijnen ML, Olde Rikkert MG, Machteld Wymenga AN. Comprehensive geriatric assessment and its clinical impact in oncology. Eur J Cancer 2007;43(15):2161-9.

7) Hamaker ME, Jonker JM, de Rooij SE, Vos AG, Smorenburg CH, van Munster BC. Frailty screening methods for predicting outcome of a comprehensive geriatric assessment in elderly patients with cancer: a systematic review. Lancet Oncol. 2012;13(10):e437-44.

8) Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, et al. Evaluation of the Groningen Frailty Indicator and the G8 questionnaire as screening tools for frailty in older patients with cancer. J Geriatr Oncol 2013;4:32-38.

9) Bellera CA, Rainfray M, Mathoulin-Pélissier S, Mertens C, Delva F, Fonck M et al. Screening older cancer patients : first evaluation of the G-8 geriatric screening tool. Ann Oncol 2012;23(8):2166-2172.

10) Pottel L, Boterberg T, Pottel H, Goethals L, Van Den Noortgate N, Duprez F et al. Determination of an adequate screening tool for identification of vulnerable elderly head and neck cancer patients treated with radio(chemo)therapy. J Geriatr Oncol 2011;3(1):24-32.

11) Steverink N, Slaets JPJ, Schuurmans H, Van Lis M. Measuring Frailty. Development and testing of the Groningen Frailty Indicator (GFI). Gerontologist 2001;41(1):236-237.

12) Schuurmans H, Steverink N, Lindenberg S, Frieswijk N, Slaets JP. Old or frail: what tells us more? J Gerontol A Biol Sci Med Sci 2004;59(9):M965-5.

13) Repetto L, Fratino L, Audisio RA, Venturino A, Gianni W, Vercelli M et al. Comprehensive geriatric assessment adds information to Eastern Cooperative Oncology Group performance status in elderly cancer patients: an Italian Group for Geriatric Oncology Study. J Clin Oncol 2002;20(2):494-502.

14) Hurria A, Togawa K, Mohile SG, Owusu C, Klepin HD, Gross CP, et al. Predicting chemotherapy toxicity in older adults with cancer: a prospective multicenter study. J Clin Oncol 2011;29(25):3457-65.

15) Mathoulin-Pélissier S, Bellera C, Rainfray M, Soubeyran P. Screening methods for geriatric frailty. Lancet Oncol. 2013;14(1):e1-2.

16) Extermann M, Chen H, Cantor AB, Corcoran MB, Meyer J, Grendys E, et al. Predictors of tolerance to chemotherapy in older cancer patients: a prospective pilot study. Eur J Cancer 2002;38(11):1466-73.

17) Extermann M, Boler I, Reich RR, Lyman GH, Brown RH, Defelice J, et al. Predicting the risk of chemotherapy toxicity in older patients: The Chemotherapy Risk Assessment Scale for High-Age Patients (CRASH) score. Cancer 2012;118(13):3377-86.

18) Freyer G, Geay JF, Touzet S, Provencal J, Weber B, Jacquin JP, et al. Comprehensive geriatric assessment predicts tolerance to chemotherapy and survival in elderly patients with advanced ovarian carcinoma: a GINECO study. Ann Oncol 2005;16(11):1795-800.

19) Marinello R, Marenco D, Roglia D, Stasi MF, Ferrando A, Ceccarelli M et al. Predictors of treatment failures during chemotherapy: A prospective study on 110 older cancer patients. Arch Gerontol Geriatr 2009;48(2):222-6.

20) Clough-Gorr KM, Stuck AE, Thwin SS, Silliman RA. Older breast cancer survivors: geriatric assessment domains are associated with poor tolerance of treatment adverse effects and predict mortality over 7 years of follow-up. J Clin Oncol 2010;28(3):380-6.

21) Puts MT, Monette J, Girre V, Pepe C, Monette M, Assouline S, et al. Are frailty markers useful for predicting treatment toxicity and mortality in older newly diagnosed cancer patients? Results from a prospective pilot study. Crit Rev Oncol Hematol 2011;78(2):138-49.

22) Hamaker ME, Vos AG, Smorenburg CH, de Rooij SE, van Munster BC. The value of geriatric assessments in predicting treatment tolerance and all-cause mortality in older patients with cancer. Oncologist 2012;17(11):1439-49.

23) Aaldriks AA, Maartense E, le Cessie S, Giltay EJ, Verlaan HA, van der Geest LG, et al. Predictive value of geriatric assessment for patients older than 70 years, treated with chemotherapy. Crit Rev Oncol Hematol 2011;79(2):205-12.

Chapter 5.

The added value of geriatric screening and assessment to predict overall survival in older patients with cancer

Kenis C*, Baitar A*, Decoster L, De Grève J, Lobelle JP, Flamaing J, Milisen K, Wildiers H

This chapter has been accepted for publication in Cancer.

* equal contribution

The online version of the published paper will contain supplementary material which have not been included in this dissertation.

Abstract

Aim

The aim of this study is to determine and compare the added prognostic value of screening tools, geriatric assessment (GA) components and GA summaries to clinical information for overall survival (OS) in older patients with cancer.

Patients and Methods

A screening and a 10-item geriatric assessment (GA) were systematically performed in patients \geq 70 years with cancer. Cox regression analyses were conducted to evaluate the added prognostic value for OS of screening tools, GA and GA summaries to clinical information (age, stage, tumor type) in two cohorts (A and B). Cox models were compared based on Akaike Information Criterion and the Concordance Probability Estimate. Analyses were performed on two independent cohorts.

Results

Complete case analysis was available for 763 patients (median age 76) in cohort A and for 402 patients (median age 77) in cohort B. In both cohorts, most individual GA components were independent prognostic factors for OS. Nutritional status (assessed by the Mini Nutritional Assessment-Short Form) and functional status (assessed by Instrumental Activities of Daily Living) consistently displayed a strong capacity to predict OS. Inconsistent results were found for screening tools. GA summaries perform the best in comparison with the screening tools and the individual GA components.

Conclusions

Most individual GA components, especially nutritional status and functional status, are prognostic factors for OS in older patients with cancer. GA summaries provide more prognostic information than individual GA components, but only moderately improve the prognostic baseline model with clinical information.

Introduction

Epidemiologic research announces a significant increase of older patients with cancer in the following decades ^{1,2}. Because important treatment and outcome variations are reported and due to the heterogeneity and the lack of evidence-based treatment guidelines within this population, international oncologic organizations like the International Society of Geriatric Oncology (SIOG) and the European Organization for Research and Treatment of Cancer (EORTC) recommended that some form of comprehensive geriatric assessment (CGA) should be mandatory to guide oncologic treatment decisions ^{3,4}.

CGA is currently the gold standard to evaluate the global health status and clinical frailty level of individuals and is defined as a multidimensional, interdisciplinary diagnostic process focusing on determining an older person's medical, psychosocial and functional capability in order to develop a coordinated and integrated plan for treatment and long-term follow-up ⁵. In the general (non-oncologic) geriatric population, CGA-guided treatment plans have been shown to improve overall survival (OS), quality of life (QoL), functional status (FS) and decrease the risk of hospitalization and nursing home placement ⁶⁻¹⁰. When applied to patients with cancer, CGA has been shown to identify previously unknown health problems, to predict treatment-related toxicity and oncologic outcomes including OS, and to influence cancer treatment decisions ¹¹⁻¹³. However, the majority of studies of 'CGA' in the older population with cancer focused on the implementation of systematic geriatric screening and geriatric assessment (GA), the first two steps of the CGA process ^{5,8}. Several studies have identified items within the GA itself (e.g. FS ¹⁴⁻¹⁶, nutritional status ¹⁴⁻¹⁸ and mental health ¹⁴⁻¹⁶) that were independent predictors of mortality.

To select patients who would benefit from GA, a number of geriatric screening tools have been developed, like G8, Vulnerable Elders Survey – 13 (VES-13) and the Flemish version of the Triage Risk Screening Tool (fTRST). Some of these screening tools can also provide important information about treatment-related toxicity, risk of functional decline and OS 4,19,20 .

However, a GA does not yield a validated 'summary score', so it remains difficult to precisely quantify the patients' global health status. For this reason, some attempts to summarize and categorize GA results have been proposed in geriatric oncology, but the included elements and cut-offs are arbitrary, and do not capture the complexity of GA and the ageing process itself ²¹. Despite this, the most consistent finding is the association between mortality and a GA summary score (e.g. frail/vulnerable/fit) ^{16,22-25}.

The aim of this study is to determine and compare the added prognostic value for OS of two geriatric screening tools, individual GA components, and two GA summaries to a baseline model of clinical information in older patients with cancer.

PATIENTS AND METHODS

Design

To determine and to compare the added value of screening, individual GA components, and two GA summaries (henceforth, GA components and GA summaries are called together 'geriatric information') to a baseline model of clinical information to predict OS, this study integrates two different study cohorts, cohort A⁴ and cohort B²⁶ that were extensively reported previously (see figure 1). Both study cohorts come from prospective, multicenter, observational cohort studies. Patients with cancer aged 70 years or older were approached for inclusion during a hospital visit at diagnosis or at disease progression/relapse, when a treatment decision had to be made. In the study of cohort A (n=937; period October 2009 – July 2011), inclusion was limited to the following six tumor types: breast, colorectal, ovarian, lung and prostate cancer and hematologic malignancies. This study focused on the implementation of a systematic geriatric screening and assessment⁴. In the study of cohort B (n=1550; period August 2011 – July 2012), patients with all tumor types (except non-melanoma skin cancer) and hematologic malignancies were included. The focus of this study was on geriatric recommendations based on GA-results²⁶. In both studies, patients were assessed in a uniform manner: for study cohort A in two Belgian hospitals ('Universitair Ziekenhuis Brussel' (UZB) and 'Universitaire Ziekenhuizen Leuven' (UZL))⁴ and for study cohort B in six academic and three non-academic hospitals spread all over Belgium²⁶. The analysis for cohort B was limited to patients of UZB and UZL and to patients with breast, colorectal, lung, prostate, and ovarian cancer so this would correspond with the hospitals and the studied tumor types in cohort A. Patients with a hematologic malignancy were excluded for the analysis presented in this study because classical staging is not applicable for most of these tumors (and so cannot be integrated in prognostic models). Complete case analysis was taken into account, leaving 763 patients in cohort A and 402 patients in cohort B.

Screening and geriatric assessment

At baseline, a trained health care worker (THCW) performed a screening and GA in all patients, as previously reported ^{4,26}.

Patients were assessed by two geriatric screening tools, G8 and fTRST, each evaluating the presence of a geriatric risk profile. G8 includes seven items from the Mini Nutritional Assessment (MNA) and one age-related item. The total score varies from 0–17. A G8 score of ≤ 14 indicates the presence of a geriatric risk profile ^{27,28}. The fTRST is a five-item screening tool with a range from 0–6^{4,29,30}. Within the oncologic population, a score of ≥ 1 indicates the presence of a geriatric risk profile⁴. Also, the Eastern Cooperative Oncology Group - Performance Status (ECOG-PS) was evaluated³¹. This screening tool, not age-specific, is frequently used in oncology to classify the performance status of patients.

The GA included 10 components based on the 2014 SIOG recommendation guideline for GA¹³: social data with living situation, FS assessed by the Katz's Activities of Daily Living (ADL)³² and by Lawton's

Instrumental Activities of Daily Living (IADL)³³, fall history during the last year³⁴, fatigue assessed by the MOB-T³⁵, mental status assessed by the Mini Mental State Examination (MMSE)³⁶ and Geriatric Depression Scale (GDS-15)³⁷, nutritional status assessed by the Mini Nutritional Assessment-Short Form (MNA-SF)³⁸, comorbidities assessed by the Charlson Comorbidity Index (CCI)³⁹ and a polypharmacy assessment⁴⁰.

GA summaries

In this study, the 10-item GA is a GA summary which integrates the results of every GA component mentioned above and thus captures the multidimensional GA evaluating the global health of the older patient with cancer as a whole. All components are integrated dichotomously. However, it doesn't calculate a global GA end score and no cut-off is defined to differentiate between fit and frail.

The 'Leuven Oncogeriatric Frailty Score' (LOFS) is a recently developed tool to summarize clinical frailty. The LOFS is a GA summary score as it calculates a global GA end score that integrates 5 fundamental aspects determining a patient's fitness/frailty status, i.e. capability to autonomously perform activities of daily living (ADL and IADL), mental status (MMSE), nutritional status (MNA-SF) and comorbidities (CCI) into a single, semi-continuous score on a scale from 0 = severely frail to 10 = fit ⁴¹ (see appendix A, available in the online publication). In detail, the following categories are included: score 0-2 (severely frail), score 3-4 (frail), score 5-6 (vulnerable), score 7-8 (slightly vulnerable) and score 9-10 (fit).

Statistical analysis

The baseline model of clinical information consisted of age, stage and tumor type. Age was categorized in three levels: 70-74, 75-79, and \geq 80 years old. Comparisons of continuous data were done with Student's t test and by the Wilcoxon's test as appropriate, while comparisons of categorical data were performed with Chi Square test and/or with the Fisher's exact test as appropriate. The statistical significance level used was 0.05.

Follow-up time was defined as the difference from the first GA assessment to the last follow-up for all censored patients. Median, minimum and maximum follow-up time are reported.

The Kaplan-Meier method was used to describe survival and log-rank and Wilcoxon tests were used to compare different groups.

Univariable cox proportional hazard regression analyses were conducted to evaluate the prognostic value of patient and clinical characteristics, screening tools and geriatric information. Multivariable analyses were performed as well for the analyses with screening tools and geriatric information to adjust

for clinical information. OS was calculated from date of GA to date of death from any cause or last follow-up for censored patients. Results were considered significant if p<0.05.

We performed Cox proportional hazard regression analyses to build multiple prognostic models. To assess the contribution of screening tools and geriatric information, each component was added separately to the baseline model of clinical information. The proportional hazards assumption was tested and not violated. Multicollinearity was also assessed with the variance inflation factor (VIF) and was not present (VIF's<3).

The discriminatory ability of each prognostic model was assessed using the Concordance Probability Estimate (CPE)⁴². The interpretation of the CPE is identical to the widely used but less robust c-index and indicates the proportion of all pairs of patients whose predicted survival times are correctly ordered among all patients that can actually be ordered. In other words, a model with a higher CPE corresponds with a model with a better predictive accuracy. The global fit of the Cox models was compared by means of the difference of Akaike Information Criterion (AIC) of the baseline model and the extended model (Δ AIC). A higher Δ AIC value corresponds with a more explanatory and informative model. The screening tools and all individual GA components were assessed as dichotomous variables. The LOFS was assessed as a categorical variable. All analyses were performed using SAS version 9.3 (SAS Institute, Cary, NC) and SPSS 17.0 (Chicago, IL).

The added prognostic value of screening tools and geriatric information was tested on two different consecutive large cohorts of patients, to evaluate the consistency between the results. The aim was not to perform a formal validation, as we did not develop a prognostic tool.

Results

Patient population

In cohort A, a total of 788 patients with carcinoma were considered for data analysis and in cohort B a total of 491 patients. Complete case analysis was available for cohort A and B in 763 and 402 patients respectively (see figure 1).

In cohort A, the median age was 76 years (range: 70-95) with 67.8% women and in cohort B, the median age was 77 years (range: 70-95) with 66.7% women. In both cohort A and B, the most prevalent carcinoma were breast and colorectal cancer in 48.8% and 24.4% of the patients and in 42.0% and 32.8% of the patients respectively. At the moment of inclusion in the study, 63.7% had a newly diagnosed cancer in cohort A and 71.4% in cohort B, whereas 36.3% had disease progression or relapse in cohort A and 28.6% in cohort B. More details of patient demographics and clinical characteristics are listed in table 1. Significant differences between both cohorts were present for tumor type (p=0.019), oncologic setting (p=0.008) and surgical treatment (p=0.006).

After a median follow-up of 61.4 months (range 0.7-75.6 months) in cohort A, and 45.7 months (range 7.3-54.5 months) in cohort B, there was no significant difference (p=0.496) in OS between both cohorts. The median survival time was 33.2 months (range: 0.09-75.6; 471 deaths) and 37.6 months (range: 0.16-54.5; 214 deaths) respectively.

The results of the screening and GA are listed in table 2. Both cohorts were similar based on the results of the G8, ECOG-PS and most individual GA components, though with a significant difference for fTRST (p<0.001), ADL (p<0.001), IADL (p=0.01) and MMSE (p<0.001).

The prognostic value for OS of patient and clinical characteristics, screening tools and geriatric information

The following variables were analyzed: patient and clinical characteristics (gender, age, stage and tumor type), screening tools (G8, fTRST, ECOG-PS), individual GA components (living situation, ADL, IADL, fall history, MOB-T, MMSE, GDS-15, MNA-SF, CCI and polypharmacy), and two GA summaries (10-item GA, LOFS).

The univariable analyses show that all variables were prognostic for OS in both cohorts with the exception of living situation in cohort A and with the exception of living situation and fTRST in cohort B (see table 3 and table 4). Kaplan-Meier curves with associated log-rank tests are shown in appendix B (available in the online publication). The analyses with screening tools and geriatric information were repeated and adjusted for clinical information. This showed that all variables were prognostic for OS with the exception of living situation and comorbidity in cohort A and with the exception of the G8, living situation, and polypharmacy in cohort B (table 4).

The comparison of the performance of models extended with screening tools and geriatric information

Detailed results for cohort A and cohort B are summarized in table 5. A visual overview of the discriminatory ability of the models can be found in figure 2. The baseline model of clinical information had a good discriminatory ability with CPE 0.728 and CPE 0.750 respectively. The addition of the screening tools improved the discriminatory ability (CPE) and the fit of the model (Δ AIC) compared to the baseline model in cohort A. The G8 model improved the baseline model the most among the screening tools with CPE 0.747 and Δ AIC 57.26. In contrast, in cohort B the G8 model decreased the quality of the baseline model (Δ AIC -0.10) while slightly improving in discriminatory ability (CPE 0.753). The ECOG-PS model had the best performance in cohort B with CPE 0.762 and Δ AIC 16.80.

The models extended with the 10 individual GA components did not always improve the discriminatory ability while the fit of the model always improved with the exception of the addition of comorbidity measured by the CCI in cohort A. Nutritional status measured by the MNA-SF (CPE 0.756; Δ AIC 34.11) followed by FS measured by the IADL (CPE 0.754; Δ AIC 31.89) improved the baseline model the most in cohort A in terms of both CPE and Δ AIC. In cohort B, the CPE improved the most by adding

nutritional status measured by the MNA-SF (CPE: 0.772) followed by fatigue measured by the MOB-T (CPE: 0.771) while the fit of the model improved the most by adding IADL (Δ AIC: 10.00) followed by MOB-T (Δ AIC: 9.84) and then by MNA-SF (Δ AIC: 8.78).

The 10-item GA improved the baseline model with CPE 0.755 and Δ AIC 57.72 in cohort A and with CPE 0.775 and Δ AIC 23.58 in cohort B. In comparison, the LOFS improved the baseline model with CPE 0.745 and Δ AIC 37.82 in cohort A and with CPE 0.755 and Δ AIC 34.81 in cohort B.

Discussion

As outlined in the introduction, several GA components have been shown to be independent predictors of mortality in older patients with cancer. This study shows in two separate cohorts that most individual GA components are prognostic factors for OS in patients with cancer independent from clinical information (age, stage, and tumor type) in a heterogeneous oncologic population. Nutritional status and IADL consistently displayed strong individual prognostic capacity compared to the other GA components. Although previous research shows inconsistent results on which GA components are prognostic for OS ^{4,10}. Besides living situation in both cohorts, comorbidity measured by the CCI in cohort A and polypharmacy as a proxy for comorbidity in cohort B were not prognostic factors when adjusted for clinical information. The lack of a prognostic impact of comorbidity might be related to an already poor prognosis in patients with certain tumor characteristics and/or to limitations of the CCI instrument.

In this study, we also examined the added prognostic value of screening tools and geriatric information to predict OS by comparing the performance of multiple extended prognostic models in reference to a baseline model of clinical information (age, tumor type, and stage). This method also allows us to have a better view on the relative importance of the individual GA components and the GA as a whole, i.e. the 10-item GA. In contrast to other studies, we compared the performance of prognostic models with the AIC, which is a more sensitive measure than concordance indices (i.e. CPE, c-index) for model selection. This showed that nutritional status and IADL have a high added value to clinical information compared to the other GA components, which is not surprising since both are strong prognostic factors in cohort A and cohort B.

Furthermore, the AIC results indicate that the models integrating and summarizing different components of GA (10-item GA and LOFS) perform better than the individual GA components. The 10-item GA had the best performance in cohort A and the second best in cohort B after the LOFS. These GA summaries capture more the multidimensional process of ageing, so it is not surprising that they perform better than individual GA components. The prognostic value of the LOFS, a GA summary score, was demonstrated for the first time in this study. The added value of the LOFS in addition to clinical

information was consistent in the two cohorts and was higher than the added value of the best individual GA components in both cohorts. The LOFS seems to be an interesting tool as it has a good prognostic value in addition to clinical information and it summarizes the results of the GA in a single score, which has the advantage of simplicity towards interpretation necessary for daily practice.

While similar results were observed in both cohorts for the most prognostic individual GA components and the screening tool ECOG-PS, inconsistent results are found for some of the other individual GA components and the two geriatric screening tools (G8 and fTRST). The conflicting results of the G8 are especially remarkable when clinical information is taken into account. The G8 showed a poor prognostic contribution in cohort B, in contrast to results in cohort A where the G8 added almost as much prognostic information than the 10-item GA. The majority of the patients in both cohorts had breast cancer or colorectal cancer (CRC), however there were significantly more CRC diagnoses in cohort B, though there was no significant difference in OS between the two cohorts. Given that malnutrition is more prevalent in CRC and the G8 consists of seven questions from the MNA questionnaire, it is likely that the prognostic information provided by both variables overlap which might have contributed to the lack of an added value of the G8 in cohort B. The other variables included in the models (i.e. age, stage, G8 scores) did not differ between the cohorts and so do not explain the different results for the prognostic value of the G8.

Patients in cohort B had more often CRC as discussed previously, more often a new diagnosis, more often surgery, more had a geriatric risk profile according to the fTRST (but not according to the G8), were more dependent for ADL and IADL, and had more often an abnormal score for cognition. This indicates that patients in cohort B might have been more vulnerable than in cohort A, though as mentioned before this did not influence survival but might have influenced the prognostic value of certain aspects of geriatric information in addition to clinical information. Of course, other confounding factors should be taken into account in relation to the heterogeneity of the studied populations, some of which are discussed further at the end of the discussion.

Until now, the discussion has been based on the calculated hazard ratios of the individual parameters and on the performance of the extended models according to the AIC values. With the AIC values, models can be ranked allowing the selection of the best model. However, the AIC does not provide a measure of performance as such as it might select the best model out of a series of poor models. The CPE, on the other hand, does provide an interpretable measure of performance. The baseline models showed good discrimination in both cohorts (cohort A: CPE 0.728; cohort B: CPE 0.750) and were able to correctly order the survival time of two randomly chosen patients about three out of four times. Only small differences in CPE values were observed between the better models and the ranking of models based on the more sensitive AIC and CPE yields different patterns, though the model extended with the 10-item GA was always the best or second best model in accordance with results based on AIC. However, the concordance indices (here CPE) are insensitive to detecting improvements in model

performance when a parameter is added to a model with already important predictors. (43). Based on the CPE, the best models in our study (i.e. the 10-item GA model and the MNA model) improved the overall performance of the model of clinical information by about 2.5 %, so a significant amount of deaths are still not explained by the abovementioned parameters.

Several studies have evaluated the association between GA and mortality in patients with cancer for different purposes. In accordance with our first analyses, most studies focus on the most prognostic individual GA components to improve the identification of patients with an increased mortality risk for appropriate interventions. From these analyses can be concluded that GA has some prognostic value but that due to inconsistent results no specific recommendations can be given to guide treatment-decision making. In contrast to some of these studies, we did not show that certain GA components predict mortality but that most GA components do. This in two separate cohorts. Others focus on the prediction of early death or aim to develop nomograms for example. Prognosis or the estimation of life expectancy is crucial in the treatment decision-making process. However, none of these studies focused on the additional value of the GA as a whole in estimating life expectancy prior to the start of treatment. In this study, we quantified how much the GA as whole improves prognosis, which no prior study has done before.

When interpreting our results some considerations need to be made. Our study population was heterogeneous in terms of oncologic parameters; different tumor types, different staging, different oncologic (molecular) subtypes, different treatments, different susceptibility to treatment, etc. Many of these oncologic factors can have an important impact on OS and are not captured in this study. Ideally, the impact of screening tools and geriatric information should be evaluated in tumor populations with exactly the same tumor and treatment characteristics, while only having differences in geriatric parameters. It is extremely difficult to obtain large samples sizes with such criteria. Moreover, findings in such populations are only relevant for those highly selected populations and would need to be evaluated separately in many other specific tumor settings, which is not realistic. It is possible that in such homogeneous populations, the incremental value of screening tools and geriatric information is larger than about 2.5% because in our current heterogeneous population, these effects might be diluted by major differences in oncologic prognosis. On the other hand, our heterogeneous population in both cohorts can be an advantage as it is representative of a large cancer population. Moreover, OS prediction is far from the only goal of GA. GA is important to estimate the risk of treatment induced toxicity^{41,42}, to detect previously unknown health problems⁹, and to allow directed geriatric interventions that can improve therapy compliance, quality of life, and outcome.

Conclusions

Most individual GA components, especially nutritional status assessed by MNA-SF and functional status assessed by IADL, are independent prognostic factors for OS in older patients with cancer. GA summaries provide more prognostic information than individual GA components, but only moderately improve the prognostic baseline model with clinical information.

Author contributions

conception and design: CK, LD, JPL, JF, KM, HW

- data acquisition: CK, LD, HW
- data analysis and interpretation: CK, AB, LD, JDG, JPL, JF, KM, HW
- writing-initial draft: CK, AB, LD, JPL, HW
- writing-critical revision: CK, AB, LD, JDG, JPL, JF, KM, HW
- statistical analyses: JPL
- supervision: CK, AB, LD, JDG, JPL, JF, KM, HW

References

- Howlader N NA, Krapcho M, Garshell J, Neyman N, Altekruse SF, Kosary CL, Yu M, Ruhl J, Tatalovich Z, Cho H, Mariotto A, Lewis DR, Chen HS, Feuer EJ, Cronin KA (eds). SEER Cancer Statistics Review, 1975-2010. 2013; based on November 2012 SEER data submission, posted to the SEER website, April 2013. Available at: http://seer.cancer.gov/csr/1975_2010/.
- 2. Smith BD, Smith GL, Hurria A, Hortobagyi GN, Buchholz TA. Future of cancer incidence in the United States: burdens upon an aging, changing nation. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology*. 2009;27(17):2758-2765.
- Giantin V, Valentini E, Iasevoli M, et al. Does the Multidimensional Prognostic Index (MPI), based on a Comprehensive Geriatric Assessment (CGA), predict mortality in cancer patients?
 Results of a prospective observational trial. *Journal of geriatric oncology*. 2013;4(3):208-217.
- 4. Kenis C, Decoster L, Van Puyvelde K, et al. Performance of two geriatric screening tools in older patients with cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology*. 2014;32(1):19-26.
- Rubenstein LZ, Stuck AE, Siu AL, Wieland D. Impacts of geriatric evaluation and management programs on defined outcomes: overview of the evidence. *Journal of the American Geriatrics Society*. 1991;39(9 Pt 2):8S-16S; discussion 17S-18S.
- 6. Cohen HJ, Feussner JR, Weinberger M, et al. A controlled trial of inpatient and outpatient geriatric evaluation and management. *The New England journal of medicine*. 2002;346(12):905-912.
- Ellis G, Whitehead MA, Robinson D, O'Neill D, Langhorne P. Comprehensive geriatric assessment for older adults admitted to hospital: meta-analysis of randomised controlled trials. *BMJ (Clinical research ed).* 2011;343:d6553.
- Stuck AE, Siu AL, Wieland GD, Adams J, Rubenstein LZ. Comprehensive Geriatric Assessment - A Metaanalysis of Controlled Trials. *Lancet.* 1993;342(8878):1032-1036.
- 9. Van Grootven B, Flamaing J, Dierckx de Casterle B, et al. Effectiveness of in-hospital geriatric co-management: a systematic review and meta-analysis. *Age Ageing*. 2017:1-8.
- Deschodt M, Flamaing J, Haentjens P, Boonen S, Milisen K. Impact of geriatric consultation teams on clinical outcome in acute hospitals: a systematic review and meta-analysis. *BMC medicine*. 2013;11:48.
- Kenis C, Bron D, Libert Y, et al. Relevance of a systematic geriatric screening and assessment in older patients with cancer: results of a prospective multicentric study. *Annals of oncology : official journal of the European Society for Medical Oncology / ESMO*. 2013;24(5):1306-1312.
- 12. Puts MT, Santos B, Hardt J, et al. An update on a systematic review of the use of geriatric assessment for older adults in oncology. *Annals of oncology : official journal of the European Society for Medical Oncology*. 2014;25(2):307-315.

- 13. Wildiers H, Heeren P, Puts M, et al. International Society of Geriatric Oncology consensus on geriatric assessment in older patients with cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology*. 2014;32(24):2595-2603.
- Girones R, Torregrosa D, Maestu I, Gomez-Codina J, Tenias JM, Costa RR. Comprehensive Geriatric Assessment (CGA) of elderly lung cancer patients: A single-center experience. *Journal of geriatric oncology*. 2012;3(2):98-103.
- 15. Kanesvaran R, Li H, Koo KN, Poon D. Analysis of prognostic factors of comprehensive geriatric assessment and development of a clinical scoring system in elderly Asian patients with cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology*. 2011;29(27):3620-3627.
- Puts MT, Hardt J, Monette J, Girre V, Springall E, Alibhai SM. Use of geriatric assessment for older adults in the oncology setting: a systematic review. *Journal of the National Cancer Institute*. 2012;104(15):1133-1163.
- Aaldriks AA, Giltay EJ, le Cessie S, et al. Prognostic value of geriatric assessment in older patients with advanced breast cancer receiving chemotherapy. *Breast (Edinburgh, Scotland)*. 2013;22(5):753-760.
- 18. Soubeyran P, Fonck M, Blanc-Bisson C, et al. Predictors of early death risk in older patients treated with first-line chemotherapy for cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology*. 2012;30(15):1829-1834.
- 19. Decoster L, Van Puyvelde K, Mohile S, et al. Screening tools for multidimensional health problems warranting a geriatric assessment in older cancer patients: an update on SIOG recommendationsdagger. Annals of oncology : official journal of the European Society for Medical Oncology / ESMO. 2014.
- 20. Hamaker ME, Jonker JM, de Rooij SE, Vos AG, Smorenburg CH, van Munster BC. Frailty screening methods for predicting outcome of a comprehensive geriatric assessment in elderly patients with cancer: a systematic review. *The Lancet Oncology*. 2012;13(10):e437-444.
- Clegg A, Young J, Iliffe S, Rikkert MO, Rockwood K. Frailty in elderly people. *Lancet*. 2013;381(9868):752-762.
- 22. Kristjansson SR, Ronning B, Hurria A, et al. A comparison of two pre-operative frailty measures in older surgical cancer patients. *Journal of geriatric oncology*. 2012;3(1):1-7.
- Olivieri A, Gini G, Bocci C, et al. Tailored therapy in an unselected population of 91 elderly patients with DLBCL prospectively evaluated using a simplified CGA. *The oncologist*. 2012;17(5):663-672.
- 24. Spina M, Balzarotti M, Uziel L, et al. Modulated chemotherapy according to modified comprehensive geriatric assessment in 100 consecutive elderly patients with diffuse large B-cell lymphoma. *The oncologist*. 2012;17(6):838-846.

- 25. Hamaker ME, Seynaeve C, Wymenga AN, et al. Baseline comprehensive geriatric assessment is associated with toxicity and survival in elderly metastatic breast cancer patients receiving single-agent chemotherapy: Results from the OMEGA study of the Dutch Breast Cancer Trialists' Group. *Breast.* 2013.
- Baitar A, Kenis C, Moor R, et al. Implementation of geriatric assessment-based recommendations in older patients with cancer: A multicentre prospective study. *Journal of geriatric oncology*. 2015;6(5):401-410.
- 27. Bellera CA, Rainfray M, Mathoulin-Pelissier S, et al. Screening older cancer patients: first evaluation of the G-8 geriatric screening tool. *Annals of oncology : official journal of the European Society for Medical Oncology*. 2012;23(8):2166-2172.
- Soubeyran P BC, Gregoire F et al. Validation of a screening test for elderly patients in oncology. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology*. 2008;26(suppl 15S), abstr 20568.
- 29. Braes T, Flamaing J, Sterckx W, et al. Predicting the risk of functional decline in older patients admitted to the hospital: a comparison of three screening instruments. *Age and ageing*. 2009;38(5):600-603.
- 30. Meldon SW, Mion LC, Palmer RM, et al. A brief risk-stratification tool to predict repeat emergency department visits and hospitalizations in older patients discharged from the emergency department. *Academic Emergency Medicine*. 2003;10(3):224-232.
- Oken MM, Creech RH, Tormey DC, et al. Toxicity and Response Criteria of the Eastern-Cooperative-Oncology-Group. *American Journal of Clinical Oncology-Cancer Clinical Trials.* 1982;5(6):649-655.
- 32. Katz S, Ford AB, Moskowitz RW, Jackson BA, Jaffe MW. Studies of Illness in the Aged the Index of Adl - A Standardized Measure of Biological and Psychosocial Function. *Jama-Journal of the American Medical Association*. 1963;185(12):914-919.
- Lawton MP, Brody EM. Assessment of Older People Self-Maintaining and Instrumental Activities of Daily Living. *Gerontologist*. 1969;9(3P1):179-&.
- 34. Lamb SE, Jorstad-Stein EC, Hauer K, Becker C. Development of a common outcome data set for fall injury prevention trials: the Prevention of Falls Network Europe consensus. *Journal of the American Geriatrics Society*. 2005;53(9):1618-1622.
- Avlund K, Holstein BE. Functional ability among elderly people in three service settings: The discriminatory power of a new functional ability scale. *European Journal of Epidemiology*. 1998;14(8):783-790.
- Folstein MF, Folstein SE, Mchugh PR. Mini-Mental State Practical Method for Grading Cognitive State of Patients for Clinician. *Journal of Psychiatric Research*. 1975;12(3):189-198.
- 37. Yesavage JA, Brink TL, Rose TL, et al. Development and Validation of A Geriatric
 Depression Screening Scale A Preliminary-Report. *Journal of Psychiatric Research*.
 1983;17(1):37-49.
- 38. Vellas B, Guigoz Y, Garry PJ, et al. The mini nutritional assessment (MNA) and its use in grading the nutritional state of elderly patients. *Nutrition*. 1999;15(2):116-122.
- Charlson ME, Pompei P, Ales KL, Mackenzie CR. A New Method of Classifying Prognostic Co-Morbidity in Longitudinal-Studies - Development and Validation. *Journal of Chronic Diseases*. 1987;40(5):373-383.
- Sharma M, Loh KP, Nightingale G, Mohile SG, Holmes HM. Polypharmacy and potentially inappropriate medication use in geriatric oncology. *Journal of geriatric oncology*. 2016;7(5):346-353.
- 41. Brouwers B, Dalmasso B, Hatse S, et al. Biological ageing and frailty markers in breast cancer patients. *Aging*. 2015;7(5):319-333.
- 42. Gönen M, Heller G. Concordance probability and dicriminatory power in proportional hazards regression. Biometrika. 2005;92(4):965-970.
- 43. Cook NR. Use and misuse of the receiver operating characteristic curve in risk prediction. Circrulation 2007;115:928-935.

Figure 1: Flow-chart of patient selection



Figure 1: Flow-chart of patient selection. Analyses were conducted on two separate study cohorts: cohort A (n=763) and cohort B (n=402). Abbreviations: UZB = Universitair Ziekenhuis Brussel; UZL = Universitaire Ziekenhuizen Leuven

	Cohort A	A (n=763)	Cohort E	3 (n=402)	
Characteristic	N°	%	N°	%	p-value
Age	76 (7	70-95)	76 (7	70-95)	
Gender					
Male	517	67,8	268	66,7	0,705
Female	246	32,2	134	33,3	
Tumor type					
Breast	372	48,8	169	42.0	0.019
Colorectal	186	24,4	132	32,8	
Lung	72	9,4	40	10,0	
Ovarian	59	7,7	33	8,2	
Prostate	74	9,7	28	7,0	
Oncologic setting					
New diagnosis	486	63,7	287	71,4	0,008
Progression/relapse	277	36,3	115	28,6	
Stage					
I	88	11,5	46	11,4	0,270
II	171	22,4	102	25,4	
III	112	14,7	70	17,4	
IV	392	51,4	184	45 <i>,</i> 8	
Treatment received					
Surgery	345	45,2	216	53,7	0,006
Chemotherapy	312	40,9	182	45 <i>,</i> 3	0,150
Radiotherapy	235	30,8	134	33,3	0,377
Hormonal therapy	297	38,9	140	34,8	0,170
Pain: VAS (0-10)					
Score 0	374	49,1	183	45 <i>,</i> 5	0.473
Score 1-3	118	15,5	70	17,4	
Score ≥4	270	35,4	149	37,1	

Table 1: Patient and clinical characteristics

Abbreviations: CI = confidence interval; VAS = Visual Analogue Scale

*Treatment received = patients may have received a combination of treatment modalities.

Table 2: Results of the screening and geriatric assessment

			Coh (n =	Cohort A (n = 763)		ort B 402)		
Screening tools	Cut-off	Score	n	%	n	%	p-value	
fTRST (0-6)	≥1	Absence of a geriatric risk profile: score 0 Presence of a geriatric risk profile: score >1	131	17.2 82.8	38	9.5 90.6	<.001	
G8 (0-17)	<1/	Absence of a geniatric risk profile: score $> 1/$	197	25.8	122	30.0	0 099	
00(017)	214	Absence of a generic risk profile: score ≤ 14	566	74.2	280	60 7	0.055	
FCOG-PS		Score 0-1	559	74.2	280	74.4	0 681	
		Score 2-4	204	26.7	103	25.6	0.001	
GA components	ltem/ Instrument	Score						
Social data	Living situation	Not living alone	522	68.4	262	65.2	0.262	
	-	Living alone	241	31.6	140	34.8		
Functional Status	ADL (6-24)	Independent: score 6	372	48.8	155	38.6	<.001	
		Dependent: score ≥ 7	391	51.3	247	61.4		
	IADL (0-5/8)	Independent: score 8 (female) or 5 (male)	329	43.1	142	35.3	0.010	
		Dependent: score < 8 (female) or 5 (male)	434	56.9	206	64.7		
	Fall history	No falls	520	68.2	255	63.4	0.105	
		≥ 1 fall	243	31.9	147	36.6		
Fatigue	MOB-T (0-6)	No fatigue	315	41.3	163	40.6	0.808	
		Presence of fatigue	448	58.7	239	59.5		
Cognition	MMSE (0-30)	Score ≥ 24 = normal cognition	684	89.7	331	82.3	<.001	
		Score ≤ 23 = mild/severe cognitive decline	79	10.4	71	17.7		
Depression	GDS-15 (0-15)	Score 0-4 = not at risk for depression	604	79.2	300	74.6	0.078	
		Score 5-15 = at risk for depression	159	20.8	102	25.4		
Nutrition	MNA-SF (0-14)	Score ≥ 12 = normal nutritional status	288	37.8	175	43.5	0.055	
		Score 0-11 = risk of malnutrition + malnourished	475	62.3	227	56.5		
Comorbidity	CCI (0-37)	No comorbidities (score 0)	355	46.5	190	47.3	0.811	
		Comorbidity score ≥ 1	408	53.5	212	52.7		
Polypharmacy		0-4 different drugs	369	48.4	184	45.8	0.400	
		≥ 5 different drugs	394	51.6	218	54.2		
GA summary								
LOFS		Fit: score 9-10	206	27.0	104	25.9	0.730	
		Slightly vulnerable: score 7-8	283	37.1	137	34.1		
		Vulnerable: score 5-6	159	20.8	92	22.9		
		Frail: score 3-4	72	9.5	43	10.7		
		Very frail: score 0-2	43	5.6	26	6.4		

Table 2: Results of the screening and geriatric assessment (Continued)

Abbreviations: GA = Geriatric Assessment; fTRST = Flemish version of the Triage Risk Screening Tool; ECOG PS = Eastern Cooperative Oncology Group Performance Status; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MOB-T = Mobility – Tiredness Test; MMSE = Mini Mental State Examination; GDS = Geriatric Depression Scale; MNA-SF = Mini Nutritional Assessment – Short Form; CCI = Charlson Comorbidity Index; LOFS: Leuven Oncology Frailty Score

	Cohort A (n=763)		Cohort	B (n=402)
	HR	95%CI	HR	95%CI
Gender (reference group: men)	0.55*	0.46-0.67	0.59*	0.45-0.78
Age				
70-74 vs 75-79	0.72*	0.57-0.90	0.70*	0.49-1.00
70-74 vs ≥80	0.59*	0.48-0.74	0.54*	0.39-0.76
75-79 vs ≥80	0.83	0.67-1.03	0.78	0.57-1.06
Tumor type				
Breast vs Colorectal	0.42*	0.33-0.52	0.52*	0.37-0.74
Breast vs Lung	0.18*	0.13-0.24	0.12*	0.08-0.19
Breast vs Ovarian	0.37*	0.27-0.51	0.27*	0.17-0.43
Breast vs Prostate	0.70*	0.50-0.98	0.51*	0.29-0.87
Colorectal vs Lung	0.43*	0.32-0.57	0.23*	0.16-0.35
Colorectalvs Ovarian	0.88	0.64-1.22	0.52*	0.33-0.82
Colorectal vs Prostate	1.68*	1.20-2.35	0.97	0.57-1.64
Lung vs Ovarian	2.08*	1.43-3.00	2.25*	1.36-3.73
Lung vs Prostate	3.95*	2.69-5.79	4.17*	2.34-7.45
Ovarian vs Prostate	1.90*	1.27-2.86	1.85*	1.00-3.42
Stage				
l vs ll	0.97	0.57-1.65	0.57	0.23-1.40
l vs III	0.32*	0.19-0.53	0.29*	0.12-0.70
l vs IV	0.11*	0.07-0.18	0.07*	0.03-0.16
ll vs III	0.33*	0.22-0.49	0.51*	0.29-0.89
II vs IV	0.12*	0.08-0.16	0.13*	0.08-0.19
III vs IV	0.35*	0.27-0.46	0.25*	0.16-0.37

Table 3: Prognostic value for OS of gender and clinical information (univariable analysis)

Abbreviations: HR = Hazard Ratio; CI = confidence limits

* indicates 95% CI below 1 (significant)

Table 4: Prognostic value for OS of screening tools and geriatric information

		Cohort A			Cohort B				
		Univariat	ole analysis	adjusted	analysis**	Univaria	ble analysis	adjusted	analysis**
		HR	95%CI	HR	95%CI	HR	95%CI	HR	95%CI
Screen	ing tools								
	fTRST	0.60*	0.42-0.73	0.62*	0.47-0.82	0.66	0.40-1.10	0.51*	0.31-0.86
	G8	0.33*	0.26-0.43	0.39*	0.30-0.50	0.45*	0.32-0.63	0.78	0.55-1.12
	ECOG-PS 0-1 vs 2-4	0.46*	0.38-0.56	0.58*	0.47-0.70	0.41*	0.31-0.54	0.50*	0.36-0.68
GA con	nponents								
	living situation	0.99	0.81-1.20	0.87	0.71-1.06	1.03	0.78-1.37	0.80	0.60-1.07
	ADL	0.69*	0.57-0.82	0.63*	0.52-0.76	0.75*	0.57-0.99	0.72*	0.53-0.96
	IADL	0.55*	0.46-0.67	0.56*	0.46-0.69	0.55*	0.41-0.75	0.58*	0.42-0.79
	falls	0.81*	0.67-0.98	0.78*	0.64-0.95	0.76*	0.58-1.00	0.72*	0.54-0.95
	MOBT	0.58*	0.48-0.70	0.75*	0.61-0.91	0.60*	0.45-0.80	0.61*	0.45-0.81
	MMSE	0.62*	0.47-0.81	0.67*	0.51-0.89	0.53*	0.39-0.73	0.64*	0.46-0.89
	GDS15	0.61*	0.50-0.76	0.66*	0.53-0.82	0.49*	0.37-0.65	0.71*	0.52-0.95
	MNA-SF	0.42*	0.34-0.51	0.54*	0.44-0.66	0.53*	0.40-0.70	0.61*	0.46-0.83
	CCI	0.78*	0.65-0.94	0.94	0.78-1.13	0.66*	0.50-0.87	0.65*	0.49-0.86
	polypharmacy	0.72*	0.60-0.86	0.70*	0.58-0.85	0.68*	0.52-0.90	0.79	0.60-1.05
GA sun	nmary				'				
LOFS									
	0-2 vs 3-4	0.61*	0.47-0.79	0.51*	0.33-0.79	0.56*	0.37-0.85	0.77	0.59-1.00
	0-2 vs 5-6	0.38*	0.29-0.51	0.40*	0.26-0.61	0.31*	0.20-0.47	0.57*	0.43-0.76
	0-2 vs 7-8	0.29*	0.21-0.41	0.26*	0.15-0.42	0.26*	0.16-0.43	0.41*	0.29-0.59
	0-2 vs 9-10	0.26*	0.18-0.38	0.16*	0.09-0.29	0.20*	0.11-0.35	0.32*	0.22-0.48
	3-4 vs 5-6	0.63*	0.50-0.80	0.77	0.54-1.09	0.55*	0.39-0.78	0.74*	0.58-0.94
	3-4 vs 7-8	0.48*	0.35-0.65	0.51*	0.32-0.79	0.47*	0.31-0.71	0.54*	0.40-0.73
	3-4 vs 9-10	0.43*	0.30-0.61	0.30*	0.18-0.52	0.36*	0.21-0.59	0.42*	0.29-0.60
	5-6 vs 7-8	0.76	0.55-1.04	0.66	0.42-1.03	0.85	0.55-1.29	0.73	0.53-1.00
	5-6 vs 9-10	0.68*	0.48-0.97	0.40*	0.23-0.68	0.64	0.39-1.07	0.57*	0.39-0.82
	7-8 vs 9-10	0.90	0.60-1.35	0.60	0.33-1.09	0.76	0.43-1.34	0.78	0.52-1.17

Table 4: Prognostic value for OS of screening tools and geriatric information (continued)

Abbreviations: HR = Hazard Ratio; CI = confidence interval; ECOG PS = Eastern Cooperative Oncology Group Performance Status; fTRST = Flemish version of the Triage Risk Screening Tool; GA = Geriatric Assessment; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MOB-T = Mobility – Tiredness Test; MMSE = Mini Mental State Examination; GDS = Geriatric Depression Scale; MNA-SF = Mini Nutritional Assessment – Short Form; CCI = Charlson Comorbidity Index; LOFS = Leuven Oncogeriatric Frailty Score

* indicates 95% CI below 1 (significant)

** adjusted for age, stage, and tumor type

Table 5: Comparison of the prognostic performance for OS of extended models including screening tools and geriatric information compared to the baseline model of clinical information.

	C	ohort A (n=76	3)	Cohort B (n=402)			
Baseline model (B)	CPE	AIC	ΔΑΙC	CPE	AIC	ΔΑΙΟ	
Clinical information (age+stage+tumor type)	0.728	5411.57	0	0.750	2179.53	0	
Screening tools				<u>1</u>			
B + fTRST	0.742	5401.09	10.48	0.746	2173.81	5.72	
B + G8	0.747	5354.31	57.26	0.753	2179.63	-0.10	
B + ECOG-PS 0-1 vs 2-4	0.737	5384.59	26.98	0.762	2162.73	16.80	
GA components							
B + living situation	0.729	5411.54	0.03	0.752	2179.37	0.16	
B + ADL	0.726	5390.31	21.26	0.748	2176.37	3.15	
B + IADL	0.754	5379.68	31.89	0.745	2169.53	10.00	
B + falls	0.730	5407.45	4.12	0.752	2176.21	3.32	
B + MOBT	0.724	5405.01	6.56	0.771	2169.69	9.84	
B + MMSE	0.727	5406.65	4.92	0.763	2175.08	4.45	
B + GDS-15	0.729	5400.16	11.41	0.745	2176.37	3.16	
B + MNA-SF	0.756	5377.46	34.11	0.772	2170.75	8.78	
B + CCI	0.727	5413.12	-1.54	0.753	2172.46	7.07	
B + polypharmacy	0.728	5399.72	11.85	0.749	2178.82	0.71	
GA summaries							
B + 10-item GA	0.755	5353.85	57.72	0.775	2155.95	23.58	
B + LOFS	0.745	5373.75	37.82	0.755	2144.71	34.81	

Abbreviations: fTRST = Flemish version of the Triage Risk Screening Tool; ECOG PS = Eastern Cooperative Oncology Group Performance Status; GA = Geriatric Assessment; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MOB-T = Mobility – Tiredness Test; MMSE = Mini Mental State Examination; GDS = Geriatric Depression Scale; MNA-SF = Mini Nutritional Assessment – Short Form; CCI = Charlson Comorbidity Index; LOFS = Leuven Oncogeriatric Frailty Score; CPE = Concordance Probability Estimate; AIC = Akaike Information Criterion Figure 2: CPE values demonstrating the discriminatory ability for OS of screening tools and geriatric information in older patients with



carcinoma.

Figure 2: CPE values demonstrating the discriminatory ability for OS of screening tools and geriatric information in older patients with carcinoma. Results are shown for cohort A and cohort B. The baseline model consisted of age, tumor type, and stage as categorical variables. Abbreviations: CPE = Concordance Probability Estimate; fTRST = Flemish version of the Triage Risk Screening Tool; ECOG PS = Eastern Cooperative Oncology Group Performance Status; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MOB-T = Mobility - Tiredness Test; MMSE = Mini Mental State Examination; GDS = Geriatric Depression Scale; MNA-SF = Mini Nutritional Assessment - Short Form; CCI = Charlson Comorbidity Index; GA = Geriatric Assessment; LOFS = Leuven Oncogeriatric Frailty Score.

Chapter 6.

The prognostic value of three commonly measured blood parameters and geriatric assessment to predict overall survival in addition to clinical information in older patients with cancer

Baitar A*, Kenis C*, Decoster L, De Grève J, Lobelle JP, Flamaing J, Milisen K, Wildiers H

This chapter has been accepted for publication in Cancer.

* equal contribution

The online version of the published paper will contain supplementary material which have not been included in this dissertation.

Abstract

Aim

To evaluate the prognostic value of laboratory parameters and geriatric assessment (GA) in addition to a baseline model with clinical information for overall survival (OS) in patients with cancer.

Patients and Methods

A GA was systematically performed in patients \geq 70 years. Our baseline model consisted of age, tumor type, and stage. The incremental prognostic value of the GA as a whole (=10-item GA) and laboratory parameters was assessed separately and combined. The parameters included: hemoglobin (Hb), albumin, C-reactive-protein (CRP), the Glasgow Prognostic Score (GPS). Analyses were conducted with continuous and dichotomized variables. Cox models were compared based on Akaike Information Criterion (Δ AIC) and their discriminatory ability was assessed using the Concordance Probability Estimate (CPE).

Results

A total of 328 patients were considered for this analysis. The baseline model had a CPE of 0.725. The addition of CRP, albumin, and Hb combined resulted in the best performing model (Δ AIC: 40.12; CPE: 0.757) among the laboratory parameters. However, the 10-item GA improved the baseline model even more (Δ AIC: 46.03; CPE: 0.769). Similar results were observed in the analysis with dichotomous variables. The addition of the three laboratory parameters (CRP, albumin, Hb) improved the CPE with 1.4% compared to the baseline model already extended with the 10-item GA. The CPE increase (1.7%) was the highest with the GPS in the analysis with dichotomous variables.

Conclusions

GA adds slightly more prognostic information than laboratory parameters besides clinical information. The laboratory parameters have an additional prognostic value beyond clinical and geriatric information.

Introduction

The implementation of geriatric screening and geriatric assessment (GA) in daily oncology practice is feasible and relevant¹. To obtain a better view of the global health status of the patient, GA includes the assessment of social needs, functional status (FS), co-morbidity and medication use, cognitive and psychological status, and nutritional status². One of the aims of the GA-based approach is to improve the estimation of life-expectancy which is sometimes a difficult task though very important in the treatment decision-making process. Quick online tools are available that can help estimate the patient's life expectancy in the general population³. However, these tools are not entirely suitable in geriatric oncology. In our recent work, we showed that geriatric screening and GA provide additional prognostic information for overall survival (OS) besides clinical information in older patients with a solid tumor⁴. While GA is currently the best tool to estimate a patient's degree of frailty, many efforts are made to identify suitable biomarkers to measure biological age or frailty and to evaluate their predictive value for clinical outcomes like mortality⁵⁻⁶. Although many 'aging biomarkers' for biological age, have been proposed in geriatric oncology, none have presently reached a sufficient evidence-base for use in routine clinical practice. Given the complexity of the aging process, it is likely that the best approach would be for the future to work with a combination of geriatric information and biomarkers besides clinical information to aid treatment decision making. Commonly measured laboratory blood values, not necessarily aging biomarkers, might also be interesting for this purpose. A study conducted in 85-yearold persons from the general population showed that a profile of 7 routine blood measurements predicted 5-year mortality as accurately as the known predictors gait speed and instrumental activities of daily living (IADL)⁷. For patients with cancer, various studies have identified several routinely measured blood values at diagnosis as prognostic factors for OS. In a recent study for example, a score was developed to improve the estimation of survival in patients with newly diagnosed brain metastases based solely on standard clinical blood values⁸. In this study, we will focus on hemoglobin (Hb), albumin, and C-reactive protein (CRP). Their prognostic value for OS has been shown in a range of tumor types independent from other prognostic factors⁹⁻¹¹. Simple inflammatory-based scores have also received much interest in the assessment of prognosis over the past recent years. Three composite scores that combine CRP and albumin values will be included in this study: the Glasgow Prognostic Score (GPS)¹², the modified GPS (mGPS)¹³, and the CRP/albumin ratio. Their prognostic value for mortality has already been shown in multiple studies with various types of cancer in different settings¹⁴⁻¹⁶. The aim of the present paper is to compare the added prognostic value of the aforementioned laboratory parameters (Hb, albumin, CRP) and the composite scores (GPS, mGPS, CRP/albumin ratio) with the added prognostic value of GA and to evaluate their combined prognostic value for OS besides clinical information in patients with cancer.

Patients and methods

Design

This study was part of a prospective, multicenter, observational study with 763 patients with a solid tumor evaluating the added prognostic value of screening, individual GA components, and two GA summaries for OS to a baseline model of clinical information⁴. This study has been reported extensively previously. In short, the GA-based approach was implemented in routine oncology practice in two academic hospitals ('Universitair Ziekenhuis Brussel' and 'Universitaire Ziekenhuizen Leuven') in Belgium between October 2009 and July 2011. Patients \geq 70 years were included during a hospital visit at diagnosis or at disease progression/relapse, when a treatment decision had to be made. Inclusion was limited to the following 5 tumor types: breast, colorectal, ovarian, lung and prostate cancer. All data were collected by trained health care workers in both centers¹². The study was approved by the Ethical Committee of both participating hospitals.

Laboratory parameters

Data on serum Hb, serum albumin, and CRP were used from routinely collected blood samples shortly before or after the date of GA. The three laboratory parameters were categorized for descriptive statistics. Local lab ranges were used from both participating hospitals to dichotomize the laboratory values. Since these did not correspond, different cut-off values were used for each hospital. Hb was dichotomized in below lower limit of normal (<LLN) and ≥LLN. The LLN was 12.0 g/dl or 11.8 g/dl dependent of which hospital. Since the local lab ranges did not distinguish cut-off values for men and women, we used the 1968 WHO criteria (men: Hb < 12.0 g/dl; women: Hb <11.8 g/dl) but also

alternative cut-off values proposed by Beutler E and Waalen J. 2006^{18} for older Caucasian adults (men: Hb < 13.2 g/dl; women: Hb < 12.2 g/dl) to evaluate if gender-specific cut-off values would improve the prognostic value of Hb.

Albumin was dichotomized in <LLN and \geq LLN with a LLN of 35.0 g/l or 37.0 g/l. CRP as a continuous variable was analyzed per 10-point increase to allow a more practical interpretation of the results. CRP was also categorized in elevated values with cut-off value >5.0 mg/l or \geq 5.0 mg/l depending which hospital, and normal values.

The inflammatory-based composite score GPS (score range: 0-2) was defined as follows: score 0: CRP $\leq 10 \text{ mg/dl}$ and albumin $\geq 35.0 \text{ g/l}$; score 1: CRP > 10 mg/l or albumin < 35.0 g/l; score 2: CRP > 10 mg/l and albumin $< 35.0 \text{ g/l}^{12}$. The only difference between the GPS and the mGPS is that the mGPS does not consider hypoalbuminemia (< 35.0 g/l) in the allocation of score 1¹³.

Geriatric assessment

The GA included 10 components based on the 2014 SIOG recommendation guideline for GA ²: social data with living situation, FS assessed by the Katz's Activities of Daily Living (ADL)¹⁹ and by Lawton's Instrumental Activities of Daily Living (IADL)²⁰, fall history during the last year²¹, fatigue assessed by the MOB-T²², mental status assessed by the Mini Mental State Examination (MMSE)²³ and Geriatric Depression Scale (GDS-15)²⁴, nutritional status assessed by the Mini Nutritional Assessment-Short Form (MNA-SF)²⁵, comorbidities assessed by the Charlson Comorbidity Index (CCI)²⁶ and a polypharmacy assessment²⁷. When taken together, this 10-item GA is a GA summary, which integrates the results of every GA component mentioned above and thus captures the multidimensional GA as a whole evaluating the global health of the older patient with cancer. However, it doesn't calculate a global GA end score. In our other paper, we showed that the 10-item GA provides more prognostic information than the individual GA components and screening tools because 'GA summaries' are likely better at capturing the multidimensional process of aging⁴. Therefore, in this paper, we primarily compared the added prognostic value of the 10-item GA with that of laboratory parameters. However, for comparative reasons we also included the individual GA components, the 'Leuven Oncogeriatric Frailty Score' (LOFS) (also a GA summary), the G8, the Flemish version of the Triage Risk Screening Tool (fTRST),

and the Eastern Cooperative Oncology Group - Performance Status (ECOG-PS) in the current analysis. The G8 (score range: 0-17, cut-off \leq 14) and the fTRST (score range: 0-6, cut-off \geq 1) are geriatric screening tools to identify patients who would benefit from a GA²⁸⁻³¹. The ECOG-PS is frequently used in oncology to classify the performance status of patients³². The LOFS is a recently developed tool to summarize clinical frailty^{4,6}. It integrates information from 5 GA components into a single global end score (range: 0-10). The following categories are included: score 0-2 (severely frail), score 3-4 (frail), score 5-6 (vulnerable), score 7-8 (slightly vulnerable) and score 9-10 (fit).

Statistical analysis

Patient and clinical characteristics, screening and GA results were compared between the groups of patients included and excluded from the main analysis by means of the chi-square test and the Mann-Whitney test as appropriate. The median survival time of the two groups was compared with the log-rank test.

Univariable cox proportional hazard regression analyses were conducted to evaluate the prognostic value of laboratory parameters, screening tools, individual GA components, and the LOFS. Multivariable analyses were performed as well to adjust for clinical information (age, stage, tumor type). OS was calculated from date of GA to date of death from any cause or last follow-up for censored patients. Results were considered significant if p<0.05.

We performed Cox proportional hazard regression analyses to build multiple prognostic models. The baseline clinical model consisted of age as a continuous variable and tumor type and stage as categorical variables. To assess the contribution of the laboratory parameters, the 10-item GA and the two combined, each was added separately to the baseline model of clinical information. The goodness-of-fit of the nested models were compared with the likelihood-ratio test. We added the individual GA components, the screening tools, and the LOFS as well in the analysis. Multicollinearity was assessed with the variance inflation factor (VIF) and was not present (VIF's<3). The cox.zph function revealed that the proportional hazards assumption was not satisfied for age, albumin as a continuous variable, CRP and Hb as a dichotomous variable, and the (m)GPS in some of the multivariable models. We

decided not to modify these models, since our objective was to compare the performance of the examined models and not to adjust the construction of the models to get a better fit.

The discriminatory ability of each prognostic model was assessed using the Concordance Probability Estimate (CPE)³³. The global fit of the Cox models was compared by means of the difference of Akaike Information Criterion (AIC) of the baseline model and the extended model (Δ AIC). A higher Δ AIC value corresponds with a more explanatory and informative model.

Separate analyses were conducted with the GA components and laboratory parameters as continuous variables and as dichotomized variables where possible. Because the baseline model was identical in both analyses, we were able to quantify the expected loss of information and thus loss in prediction due to the dichotomization of the data. The LOFS was analyzed as a categorical variable and a continuous variable (per 1-point increase). The (m)GPS (score range: 0-2) was analyzed as a categorical variable but also as a continuous variable when calculating the hazard ratios in order to have the risk per unit increase as well. All analyses were performed using SPSS 23 (Chicago, IL) and R version 3.3.3.

Results

Patient population

For more information on the total sample we refer to our previous publication (cohort A, n=763)⁴. For the current paper, a subgroup of 328 patients with information available on the three laboratory parameters was considered. The median age was 77 years (range: 70-95 years). The most prevalent tumors were from the breast (38.4%), colorectal (35.4%), lung (15.5%), and prostate (6.4%). At the moment of inclusion in the study, 63.7% had a newly diagnosed cancer, whereas 36.3% had disease progression or relapse. Considering comorbidity, 54.9% had a score of ≥ 1 on the CCI. More details of patient demographics and clinical characteristics, including patients that were excluded from the analysis, are listed in Table 1. The results of the GA are listed in Table 2. The studied subgroup was more often male, had a different pattern of tumor types, had more advanced disease, received less surgery but more often chemotherapy and radiotherapy than the group of patients excluded from the analysis (n=435). The GA revealed that the studied subgroup had more often a geriatric risk profile according to the G8 and was more often depressed. Furthermore, the survival was significantly lower with a median survival time of 21.3 months (95%CI: 18.1-29.3; 239 deaths (72.9%)) compared to a median survival time of 50.7 months (95%CI: 40.2-65.4; 232 deaths (53.3%); log-rank test: p<0.001). The median follow-up, calculated with the reversed Kaplan-Meier method³⁴, was 60.3 months (95%CI: 58.6-62.6) and 62.6 months (95%CI: 61.4-64.3), respectively.

The prognostic value of laboratory parameters

Blood samples were on average 7.7 days (SD 10.0) collected from the date on which a GA was performed. In total, 58.8% (n=193) of the patients had an elevated CRP level; 28.4% (n=93) of the patients had an albumin level <LLN; and 41.8% (n=137) of the patients had a Hb level <LLN. For the GPS, 169 patients (51.5%) were allocated score 0, 93 patients (28.4%) score 1, and 66 patients (20.1%) score 2. The results according to the mGPS were as follows: score 0: 178 patients (54.3%); score 1: 84 patients (25.6%); and score 2: 66 patients (20.1%).

Kaplan-Meier plots from the three individual laboratory parameters and the GPS are shown in Figure 1. The prognostic values of the laboratory parameters (CRP, albumin, Hb) and the composite scores (GPS, mGPS, CRP/albumin) are shown in Table 3. Every laboratory-based parameter (continuous, dichotomized, categorical) was prognostic for OS in univariable and adjusted analyses. Results for the individual GA components, screening tools and LOFS are also included in Table 3.

We conducted additional analyses for Hb taken into account gender-specific cut-off values. This resulted in an additional number of 30 patients and 41 patients who were categorized as having abnormal Hb values according to the WHO criteria and the alternative cut-off values proposed by Beutler E and Waalen J. 2006, respectively. Their estimated hazard ratios are included in Table 3.

Comparison of prognostic models

The baseline model had a good discriminatory ability with CPE 0.725. The added value of the laboratory parameters and GA are shown in Table 4. Based on CPE and AIC results, albumin performed better than CRP in the analyses with continuous and dichotomous variables. Both parameters clearly performed far better than Hb. The GPS (Δ AIC: 32.54; CPE: 0.750) had the best performance compared to the 2 other composite scores (mGPS, CRP/albumin ratio), though not better than albumin as a continuous variable

(Δ AIC: 34.22; CPE: 0.755) or when all three continuous laboratory parameters (CRP, albumin, Hb) were combined which resulted in the best performing model among the models extended with laboratory parameters (Δ AIC: 40.12; CPE: 0.757). However, this model (with CRP, albumin, Hb) did not improve the baseline model as much as the model extended with the 10-item GA (Δ AIC: 46.03; CPE: 0.769). The model with the three laboratory parameters (Δ AIC: 25.49; CPE: 0.747) or any other model laboratory parameters did also not outperform the model with the 10-item GA (Δ AIC: 37.52; CPE: 0.763) when analyzed with dichotomous variables.

Results for the individual GA components, the LOFS, and the screening tools (G8, fTRST, ECOG-PS) were included in Table 4 for comparison. None of these models performed better than the best performing models with laboratory parameters. When comparing the individual GA components with the laboratory parameters, it is important to note that a model might have a better discriminatory ability (CPE) than another while the prognostic information (Δ AIC) contained in that model might be lower. Nutritional status measured by the MNA-SF as a continuous variable for example added more prognostic information than albumin (Δ AIC: 37.91 vs Δ AIC: 34.22) while the MNA-SF model had a lower CPE (CPE: 0.749 vs CPE: 0.755).

The analyses that evaluated the laboratory parameters in addition to the baseline model already extended with the 10-item GA show that the laboratory parameters have an added prognostic value beyond clinical and geriatric information. Results are summarized in Table 4. We added the GPS and the mGPS, which are categorical variables, in the two analyses with the 10-item GA analyzed with continuous and dichotomous variables. In order of the Δ AIC results, the following models had the best performance when analyzed with continuous variables: the models with CRP (Δ AIC: 68.03; CPE: 0.780, the three laboratory parameters combined (Δ AIC:67.18; CPE: 0.783), CRP/albumin ratio (Δ AIC: 66.99; CPE: 0.780), and the GPS (Δ AIC: 66.22; CPE: 0.782). The following ranking was observed in the analyses with dichotomous variables: the models with the GPS (Δ AIC: 63.58; CPE: 0.780), the mGPS (Δ AIC: 58.48; CPE: 0.778), the three laboratory parameters combined (Δ AIC: 49.12; CPE: 0.773), and CRP (Δ AIC: 48.28; CPE: 0.771).

A visual overview of the improved performance of the models extended with the three laboratory parameters (CRP, albumin, Hb) and the 10-item GA can be found in Figure 2. The addition of every laboratory parameter improved the goodness-of-fit of each model, with the exception of Hb as a dichotomous variable beyond the model already extended with the 10-item GA. The small improvement with Hb was only significant when it was dichotomized according to the gender-specific cut-off values proposed by Beutler E and Waalen J. 2006 (likelihood-ratio test: p=0.04).

Discussion

CRP, albumin, and Hb in cancer patients

CRP levels increase in response to inflammation and are elevated in a variety of illnesses, including cancer. Inflammation markers like CRP are also potential aging biomarkers since low-level increases in general/non-specific inflammation accompany aging (inflammaging)⁵. A prognostic value of CRP for mortality in older patients with cancer might thus be explained by processes related to the cancer and/or aging. The two other commonly measured laboratory parameters studied here, albumin and Hb, should be viewed less in the context of aging and more in the context of underlying diseases like cancer. Serum albumin is generally used to assess the nutritional status. However, both inflammation and malnutrition reduce albumin levels by decreasing its rate of synthesis¹⁰. Lower Hb levels or anemia is common in patients with cancer and can develop from multiple causes. Also, when related to cancer itself, multiple mechanisms can interfere with normal erythrocyte production but is usually secondary to an imbalance of cytokines²⁹. In this study, anemia is unlikely related to cancer treatment since blood samples were used around the time of a treatment decision.

Irrespective of the cause of changes in levels of these three commonly measured laboratory measures, this study confirms that CRP, albumin, Hb, and the composite scores based on CRP and albumin (GPS, mGPS, CRP/albumin ratio) are prognostic factors for OS independent from clinical information (age, stage, tumor type) in older patients with cancer.

The comparison of the added prognostic value of laboratory parameters and GA

We previously showed that most individual GA components are prognostic factors independent from clinical information for OS⁴. Studies that evaluate the prognostic value of GA generally focus on the most prognostic individual GA components or on the best set of GA components³⁶⁻⁴⁰. In our previous analysis, we took a more extensive approach and also studied the added prognostic value of the GA as a whole (the 10-item GA) by comparing the performance of a baseline model of clinical information with models extended with geriatric information⁴. This allowed us for example to quantify the superior prognostic value of the GA as a whole compared to the individual GA components in addition to clinical information.

A comparison of the added prognostic value of the laboratory parameters and the individual GA components in the current study is sometimes difficult since the comparison varies whether you look at Δ AIC or CPE results and whether you look at the analysis with continuous or dichotomous variables. On the other hand, sometimes results were easy to interpret. From the individual GA components, only nutritional status (MNA-SF) rivaled the performance of that of the best performing laboratory parameters in the analysis with continuous variables. Furthermore, as expected, the dichotomization of the laboratory parameters and the GA components significantly reduced the performance of the models. This is important to note since many studies dichotomize data and because many clinicians are used to work with categorized information. PS (ADL, IADL, ECOG-PS) is an important domain in the assessment of the patients' global health status. IADL was one of the most prognostic GA components in our analysis of the full sample (n=763, ref 4). However, PS did not perform as well in this subgroup, regardless of the data type. A possible explanation might be that PS is more driven by tumor type and stage compared to other GA components and that the more advanced disease (and different pattern of tumor types) observed in this subgroup might have reduced the independent prognostic impact of PS while other GA components that reflect other aspects of the patients' global health status were less influenced. In this regard, we note that the inflammation-based composite scores, which may also be regarded as driven by tumor characteristics, performed better than PS. The comparison of the composite scores GPS and mGPS with the other variables should be interpreted carefully. The (m)GPS is a score with three levels and was added unchanged in the analyses with continuous and dichotomous variables. This explains why the GPS performed in the same range as albumin and CRP in the analysis with continuous variables while it clearly performed better than albumin and CRP in the analysis with dichotomous variables. In the analysis with dichotomous variables, we also conducted additional analyses with Hb since the threshold defining anemia has received considerable scientific debate. Various cut-off values are used in the literature. In this study, we considered two alternative gender-specific cut-off values to examine if this would improve the prognostic value of Hb. A small improvement was indeed observed in the stratification of patients, however the added prognostic value of Hb was throughout the study inferior to the other laboratory parameters.

The comparison of the laboratory parameters with the 10-item GA shows that the latter is superior based on Δ AIC and CPE results. The AIC is a more sensitive method than the CPE to rank different models, however the CPE provides an interpretable measure of performance. According to the CPE results, the addition of the 10-item GA increased the performance of the baseline model with 4.4% whereas the best performing model with laboratory parameters (i.e., with CRP, albumin, Hb) resulted in a lower increase of 3.2% in the analysis with continuous variables. In the analysis with dichotomous variables, the GPS added more than the three laboratory parameters combined, i.e. 2.5% vs 2.2%. However, the 10-item GA added more with a CPE increase of 3.9%. In other words, the model with the 10-item GA performs better than the best model incorporating laboratory parameters with a CPE difference of 1.2% or 1.4%, considering the data type. Obviously, the laboratory parameters have the advantage of being objective, reproducible, inexpensive, and fast whereas the implementation of GA in routine practice requires the necessary time and resources. However, we emphasize that the GA-based approach is also important for other reasons in this context like to estimate the risk of treatment-induced toxicity, to detect previously unknown health problems, and to allow directed geriatric interventions that can improve therapy compliance and outcome.

The incremental prognostic value of laboratory parameters beyond GA

Our findings show that laboratory parameters and GA improve the estimation of life expectancy separately. However, given the complexity of the processes related to aging, cancer, and their

interaction, the combination of different types of information is likely to improve the accuracy of prognosis assessment. Very few studies have evaluated the prognostic value of both GA and biomarkers for OS. Aaldriks et al. 2015 studied patients with non-Hodgkin lymphoma by looking at the hazard ratios after entering individual GA components and biomarkers in a multivariable model⁴¹. A recent study from Honecker et al. 2017 evaluated the simultaneous impact of GA and biomarkers in patients with breast cancer by selecting the best set of factors through elimination starting from a multivariable model incorporating all candidate parameters (backward stepwise analysis)⁴². In this study, we used another analysis approach since we were interested in the incremental value of biomarkers for OS. To the best of our knowledge, this is the first study to compare the incremental value of different biomarkers beyond the GA as a whole. Our analysis showed that the laboratory parameters continue to add prognostic information beyond the model with both clinical information and the 10-item GA. As previously noted, differences were sometimes small, and the ranking of the models depends of AIC and CPE results. According to the more sensitive AIC results, CRP added the most prognostic information. However, the CPE increased the most when the combination of the three laboratory parameters was added resulting in CPE increase of 5.8% relative to the baseline model and 1.4% relative to the baseline model already extended with the 10-item GA. In the analysis with dichotomous variables, it was the GPS that added the most in terms of both AIC and CPE results. The CPE increased with 5.6% relative to the baseline model and with 1.7% relative to baseline model extended with the 10-item GA.

In theory there exists some overlap in prognostic information provided by geriatric information and biomarkers (e.g. nutritional status and albumin). Honecker et al. 2017 suggested there was almost no overlap based on the variables retained in their final model for OS. In this study, we were able to investigate this more accurately when we put the analyses where the baseline model is extended with the laboratory parameters and the 10-item GA separately together with the analyses where both are added simultaneously. To illustrate this with numbers from the analysis with continuous variables. The model extended with the three laboratory parameters combined (CRP, albumin, Hb) improved the CPE with 3.2% (Δ AIC: 40.12) and when extended with the 10-item GA with 4.4% (Δ AIC: 46.03). One might expect an CPE increase of 7.6% (Δ AIC: 86.15) when the baseline model is extended with both.

However, when both the three laboratory parameters and the 10-item GA are actually added to the baseline model, the CPE only increased with 5.8% (Δ AIC: 67.18). Still, as quantified in this study, the combination of geriatric information and biomarkers improves the estimation of life expectancy more than separately.

Strengths and limitations

The strengths of this study include the relatively large sample size, the prospective study design, the many domains covered by the GA, the analysis approach and the long follow-up. However, some considerations need to be taken into account when interpreting the results. Ideally, 'aging' biomarkers should not be influenced by underlying health conditions and the timing of the blood collections is another possible confounding factor limiting their use in routine clinical practice⁵. In this study, this pertains to CRP because CRP levels might be increased due to inflammatory reactions caused by the cancer or another underlying health condition but also due to recent surgery for example. However, we did work with blood samples that were taken at a fixed time, i.e. shortly before or after GA, and hence at the time of diagnosis or progression/relapse. Future studies with a more homogenous population and more detailed information on treatment should make a distinction between these two settings, given the hypothesized pro-aging effects of many cancer treatments.

Our findings are applicable to a population for which all three studied laboratory parameters are evaluated in routine practice. As our results show, this subgroup is as expected more vulnerable and has a worse survival compared to the group of patients for who all three laboratory parameters were not tested. Patients were often excluded from our analysis because the combination CRP and albumin was not tested (reducing the sample size to n=329). However, additional analyses considering one laboratory parameter and thus larger sample sizes due to less missing data (CRP: n=501; albumin: n=423; Hb: n=692), did not change our conclusions which suggests that our results are also relevant for patients for who generally all three laboratory parameters are not tested (Additional file 1).

Furthermore, our heterogeneous population is a disadvantage since many relevant prognostic parameters are not captured but at the same time it is an advantage as it is representative of a larger cancer population.

92

Conclusions

GA adds slightly more prognostic information than laboratory parameters besides clinical information. The laboratory parameters have an additional prognostic value beyond clinical and geriatric information.

Author contributions

conception and design: CK, LD, JPL, JF, KM, HW data acquisition: CK, LD, HW data analysis and interpretation: AB, CK, LD, JDG, JPL, JF, KM, HW writing–initial draft: AB, CK, LD, JPL, HW writing–critical revision: AB, CK, LD, JDG, JPL, JF, KM, HW statistical analyses: AB supervision: AB, CK, LD, JDG, JPL, JF, KM, HW

References

1) Kenis C, Bron D, Libert Y, et al. Relevance of a systematic geriatric screening and assessment in older patients with cancer: results of a prospective multicentric study. Ann Oncol. 2013;24(5):1306-12.

2) Wildiers H, Heeren P, Puts M, et al. International Society of Geriatric Oncology consensus on geriatric assessment in older patients with cancer. J Clin Oncol. 2014;32(24):2595-603.

3) www.eprognosis.ucf.org

4) Kenis C, Baitar A, Heeren L, et al. The added value of geriatric screening and assessment to predict overall survival in older patients with cancer. Cancer. submitted

5) Pallis AG, Hatse S, Brouwers B, et al. Evaluating the physiological reserves of older patients with cancer: the value of potential biomarkers of aging? J Geriatr Oncol. 2014;5(2):204-18.

6) Brouwers B, Dalmasso B, Hatse S, et al. Biological ageing and frailty markers in breast cancer patients. Aging (Albany NY). 2015;7(5):319-33.

7) van Houwelingen AH, den Elzen WP, Mooijaart SP, et al. Predictive value of a profile of routine blood measurements on mortality in older persons in the general population: the Leiden 85-plus study. PloS One. 2013;8(3):e58050.

8) Berghoff AS, Wolpert F, Holland-Letz T, et al. Combining standard clinical blood values for improving survival prediction in patients with newly diagnosed brain metastases-development and validation of the LabBM score. Neuro Oncol. 2017;19(9):1255-1262.

9) Caro JJ, Salas M, Ward A, Goss G. Anemia as an independent prognostic factor for survival in patients with cancer: a systematic, quantitative review. Cancer. 2001;91(12):2214-21.

10) Gupta D, Lis CG. Pretreatment serum albumin as a predictor of cancer survival: a systematic review of the epidemiological literature. Nutr J. 2010;9:69.

11) Roxburgh CS, McMillan DC. Role of systemic inflammatory response in predicting survival in patients with primary operable cancer. Future Oncol. 2010;6(1):149-63.

12) Forrest LM, McMillan DC, McArdl CS, Angerson WJ, Dunlop DJ. Evaluation of cumulative prognostic scores based on the systemic inflammatory response in patients with inoperable non-small-cell lung cancer. Br J Cancer. 2003;89(6):1028-1030.

13) McMillan DC, Crozier JE, Canna K, Angerson WJ, McArdle CS. Evaluation of an inflammationbased prognostic score (GPS) in patients undergoing resection for colon and rectal cancer. Int J Colorectal Dis. 2007;22(8):881-6.16.

14) McMillan DC. The systemic inflammation-based Glasgow Prognostic Score: a decade of experience in patients with cancer. Cancer Treat Rev. 2013;39(5):534-40.

15) Li N, Tian GW, Wang Y, Zhang H, Wang ZH, Li G. Prognostic role of the pretreatment C-Reactive Protein/Albumin ratio in solid cancer: a meta-analysis. Sci Rep. 2017;7:41298.

16) Xu HJ, Ma Y, Deng F, Ju WB, Sun XY, Wang H. The prognostic value of C-reactive protein/albumin ratio in human malignancies: an updated meta-analysis. Onco Targets Ther. 2017;10:3059-3070.

17) Baitar A, Kenis C, Moor R, et al. Implementation of geriatric assessment-based recommendations in older patients with cancer: a multicentre prospective study. J Geriatr Oncol. 2015;6(5):401-10.

18) Beutler E, Waalen J. The definition of anemia: what is the lower limit of normal of the blood haemoglobin concentration? Blood. 2006;107(5):1747-1750.

19) Katz S, Ford AB, Moskowitz RW, Jackson BA, Jaffe MW. Studies of Illness in the Aged. The Index of Adl: A Standardized Measure of Biological and Psychosocial Function. Jama. 1963;185:914-9.

20) Lawton MP, Brody EM. Assessment of older People: self-maintaining and instrumental activities of daily living. Gerontologist. 1969;9(3):179-86.

21) Lamb SE, Jorstad-Stein EC, Hauer K, Becker C. Development of a common outcome data set for fall injury prevention trials: the Prevention of Falls Network Europe consensus. J Am Geriatr Soc. 2005;53(9):1618-22.

22) Avlund K, Holstein BE. Functional ability among elderly people in three service settings: The discriminatory power of a new functional ability scale. Eur J Epidemiol. 1998;14(8):783-90.

23) Folstein MF, Folstein SE, Mchugh PR. "Mini-Mental State". A practical method for grading cognitive state of patients for the clinician. J Psychiatr Res. 1975;12(3):189-98.

24) Yesavage JA, Brink TL, Rose TL, et al. Development and Validation of a geriatric depression screening Scale: a preliminary report. J Psychiatr Res. 1982-1983;17(1):37-49.

25) Vellas B, Guigoz Y, Garry PJ, et al. The Mini Nutritional Assessment (MNA) and its use in grading the nutritional state of elderly patients. Nutrition. 1999;15(2):116-22.

26) Charlson ME, Pompei P, Ales KL, Mackenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis. 1987;40(5):373-83.

27) Sharma M, Loh KP, Nightingale G, Mohile SG, Holmes HM. Polypharmacy and potentially inappropriate medication use in geriatric oncology. J Geriatr Oncol. 2016;7(5):346-53.

28) Baitar A, Van Fraeyenhove F, Vandebroek A, et al. Evaluation of the Groningen Frailty Indicator and the G8 questionnaire as screening tools for frailty in older patients with cancer. J Geriatr Oncol. 2013;4(1):32-8.

29) Kenis C, Decoster L, Van Puyvelde K, et al. Performance of two geriatric screening tools in older patients with cancer. J Clin Oncol. 2014;32(1):19-26.

30) Braes T, Flamaing J, Sterckx W, et al. Predicting the risk of functional decline in older patients admitted to the hospital: a comparison of three screening instruments. *Age Ageing*. 2009;38(5):600-3.

31) Meldon SW, Mion LC, Palmer RM, et al. A brief risk-stratification tool to predict repeat emergency department visits and hospitalizations in older patients discharged from the emergency department. Acad Emerg Med. 2003;10(3):224-32.

32) Oken MM, Creech RH, Tormey DC, et al. Toxicity and response criteria of the Eastern Cooperative Oncology Group. Am J Clin Oncol. 1982;5(6):649-55.

33) Gönen M, Heller G. Concordance Probability and Discriminatory Power in Proportional Hazards Regression. Biometrika. 2005;92(4):965-970.

34) Schemper M, Smith TL. A note on quantifying follow-up in studies of failure time. Control Clin Trials. 1996;17(4):343-6.

35) Dicato M. Anemia in cancer: some pathophysiological aspects. Oncologist. 2003;8 Suppl 1:19-21.

36) Puts et al. Use of geriatric assessment for older adults in the oncology setting: a systematic review.J Natl Cancer Inst. 2012:104(15):1133-63.

37) Hamaker ME, Vos AG, Smorenburg CH, de Rooij SE, van Munster BC. The value of geriatric assessments in predicting treatment tolerance and all-cause mortality in older patients with cancer. Oncologist. 2012;17(11):1439-49.

38) Puts MT, Santos B, Hardt J, et al. An update on a systematic review on the use of geriatric assessment for older adults in oncology. Ann Oncol. 2014;25(2):307-15.

39) Caillet et al. Optimal management of elderly cancer patients: usefulness of the Comprehensive Geriatric Assessment. Clin Interv Aging. 2014;9:1645-60.

40) Versteeg KS, Konings IR, Lagaay AM, van de Loosdrecht AA, Verheul HM. Prediction of treatment-related toxicity and outcome with geriatric assessment in elderly patients with solid malignancies treated with chemotherapy: a systematic review. Ann Oncol. 2014;25(10):1914-8.

41) Aaldriks AA, Giltay EJ, Nortier JW, van der Geest LG, Tanis BC, Ypma P, et al. Prognostic significance of geriatric assessment in combination with laboratory parameters in elderly patients with aggressive non-Hodgkin lymphoma. Leuk Lymphoma. 2015;56(4):927-35.

42) Honecker F, Harbeck N, Schnabel C, Wedding U, Waldenmaier D, Saupe S, et al. Geriatric assessment and biomarkers in patients with metastatic breast cancer receiving first-line mono-chemotherapy: results from the randomized phase III pelican trial. J Geriatr Oncol. 2017; S1879-4068(17)30195-9.

97



Figure 1. Prognostic value of laboratory parameters (n=328). Kaplan-Meier overall survival curves are shown stratified for normal and abnormal values of hemoglobin, albumin, and CRP. The Glasgow Prognostic Score (GPS) is shown as well.



Figure 2. Overview of the added performance of the extended models with laboratory parameters and GA. Overview of (A) the discriminatory ability (CPE) and (B) the model fit (Δ AIC) of the baseline model of clinical information and the models extended with laboratory parameters and GA. Models are shown analysed with continuous variables and dichotomized variables. The baseline model was analysed with age as a continuous variable; and stage and tumor type as categorical variables. Abbreviations. GA = Geriatric Assessment; CPE = Concordance Probability Estimate; AIC = Akaike Information Criterion; Hb = Hemoglobin

	Included		Excluded		
Characteristic	N°	%	N°	%	p-value
Total	328	100	435	100	
Age (years)					0.384
Median (range)	77 (7	'0-95)	76 (7	'0-93)	
Gender					<0.001
Female	194	59.1	323	74.3	
Male	134	40.9	112	25.7	
Tumor type					<0.001
Breast	126	38.4	246	56.6	
Colorectal	116	35.4	70	16.1	
Lung	51	15.5	21	4.8	
Prostate	21	6.4	53	12.2	
Ovarian	14	4.3	45	10.3	
Time point of assessment					0.991
New diagnosis	209	63.7	277	63.7	
Progression/relapse	119	36.3	158	36.3	
Stage					<0.001
I	22	6.7	66	15.2	
II	59	18.0	112	25.7	
III	50	15.2	62	14.3	
IV	197	60.1	195	44.8	
Treatment received					
Surgery	125	38.1	220	50.6	0.001
Chemotherapy	172	52.4	140	32.2	<0.001
Radiotherapy	80	24.4	155	35.6	0.001
Hormonal therapy	89	27.1	208	47.8	<0.001
Pain: VAS (0-10)					0.218
Score 0	153	46.8	221	50.8	
Score 1-3	59	18.0	59	13.6	
Score ≥4	115	35.2	155	35.6	

Table 1. Patient and clinical characteristics

Abbreviations: VAS: Visual Analogue Scale.

* more than one modality is possible

Patients that were excluded from the analysis were compared with the included patients by means of the chisquare test and the Mann-Whitney test.

			Inclu	Included Exclud			
Screening tools	Cut-off	Score	n	%	n	%	p- value
G8 (0-17)	≤14	Absence of a geriatric risk profile: score > 14 Presence of a geriatric risk profile: score ≤ 14	72 256	22.0 78.0	125 310	28.7 71.3	0.034
fTRST	≥1	Absence of a geriatric risk profile: score 0 Presence of a geriatric risk profile: score ≥ 1	51 277	15.5 84.5	80 355	18.4 81.6	0.303
ECOG-PS		Score 0-1 Score 2-4	234 94	71.3 28.7	325 110	74.7 25.3	0.298
GA components	Item/ Instrument	Score					
Social data	Living situation	Not living alone Living alone	232 96	70.7 29.3	290 145	66.7 33.3	0.232
Functional Status	ADL (6-24)	Independent: score 6 Dependent: score ≥ 7	158 170	48.2 51.8	214 221	49.2 50.8	0.779
	IADL (0-5/8)	Independent: score 8 (female) or 5 (male) Dependent: score < 8 (female) or 5 (male)	137 191	41.8 58.2	192 243	44.1 55.9	0.513
	Fall history	No falls ≥ 1 fall	216 112	65.9 34.1	304 131	69.9 30.1	0.237
Fatigue	MOB-T (0-6)	No fatigue Presence of fatigue	135 193	41.2 58.8	180 255	41.4 58.6	0.951
Cognition	MMSE (0-30)	Score ≥ 24 = normal cognition Score ≤ 23 = mild/severe cognitive decline	286 42	87.2 12.8	398 37	91.5 8.5	0.054
Depression	GDS-15 (0- 15)	Score 0-4 = not at risk for depression Score 5-15 = at risk for depression	247 81	75.3 24.7	357 78	82.1 17.9	0.023
Nutrition	MNA-SF (0- 14)	Score ≥ 12 = normal nutritional status Score 0-11 = risk of malnutrition + malnourished	112 216	34.1 65.9	176 259	40.5 59.5	0.075
Comorbidity	CCI (0-37)	No comorbidities (score 0) Comorbidity score ≥ 1	148 180	45.1 54.9	207 228	47.6 52.4	0.499
Polypharmacy		0-4 different drugs ≥ 5 different drugs	152 176	46.3 53.7	217 218	49.9 50.1	0.332
GA summary		0					
LOFS							0.085
		9-10 (fit)	79	24.1	127	29.2	
		7-8 (slightly vulnerable)	122	37.2	161	37.0	
		5-0 (VUINERADIE) 3-4 (frail)	65 27	19.8 11 2	94 25	21.6 g n	
		2-1 (very frail)	25	7.6	18	4.1	

Table 2: Results of screening and geriatric assessment

Abbreviations: CI = confidence interval; GA = Geriatric Assessment; fTRST = Flemish version of the Triage Risk Screening Tool; ECOG-PS = Eastern Cooperative Oncology Group Performance Status; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MOB-T = Mobility – Tiredness Test; MMSE = Mini Mental State Examination; GDS = Geriatric Depression Scale; MNA-SF = Mini Nutritional Assessment – Short Form; CCI = Charlson Comorbidity Index; LOFS = Leuven Oncogeriatric Frailty Scale.

Patients that were excluded from the analysis were compared with the included patients by means of the chi-square test.

	Continuous variables ^d				Dichotomous variables				
	Univa	riable	Adju	isted ^e	Univa	riable	Adju	isted ^e	
	HR	95%CI	HR	95%CI	HR	95%CI	HR	95%CI	
Laboratory parameters									
CRP (mg/L) ^a	1.12*	1.09-1.14	1.08*	1.06-1.11	2.79*	2.11-3.68	1.82*	1.37-2.43	
Albumin (g/L)	0.91*	0.89-0.93	0.92*	0.90-0.95	2.91*	2.22-3.81	2.91*	1.44-2.52	
Hb (g/dl)	0.84*	0.78-0.90	0.86*	0.79-0.93	1.73*	1.34-2.23	1.51*	1.16-1.96	
Hb gender-specific (WHO) ^b	NA	NA	NA	NA	1.88*	1.45-2.44	1.52*	1.15-2.00	
Hb gender-specific ^c	NA	NA	NA	NA	1.92*	1.48-2.49	1.62*	1.22-2.13	
GPS (categorical)					NA	NA	NA	NA	
- Score 0	Ref		Ref						
- Score 1	3.03*	2.24-4.09	2.05*	1.47-2.86					
- Score 2	4.05*	2.93-5.61	2.64*	1.88-3.71					
mGPS (categorical)					NA	NA	NA	NA	
- Score 0	Ref		Ref						
- Score 1	2.64*	1.95-3.58	1.76*	1.26-2.46					
- Score 2	3.74*	2.71-5.14	2.46*	1.76-3.43	NIA	N 4	NIA	NIA	
GPS (continuous)	2.06*	1.77-2.40	1.65*	1.40-1.94	NA	NA	NA	NA	
mGPS (continuous)	1.97*	1.69-2.30	1.58*	1.34-1.86	NA	NA	NA	NA	
CRP/albumin ratio	1.30*	1.24-1.37	1.25*	1.18-1.34	NA	NA	NA	NA	
GA components									
MNA-SF	0.82*	0.78-0.86	0.86*	0.82-0.90	2.88*	2.13-3.90	2.15*	1.58-2.93	
GDS-15	1.11*	1.06-1.16	1.14*	1.09-1.21	1.95*	1.47-2.58	2.13*	1.59-2.85	
ADL	1.09*	1.04-1.14	1.10*	1.05-1.15	1.35*	1.05-1.75	1.54*	1.17-2.04	
IADL	NA	NA	NA	NA	1.67*	1.28-2.18	1.67*	1.27-2.21	
MMSE	0.95*	0.92-0.98	0.94*	0.90-0.97	1.41	0.98-2.03	1.55*	1.04-2.32	
Fall history	NA	NA	NA	NA	1.19	0.91-1.55	1.64*	1.24-2.16	
MOB-T	0.89*	0.85-0.94	0.92*	0.87-0.97	1.44*	1.10-1.87	1.11	0.84-1.46	
CCI	1.08*	1.00-1.16	1.10*	1.01-1.19	1.11	0.86-1.43	1.00	0.77-1.30	
Living situation	NA	NA	NA	NA	1.13	0.86-1.50	1.18	0.89-1.57	
Polypharmacy	1.03	0.99-1.08	1.03	0.98-1.07	1.21	0.94-1.56	1.12	0.86-1.45	

Table 3: Prognostic value of laboratory parameters and GA for OS

Table 3: continued

	Continuous variables ^d				Dichotomous variables			
	Univa	riable	Adju	ısted ^e	Univa	riable	Adjusted ^e	
	HR	95%CI	HR	95%CI	HR	95%CI	HR	95%CI
Screening and LOFS								
G8	0.84*	0.80-0.87	0.87*	0.83-0.90	3.09*	2.11-4.51	2.80*	1.89-4.15
fTRST	1.28*	1.15-1.42	1.30*	1.16-1.47	1.81*	1.23-2.67	1.46	0.98-2.16
ECOG-PS	1.33*	1.19-1.48	1.31*	1.16-1.48	2.00*	1.53-2.61	1.77*	1.34-2.35
LOFS: continuous	0.86*	0.82-0.90	0.84*	0.79-0.89				
LOFS: categorical								
9-10 (fit)	Ref		Ref					
7-8 (slightly vulnerable)	1.53*	1.06-2.21	1.21	0.82-1.78				
5-6 (vulnerable)	2.47*	1.65-3.69	1.76*	1.16-2.67				
3-4 (frail)	3.50*	2.23-5.52	2.50*	1.52-4.09				
2-1 (very frail)	3.00*	1.82-4.95	3.91*	2.22-6.90				

Abbreviations: HR = Hazard Ratio; CI = confidence interval; GA = Geriatric Assessment; Hb = hemoglobin; (m)GPS = (modified) Glasgow Prognostic Score; MNA-SF = Mini Nutritional Assessment – Short Form; GDS = Geriatric Depression Scale; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MMSE = Mini Mental State Examination; MOB-T = Mobility – Tiredness Test; CCI = Charlson Comorbidity Index; fTRST = Flemish version of the Triage Risk Screening Tool; ECOG-PS = Eastern Cooperative Oncology Group Performance Status; LOFS = Leuven Oncogeriatric Frailty Scale

^a CRP as a continuous variable was analysed per 10-unit increase

^b Hb was dichotomized according to gender-specific cut-off values proposed by the WHO (Hb < 13.0 g/dl for men; Hb <12.0 g/dl for women)

^c Hb was dichotomized according to gender-specific cut-off values proposed by Beutler E, Waalen J. Blood. 2006 (Hb < 13.2 g/dl for men; Hb <12.2 g/dl for women)

^d Unless specified otherwise (categorical)

^e Adjusted for age, stage, tumor type

* p < 0.05

Table 4: Comparison of the performance of the baseline clinical model and extended models

for overall survival

	CPE	AIC	ΔΑΙC	CPE	AIC	ΔΑΙΟ			
Baseline model (B)									
Clinical information	0.725	2336.86	0	0.725	2336.86	0			
(age+stage+tumor type)									
	Cor	ntinuous variab	les ^e	Dich	Dichotomous variables				
Laboratory parameters									
B + CRP (mg/L) ^a	0.740	2308.96	27.90	0.738	2321.34	15.52			
B + Albumin (g/L)	0.755	2302.65	34.22	0.739	2319.48	17.39			
B + Hb (g/dl)	0.733	2325.03	11.83	0.729	2329.70	7.17			
B + (CRP + albumin + Hb)	0.757	2296.74	40.12	0.747	2311.37	25.49			
B + Hb gender-specific (WHO) ^b	NA	NA	NA	0.730	2329.89	6.97			
B + Hb gender-specific ^c	NA	NA	NA	0.732	2327.21	9.66			
B + GPS (categorical) ^d	0.750	2304.33	32.54	0.750	2304.33	32.54			
B + mGPS (categorical) ^d	0.745	2311.29	25.57	0.745	2311.29	25.57			
B + CRP/albumin ratio	0.742	2306.82	30.04	NA	NA	NA			
10-item GA and laboratory									
parameters	0 700		46.00	0.760		07 50			
B + 10-item GA	0.769	2290.83	46.03	0.763	2299.34	37.52			
B + 10-item GA + CRP ^a	0.780	2268.84	68.03	0.771	2288.59	48.28			
B + 10-item GA + albumin	0.777	2279.27	57.60	0.767	2293.00	43.86			
B + 10-item GA + Hb	0.771	2288.31	48.56	0.764	2299.20	37.66			
B + 10-item GA + (CRP +albumin	0.783	2269.69	67.18	0.773	2287.74	49.12			
B + 10-item GA +Hb gender-	NA	NA	NA	0.765	2298.25	38.62			
specific (WHO) ^b									
B + 10-item GA + Hb gender- specific ^c	NA	NA	NA	0.766	2297.12	39.74			
B + 10-item GA + GPS ^d	0.782	2270.64	66.22	0.780	2273.28	63.58			
B + 10-item GA + mGPS ^d	0.779	2275.47	61.40	0.778	2278.39	58.48			
B + 10-item GA + CRP/albumin	0.780	2269.88	66.99	NA	NA	NA			
GA components									
B + living situation	NA	NA	NA	0.726	2337.55	-0.69			
B + ADL	0.737	2323.00	13.87	0.731	2329.37	7.50			
B + IADL	NA	NA	NA	0.735	2325.09	11.77			
B + falls	NA	NA	NA	0.734	2327.21	9.66			
B + MOBT	0.730	2329.91	6.95	0.725	2338.35	-1.48			
B + MMSE	0.732	2329.17	7.70	0.727	2334.63	2.24			
B + GDS-15	0.741	2315.99	20.87	0.740	2315.43	21.43			
B + MNA-SF	0.749	2298.95	37.91	0.746	2312.65	24.22			
B + CCI	0.729	2334.50	2.36	0.725	2338.86	-2.00			
B + polypharmacy	0.725	2337.45	-0.58	0.725	2338.20	-1.33			
				•					
Table 4: continued

	CPE	AIC	ΔΑΙC	CPE	AIC	ΔΑΙϹ
Baseline model (B)						
Clinical information	0.725	2336.86	0	0.725	2336.86	0
(age+stage+tumor type)						
	Con	tinuous variab	les ^d	Dich	otomous varia	bles
Screening and LOFS						
B + G8	0.748	2300.30	36.57	0.749	2306.66	30.20
B + fTRST	0.741	2320.02	16.84	0.727	2335.07	1.79
B + ECOG-PS	0.736	2321.82	15.05	0.734	2323.77	13.10
B + LOFS continuous	0.746	2307.61	29.26	NA	NA	NA
B + LOFS categorical	0.745	2313.73	23.13	NA	NA	NA

Abbreviations: CI = confidence interval; Hb = hemoglobin; (m)GPS = (modified) Glasgow Prognostic Score; GA = Geriatric Assessment; ADL = Activities of Daily Living; IADL = Instrumental Activities of Daily Living; MOB-T = Mobility – Tiredness Test; MMSE = Mini Mental State Examination; GDS = Geriatric Depression Scale; MNA-SF = Mini Nutritional Assessment – Short Form; CCI = Charlson Comorbidity Index; fTRST = Flemish version of the Triage Risk Screening Tool; ECOG-PS = Eastern Cooperative Oncology Group - Performance Status; LOFS = Leuven Oncogeriatric Frailty Score; CPE = Concordance Probability Estimate; AIC = Akaike Information Criterion ^a CRP as a continuous variable was analysed per 10-unit increase

CRP as a continuous variable was analysed per 10-unit increase

^b Hb was dichotomized according to gender-specific cut-off values proposed by the WHO

^c Hb was dichotomized according to gender-specific cut-off values proposed by Beutler E, Waalen J. Blood. 2006 ^d The (m)GPS was analysed as a categorical variable in the left and right column while the 10-item GA was analysed with continuous where possible and dichotomous variables, respectively.

^e Unless specified otherwise (categorical)

Chapter 7.

The utilization of formal and informal home care by older patients with cancer: a Belgian cohort study with two control groups

Baitar A, Buntinx F, De Burghgraeve T, Deckx L, Bulens L, Wildiers H, van den Akker M

This chapter has been published in the BMC Health Services Research (2017) 17:644.

The online version of the published paper contains supplementary material which have not been included in this dissertation.

Abstract

Background

The purpose of this paper is to analyse the utilization of formal and informal home care among older patients with cancer (OCP) and to compare this with middle-aged patients with cancer (MCP) and older patients without cancer (ONC). Additionally, we examined predictors of transitions towards formal care one year after a cancer diagnosis.

Methods

OCP and MCP had to be recruited within three months after a cancer diagnosis and have an estimated life expectancy over six months. ONC consisted of patients without known cancer, seen by the general practitioner. Formal and informal care were compared between the patient groups at baseline, i.e. shortly after a cancer diagnosis and changes in care were studied after one year.

Results

A total of 844 patients were evaluable for formal care at baseline and 469 patients (56%) at follow-up. At baseline, about half of older adults and 18% of MCP used formal care, while about 85% of cancer patients and 57% ONC used informal care. Formal care increased for all groups after one year though not significantly in OCP. The amount of informal care only changed in MCP which decreased after one year. Cancer-related factors and changes in need factors predict a transition towards formal care after a cancer diagnosis.

Conclusions

A cancer diagnosis has a different impact on the use of formal and informal care than ageing as such. The first year after a cancer diagnosis is an important time to follow-up on the patients' needs for home care.

Introduction

The increasing population of older people and changes in health policy have resulted in a shift from institutional care to home care. It is obvious that functional status plays an important role in the use of professional home care. Therefore, in Belgium and other countries planning and financing of both home nursing and nursing homes are largely based on levels of functioning (1). Other factors known to be associated with professional home care use include age, gender, educational level, marital status, and informal care (2-5). Most of this evidence was based on cross-sectional analyses comparing older people already using professional home care with non-users. Kempen et al. specifically compared older people who started to use professional home care to matched non-users which is more appropriate to understand factors that explain the use of professional home care (2). Geerlings et al. focused on the process of becoming a user of informal and professional home care by studying transitions in the use of care based on longitudinal data in the Netherlands (6). More recent studies, based on the Survey of Health, Ageing and Retirement in Europe (SHARE) data, showed cross-national differences in the dynamics of care between various European countries (7-9). In the present study, we focused on a specific population, Flemish older patients with cancer (OCP). The rising population of older people is accompanied by an increase of cancer prevalence rates. A diagnosis of cancer and subsequent treatment can have a substantial effect on the well-being of patients and their need for home care. As shown in our recent work, patients with cancer have increased levels of depression, loneliness, and increasing difficulties in cognitive functioning over the course of one year (10,11). In light of this, we expect to see significant transitions in care after a diagnosis of cancer. The purpose of this paper is to analyse the utilization of formal and informal home care among OCP shortly after diagnosis and to examine changes after one year. The OCP will be compared to two control groups, a group of middle-aged patients with cancer (MCP) and a group of older primary care patients without cancer (ONC). A second goal is to examine predictors of transitions from no formal care to formal care following a cancer diagnosis.

Patients and methods

Study design and population

This analysis was performed on baseline and one-year follow-up data from the Klimop study, which is an ongoing study in Belgium and the Netherlands on the impact of cancer, aging, and their interaction on the well-being of OCP. For this purpose, OCP are compared with MCP (aging effect) and with ONC (diagnosis effect). The same design was used for the current analysis on home care, however based on our results it sometimes made more sense to compare all patients with cancer to ONC in the discussion section. Full details of the Klimop study have been described elsewhere (12). In short, OCP (\geq 70 years), MCP (50-69 years), and ONC (\geq 70 years) are longitudinally compared for different measures of wellbeing. The group of cancer patients consisted of patients with breast, gastro-intestinal, and lung cancer. Patients had to be recruited within three months after a cancer diagnosis and had to have an estimated life expectancy of more than six months. Data have been collected through personal interviews at baseline (T0), after one year (T1), and subsequently every two years. The analysis for this paper was restricted to patients living at home and who were recruited in Belgium only, given the different homecare system in the Netherlands.

Measurements

Formal care

Professional home care or formal care was dichotomized in 'users' and 'non-users'. Users were defined in this study as having received help from at least one of the following paid professionals in the last three months: home nurse, home help services, physiotherapist, meals on wheels, adult day care, and cleaning help. Formal care was only recorded if the participants had at least five contacts in the last three months prior to the interview in order to avoid the measurement of sporadic use and to be able to evaluate recent use.

Informal care

Participants were asked to indicate who cared for them, apart from professional help. This could be a partner, children, other relatives, friends, neighbours, or volunteers. Informal care was defined as help provided by any of them (partners, children, other relatives, friends, neighbours, or volunteers).

Independent variables

Consistent with previous research on home care, we used Andersen and Newman's behavioural model as a theoretical framework to order variables at the individual level in predisposing, enabling and need factors to predict the utilization of formal care (13). Predisposing variables were age, gender, marital status, and educational level. We considered the availability of informal care as an enabling variable. As need factors we considered functional status, depression, loneliness, fatigue, cognitive status, nutrition, polypharmacy and comorbidity. Since the focus of this study is on cancer, we also evaluated the cancer-related factors tumour type, stage, and treatment.

Functional status was measured with the Katz index of Activities of Daily Living (ADL) (range: 0-6) and the Lawton Instrumental ADL (IADL) scale (range: 0-8 for woman, range: 0-5 for men) (13-14). Dependence in one or more domains for each test was defined as having an impaired test result. Depression was measured using the Geriatric Depression Scale (range: 0-15, cut-off \geq 5), loneliness with the loneliness scale of De Jong-Gierveld (range: 0-11, cut-off \geq 3), fatigue with a Visual Analogue Scale (range: 0-10, cut-off \geq 4), cognitive status with the Mini Mental State Examination (range: 0-30, cut-off <24), and comorbidity with the Charlson Comorbidity Index (CCI) (range: 0-37, cut-off \geq 1) (16-19). Nutrition was measured with a new and adapted version of the Mini Nutritional Assessment-short form (range: 0-14, cut-off \leq 11), which is currently being validated by our group (20).

Data analysis

Firstly, formal and informal care were studied separately. Individual home care services, as specified in our definition of formal care, were studied as well. Finally, we considered formal care and informal care simultaneously. This entails four possible situations at baseline: no care, informal care only, formal care only, and the availability of both formal and informal care. Hence, 16 alternative transitions are possible after one year. Next to transitions to formal care or informal care, we also evaluated whether both types of care substitute ('substitution') or complement ('complementarity') each other to better understand the relationship between formal and informal care. Substitution was defined as the sum of transitions from formal to informal care and vice versa. Complementarity was defined as the sum of transitions from no care or (in)formal care only towards combined formal and informal care.

We performed comparative analyses between OCP and two control groups ONC and MCP at baseline by means of the chi-square test. Changes in care over time were studied by comparing formal and informal care between baseline and one-year follow-up within each patient group with the McNemar test and by calculating the percentages of every transition in and between formal and informal care. We set alpha at 0.05 for all analyses to denote statistical significance.

For the second goal of our analysis, logistic regression analyses were conducted to explain the transition of no formal care to formal care in patients with cancer, both OCP and MCP. Separate analyses for OCP and MCP were not conducted due to a small sample size for OCP. Univariate analyses were performed with all predisposing, enabling, and need factors as continuous predictors with the exception of IADL which was analyzed as a dichotomous variable due to the different score range for women and men. In addition, we considered changes of need factors over time by dichotomizing them and identifying changes in categories between baseline and one-year follow-up. All analyses were performed using SPSS 23 software (Chicago, IL).

Results

Study population

A total of 844 patients, recruited between April 2010 and November 2013, were available for analysis at baseline. At follow-up, data for formal care were also available for a total of 469 patients (56%) (see Figure 1). Missing follow-up data were due to death (2.7% ONC, 10.6% OCP, and 6.3% MCP) or to loss of follow-up/refusal (38.7%, 37.3%, and 39.4%). Patient characteristics of OCP and the two control groups are shown in Table 1. Differences in need factors for formal care are found in Table 2. The majority of the patients were female (ONC: 61.3%, OCP: 69.6%, MCP: 75.1%). Compared to ONC, OCP had a worse nutritional status, were less lonely, and had less comorbidity and polypharmacy.



Figure 1. The utilization of formal care: Patient flow chart

* In one hospital, data collection at T0 was integrated in a routine geriatric assessment; therefore data on home care were not available for 56 OCP.

Table 1. Patient characteristics of the three patient groups at baseline	2
--	---

	0	NC		0	СР		м	СР
	n	%	p-value	n	%	p-value	n	%
Total N° pts	333	100,0		161	100,0		350	100,0
Age								
Mean (SD)	78.7 (5.7)			76.9	(5.0)		59.8	(5.4)
Gender			0,07			0,19		
Female	204	61,3		112	69,6		263	75,1
Male	129	38,7		49	30,4		87	24,9
Living situation			0,75			< 0.001		
Alone	109	32,7		55	34,2		48	13,7
Not alone	224	67,3		106	65,8		302	86,3
Marital status			0,68			< 0.001		
Married/living together	209	62,8		98	60,9		285	81,4
Unmarried/widow/divorced	124	37,2		63	39,1		65	18,6
Educational level	n=	328	0,66	n=	157	< 0.001	n=	344
≤14 years	94	28,7		50	31,8		37	10,8
15-19 years	150	45,7		72	45,9		161	46,8
≥19 years	84	25,6		35	22,3		146	42,4
Tumour type						0,07		
Breast				86	53,4		217	62,0
Gastrointestinal				69	42,9		114	32,6
Lung				6	3,7		19	5,4
Stage				n=	137	0,004	n=	327
I				18	13,1		87	26,6
П				69	50,4		121	37,0
111				35	25,5		93	28,4
IV				15	10,9		26	8,0
Treatment *								
Surgery**				132	89,8	0,77	304	88,9
Chemotherapy**				63	43,4	<0,001	210	61,6
Radiotherapy**				71	49,3	<0,001	232	68,0
Hormonal**				59	40,7	0,03	175	51,6

OCP were compared to the two control groups with the chi-square test. P < 0,05 denotes statistical significance.

* More than one possibility.

** Percentages were calculated on valid cases

	ONC			ОСР			м	СР
	n	%	p-value	n	%	p-value	n	%
Total N° pts	333	100,0		161	100,0		350	100,0
ADL			0,80			<0,001		
Independent	184	55,3		87	54,0		273	78,0
Dependent	149	44,7		74	46,0		77	22,0
IADL	n=	329	0,55	n	=160	<0,001	n=	344
Independent	184	55,9		94	58,8		269	78,2
Dependent	145	44,1		66	41,3		75	21,8
Depression	n=	322	0,53	n	=148	0,34	n=	322
Normal	276	85,7		130	87,8		292	90,7
Impaired	46	14,3		18	12,2		30	9,3
Cognition			0,72			0,01		
Normal	292	87,7		143	88,8		333	95,1
Impaired	41	12,3		18	11,2		17	4,9
Nutrition	n=	280	<0,001	n	=135	0,45	n=	320
Normal	229	81,8		48	35,6		102	31,9
At risk/malnourished	51	18,2		87	64,4		218	68,1
Loneliness	n=	318	0,004	n	=138	0,03	n=	322
Notlonely	178	56,0		97	70,3		257	79,8
Lonely	140	44,0		41	29,7		65	20,2
Fatigue	n=	331	0,19	n	=149	0,18	n=	336
No fatigue	150	45,3		58	38,9		153	45,5
Fatigue	181	54,7		91	61,1		183	54,5
Polypharmacy			<0,001			<0,001		
< 5 drugs	150	45,0		109	67,7		305	87,1
≥5 drugs	183	55,0		52	32,3		45	12,9
Comorbidity	n=	n=326 < 0,001		n=148 < 0.0		<0.001	n=339	
CCI 0	145	44,5		92	62,2		269	79,4
$CCI \ge 1$	181	55,5		56	37,8		70	20,6

Table 2. Need factors for formal care of the three patient groups at baseline

OCP were compared to the two control groups with the chi-square test. P < 0,05 denotes statistical significance.

The comparison of formal and informal care at baseline

Results of the baseline comparisons between OCP and the two control groups are summarized in Table 3. About half of the OCP were users of formal care (51.6%). There was no difference with the group ONC (50.2%), but significantly less MCP (18.0%) were users.

	ONC			0	СР		Μ	СP
	n	%	p-value	n	%	p-value	n	%
Formal care	n=333		0,77	n=161		< 0.001	n=	350
Users	167	50,2		83	51,6		63	18,0
Non-users	166	49,8		78	48,4		287	82,0
Individual home care service								
Home nursing	66	19,8	0,86	33	20,5	< 0.001	14	4,0
Home help services	24	7,2	0,50	9	5,6	0,01	5	1,4
Cleaning help	140	42,0	0,41	74	46,0	< 0.001	49	14,0
Physiotherapist	37	11,1	0,003	5	3,1	0,62	14	4,0
Meals on wheels	26	7,8	0,24	8	5,0	< 0.001	0	0,0
Informal care	n=	331	< 0.001	n=157		0,86	n=	335
Present	188	56,8		135	86,0		286	85,4
Not present	143	43,2		22	14,0		49	14,6
Formal care + informal care	n=	331		n=	157		n=	335
No care	84	25,4	< 0,001	11	7,0	0,13	38	11,3
Informal care only	80	24,2	< 0,001	66	42,0	< 0,001	237	70,7
Formal care only	59	17,8	0,001	11	7,0	0,06	11	3,3
Informal care + formal care	108	32,6	0,02	69	43,9	<0,001	49	14,6

Table 3. The comparison of formal and informal care at baseline between OCP and two control groups

OCP were compared to the two control groups with the chi-square test. P < 0,05 denotes statistical significance.

The analysis of the individual home care services only show a significant difference (p=0.003) for seeking help from a physiotherapist between OCP (3.1%) and ONC (11.1%). Compared to MCP, we observed no difference for help from the physiotherapist but OCP made significantly more use for all other home care services. None of the patients made use of adult day care. An overview of the number of home care services per patient at T0 is shown in Additional file 1 (available in the online publication). While most MCP received 1 or 2 individual home care services, more than 10% of the OCP and ONC relied on 3 or more services.

Furthermore, our results showed that OCP (86.0%) could rely as much on informal care as MCP (85.4%) and more so than ONC (56.8%).

When considering both formal and informal care, the distribution of types of care differed in every aspect between OCP and ONC. ONC had a higher proportion (25.4%) of patients with no care compared to the two cancer cohorts (OCP: 7.0%, MCP: 11.3%). MCP had more informal care only (70.7%) than OCP (42.0%) while OCP relied more on both types of care (43.9%) than MCP (14.6%).

	ONC				ОСР				M	CP					
	٦	Ю	Г	1			то	-	Г1		-	ГО	٦	[1	
	n	%	n	%	p-value	n	%	n	%	p-value	n	%	n	%	p-value
Formal care	n=	333	n=	195	0,03	n	=161	n	=84	0,06	n=	=350	n=	190	< 0,001
Users	167	50,2	110	56,4		83	51,6	55	65,5		63	18,0	77	40,5	
Non-users	166	49,8	85	43,6		78	48,4	29	34,5		287	82,0	113	59,5	
Individual home care service															
Home nursing	66	19,8	42	21,5	0,08	33	20,5	20	23,8	1,00	14	4,0	21	11,1	0,01
Home help services	24	7,2	9	4,6	0,21	9	5,6	6	7,1	0,73	5	1,4	3	1,6	1,00
Cleaning help	140	42,0	92	47,2	0,29	74	46,0	43	51,2	1,00	49	14,0	44	23,2	0,02
Physiotherapy	37	11,1	23	11,8	0,66	5	3,1	16	19,0	0,01	14	4,0	34	17,9	< 0,001
Meals on wheels	26	7,8	11	5,6	0,11	8	5,0	3	3,6	1,00	0	0,0	1	0,5	1,00
Informal care	n=	331	n=	191	1,00	n	=157	n	=80	0,65	n=	-335	n=	185	0,03
Present	188	56,8	103	53,9		135	86,0	63	78,8		286	85,4	140	75,7	
Not present	143	43,2	88	46,1		22	14,0	17	21,3		49	14,6	45	24,3	
Formal care + informal care	n=	331	n=	191		n	=157	n	=80		n=	-335	n=	185	
No care	84	25,4	40	20,9	0,10	11	7,0	7	8,8	1,00	38	11,3	30	16,2	0,24
Informal care only	80	24,2	43	22,5	1,00	66	42,0	19	23,8	0,01	237	70,7	79	42,7	< 0,001
Formal care only	59	17,8	48	25,1	0,07	11	7,0	10	12,5	0,79	11	3,3	15	8,1	0,08
Informal care + formal care	108	32,6	60	31,4	1,00	69	43,9	44	55,0	0,06	49	14,6	61	33,0	< 0,001

Tabel 4. The comparison of formal and informal care between T0 and T1 in each patient group

Proportions between T0 and T1 were compared with the McNemar test. P < 0,05 denotes statistical significance.

The comparison of formal and informal care between T0 and T1

Table 4 presents the differences in care between T0 and T1 in the three patient groups. In the group ONC, there was overall an increase of formal care at T1. No other differences were observed.

In the group OCP, there was overall an increase of formal care at T1. However, this difference was not statistically significant (p=0.06). At the level of individual home care services, significantly more patients saw a physiotherapist after one year (from 3.1% to 19.0%, p=0.01). There was no significant difference in informal care between T0 and T1 (p=0.65). However, when considering both formal and informal care, fewer patients received only informal care (p=0.01) at T1.

In the group MCP, there was overall an increase of formal care at T1. Increased help from a home nurse (from 4.0% to 11.1%), cleaning help (from 14.0% to 23.2%), and a physiotherapist (from 4.0% to 17.9%) was reported. From the patients who received physiotherapy, 71.4% and 94.1% had breast cancer at respectively T0 and T1. Informal care decreased at T1 (from 85.4% to 75.7%). When considering both formal and informal care, fewer patients received only informal care and more patients received both formal and informal care.

	ONC			0	СР		Μ	СР
	n	%	p-value	n	%	p-value	n	%
				Form	al care			
Transition	n=	195		n	=84		n=:	190
No formal care at T0 and T1	80	41,0	0,05	24	28,6	< 0,001	101	53,2
Formal care at T0 and T1	94	48,2	0,93	41	48,8	< 0,001	26	13,7
No formal care> Formal care	16	8,2	0,04	14	16,7	0,07	51	26,8
Formal care> No formal care	5	2,6	0,16	5	6,0	0,91	12	6,3
				Inforn	nal care	-		
Transition	n=	190		n	=79		n=:	178
No informal care at T0 and T1	62	32,6	< 0,001	6	7,6	0,55	10	5,6
Informal care at T0 and T1	79	41,6	< 0,001	54	68,4	0,88	120	67,4
No informal care> Informal care	24	12,6	0,56	8	10,1	0,77	16	9,0
Informal care> No informal care	25	13,2	0,87	11	13,9	0,42	32	18,0

Table 5. Transitions in formal and informal home care analyzed separately

OCP were compared to the two control groups with the chi-square test. P < 0.05 denotes statistical significance.

Transitions in formal and informal care after one year

Transitions in formal and informal care were analyzed separately in Table 5. There was a significant difference in the transition from no formal care to formal care between OCP and the two control groups. For the group OCP, 16.7% made this transition compared to 8.2% ONC (p=0.04) and 26.8% MCP (p=0.07). For the transition from formal care to no formal care, a similar proportion OCP (6.0%) stopped

formal care compared to MCP (6.3%) after one year. Only a small proportion of ONC (2.6%, p=0.16) stopped relying upon formal care.

Transitions in informal care were similar between OCP and the two control groups. In the groups ONC, OCP, and MCP respectively 12.6%, 10.1%, and 9.0% made a transition to informal care at T1 while respectively 13.2%, 13.9%, and 18.0 stopped relying on informal care.

Transitions considering the availability of formal care and informal care simultaneously are summarized in Additional file 2 (available in the online publication). Transitions in care, from any type, were observed in ONC, OCP, and MCP in respectively 33.2%, 39.2%, and 49.4% of the patients. The analysis of substitution and complementarity of care shows that in 5.1% of OCP formal and informal care substitute each other, while in 17.7% both types of care complement each other at T1. Results for the control groups are shown in Table 6.

	ONC			ОСР			м	СР	
	n	%	p-value	n	%	p-value	n	%	
Transition	n=190			n=79			n=	n=178	
Substitution	0	,0	0,07	4	5,1	1,0	11	6,2	
Complementarity	19	10,0	0,08	14	17,7	0,44	39	21,9	

Table 6. Substitution and complementarity of formal and informal care

OCP were compared to the two control groups with the chi-square test or fisher's exact test where appropriate. p < 0.05 denotes statistical significance.

Predictors of the transition no formal care to formal care after a cancer diagnosis

The studied sample for this analysis consisted of a total of 190 patients with cancer (20.0% OCP), of which 65 made the transition from no formal care to formal care. Significant predictors in univariate analysis are shown in Table 7. Next to a higher value for the need factors fatigue and polypharmacy, certain changes (or the lack of) in ADL, IADL, depression, fatigue, and polypharmacy were predictive for a transition towards formal home care. Furthermore, a worse cancer stage and having received chemotherapy or radiotherapy were predictive as well. Factors that were not predictive included predisposing variables, ADL, IADL, informal care, and belonging to the group OCP or MCP.

Table 7. Predictors of the transition no formal to formal care 1 year after a cancer diagnosis (OCP and MCP combined, n=190)

	Univariate*						
	OR	95%CI	p-value				
Need factors							
Fatigue	1,24	1,09-1,41	0,001				
Polypharmacy	1,18	1,01-1,38	0,04				
Cancer-related factors							
Stage			0,09				
1	ref						
П	2,01	0,85-4,78	0,11				
III	2,67	1,02-6,96	0,05				
IV	5,93	1,18-29,68	0,03				
Chemotherapy	2,21	1,17-4,19	0,02				
Radiotherapy	2,38	1,19-4,77	0,02				
Changes in need factors							
ADL			0,01				
persistently independent	ref						
became indepedent	0,93	0,27-3,13	0,90				
became dependent	3,19	1,55-6,54	0,002				
persistently dependent	0,85	0,28-2,54	0,77				
IADL			0,04				
persistently independent	ref						
became indepedent	1,81	0,60-5,50	0,29				
became dependent	2,86	1,36-6,04	0,01				
persistently dependent	2,04	0,66-6,33	0,22				
Depression			0,06				
persistently normal	ref						
became normal	2,53	0,77-8,32	0,13				
became depressed	3,03	1,21-7,60	0,02				
persistently depressed	2,53	0,34-18,59	0,36				
Fatigue			0,003				
persistently normal	ref						
became normal	6,00	1,55-23,19	0,01				
became impaired	4,00	1,19-13,42	0,03				
persistently impaired	8,27	2,61-26,22	<0,001				
Polypharmacy			0,04				
persistently normal	ref						
became normal	2,51	0,60-10,52	0,21				
became impaired	2,72	1,15-6,46	0,02				
persistently impaired	2,93	0,93-9,25	0,07				

* Predictors of univariate logistic regression analyses are shown with p < 0,05

Discussion

Transitions in home care depend on the situation. A diagnosis of cancer might be considered a situation of greater need for care and as such we observed transitions in formal and informal care in ONC, OCP, and MCP in respectively 33.2%, 39.2%, and 49.4% after one year. Our results also showed an expected increase in new users of formal care in cancer patients: 16.7% for OCP, 26.8% for MCP compared to 8.2% for ONC. However, at baseline, i.e. shortly after diagnosis, some important differences were already observed between the patient groups in terms of care but also in patient characteristics. While many differences between OCP and MCP can be explained due to age-related factors, some differences between OCP and ONC were less obvious. OCP had a lower comorbidity burden and polypharmacy. This can partially be explained by a referral bias for OCP; the frailest patients are not always referred to the oncologist. OCP reported to be less lonely despite no difference in marital status or living situation. Previously, we already showed that at baseline fewer patients with cancer had feelings of loneliness than ONC and that after one year the proportion of cancer patients with loneliness increased significantly reaching the levels of ONC (11). Our current analysis might explain these differences in loneliness by looking at the received informal care. More patients with cancer relied on informal care compared to ONC at baseline. This care was likely provided only recently around the time the patient was informed about a cancer diagnosis which could explain the difference in baseline loneliness between patients with cancer and ONC. Furthermore, informal care decreased after one year in both cancer cohorts, although not significantly in OCP, which could contribute to the increase in loneliness in both cancer cohorts. It is to be expected that newly diagnosed patients receive much help and support from their environment at first but less so after completing their treatment and this might have an impact on feelings of loneliness, an important measure of well-being.

The analysis of the individual home care services showed that cleaning help was clearly the most used service in the three patient groups. In this regard, we note that we did not document whether the use of this service was related to any health-related issues. Furthermore, at baseline fewer cancer patients visited the physiotherapist compared to ONC. However, after one year a strong increase in physiotherapy was observed in both cancer cohorts, mainly in patients with breast cancer. This might be related to lymphoedema following breast cancer surgery or radiotherapy. While, besides physiotherapy, no other changes were observed after one year in OCP, MCP made more use of home nursing and cleaning help next to an increase in physiotherapy. A possible explanation for this is that due to lymphoedema, MCP are less able to do household tasks like cleaning or are advised to limit the strenuous use of their arm. Also, an important proportion of patients with cancer will have had an intestinal stoma. These patients, OCP likely more than MCP, might rely upon home nursing for stoma care. These increases of home nursing and cleaning help after one year in MCP are however not observed in OCP. However, the baseline percentages for both home care services are already high in OCP and new tasks might have been covered by the already available care in this group.

About half of the OCP and ONC received formal care at baseline. This formal care was more often in combination with informal care in OCP than in ONC. When considering both formal care and informal care after one year, the main trend observed in both cancer cohorts, in contrast to ONC, is the decrease in the number of patients who rely on informal care only and the increase in patients with formal care whether or not with informal care. This distinction between ONC and cancer patients can be explained by the dynamics in informal care in cancer patients as discussed previously and by the expected increased need for formal care in cancer patients which was quantified in this study. Another important observation is the lack of any change in the proportion of patients with no care whatsoever in the three patient groups.

Several studies in the general population suggest that formal and informal care complement rather than substitute each other (6-8). This is also shown in our three patient groups. Our rates for substitution and complementarity for ONC are similar with other reports for the general population in Belgium (8). These rates are however higher in our studied cancer cohorts, particularly for complementarity. In the context of cancer, more technical skills (e.g. injections, stoma care, and physiotherapy) for care might be required and this care will complement rather than substitute informal care. Several care models have been proposed for the general population (21). For patients diagnosed with cancer, formal and informal care might be better explained with a complementarity and task-specific model.

Our results show different dynamics in care between older patients with and without cancer but also between OCP and MCP. International guidelines recommend the implementation of a geriatric assessment in OCP to guide treatment decisions in routine oncology practice (22). This assessment will lead to the necessary referrals for many patients with cancer (23). In contrast to the general population, many OCP will be seen by a social worker in the hospital which will also drive changes in formal care. In this regard, we mention the Belgian implementation of the InterRAI instruments (BelRAI) which is an ambitious web-based comprehensive assessment system to improve the quality and continuity of care across different health care settings, including home care (24-26). It is advisable to harmonize recommendations in geriatric oncology to implement geriatric assessment in daily practice with efforts like the BelRAI for the general population at a national level for future policies.

For the second goal of our study, we analyzed predictors for the transition from no formal care to formal care one year after a cancer diagnosis. Functional status, the availability of informal care, and dispositional factors did not predict the transition towards formal care in patients with cancer. Cancer-related factors, i.e. more advanced disease or a more extensive treatment, and the related worsening of need factors (more than their baseline values) were significant predictors. In contrast, an analogous analysis on ONC showed that only developing a nutritional impairment predicted a transition towards formal care (data not reported). Our results show that the first year after a cancer diagnosis is an important time to follow-up on the patients' needs for home care at different time points during the

disease trajectory. More longitudinal research is needed to determine to what degree a cancer diagnosis is a turning point towards more formal care or whether the increased use of formal care is more of a temporary nature. Another evolution to follow in the future relevant to homecare, is the parenteral administration of cancer treatments at home. The first pilot projects in this regard have been started in Belgium.

When interpreting our results, some considerations should be made. The study of both formal and informal care is a strength of this study, however the collection of more detailed information on informal care would have been beneficial for our analysis. Many patients were lost to follow-up (LTFU), i.e. a total of 38.7% when not considering the patients that died (5.7%). Additional analyses (Additional file 3 available in the online publication) show no major baseline differences between LTFU patients who were alive and patients with one-year follow-up data. LTFU patients had less cleaning help and more informal care only, no other differences were observed for home care. Furthermore, there was a similar proportion of LTFU patients (excluding deaths) in the three patient groups. Another point to consider is that our results apply for Belgium, a well-developed welfare state with well-developed formal services. A lot has been written in the home care literature, which focuses on the general population, about the different dynamics of formal and informal care between European countries, about the influence of the strength of family ties, and a north-south gradient (8). To our knowledge, there are no similar studies to which we can compare our results with that evaluate changes in both formal and informal care after a cancer diagnosis.

Conclusions

A cancer diagnosis has a different impact on the use of formal and informal care than ageing as such. The first year after a cancer diagnosis is an important time to follow-up on the patients' needs for home care.

Author contributions

This study was designed by AB, FB, MvdA, whereas the umbrella study KLIMOP was designed and set up by FB, MvdA, LD, PB and HW. Data collection was performed by all authors. AB, FB, TdB, MvdA drafted the manuscript, which was commented by all authors. Statistical analysis was performed by AB and discussed with FB, MvdA, and TdB. All authors read and approved the final manuscript.

References

1) Sermeus W, Pirson M, Paquay L, Pacolet J, Falez F, Stordeur S, Leys M. Financing of home nursing in Belgium – KCE report 2010

2) Kempen GI, Suurmeijer TP. Factors influencing professional home care utilization among the elderly. Soc. Sci. Med. 1991;32(1):77-81.

3) Frederiks CM, te Wierik MJ, van Rossum HJ, Visser AP, Volovics A, Sturmans F. Why do elderly people seek professional home care? Methodologies compared. J Community Health 1992;17(3):131-41.

4) Kemper P, The use of formal and informal home care by the disabled elderly. Health Services Research 1992;27,421-49.

5) Kadushin G. Home health care utilization: a review of the research for social work. Health Soc Work. 2004;29(3):219-44.

6) Geerlings S, Pot AM, Twisk JWR, Deeg DJH. Predicting transitions in the use of informal and professional care by older adults. Ageing Soc 2005;25(1):111–130

7) Bolin K, Lindgren B, Lundborg P. Informal and formal care among single-living elderly in Europe. Health Econ. 2008;17(3):393-409.

8) Geerts J, Van den Bosch K. Transitions in formal and informal care utilization amongst older Europeans: The impact of national contexts. European Journal of Ageing 2012;9(1):27-37.

9. Suanet B, Broese Van Groenou M, Van Tilburg T. Informal and formal home-care use among older adults in Europe: can cross-national differences be explained by societal context and composition? Ageing and Society 2012;32(3):491-515.

10) Deckx L, van Abbema D, van den Akker M, van den Broeke C, van Driel M, Bulens P, et al. A cohort study on the evolution of psychosocial problems in older patients with breast or colorectal cancer: comparison with younger cancer patients and older primary care patients without cancer. BMC Geriatr 2015;15:79.

11) Deckx L, van den Akker M, van Driel M, Bulens P, van Abbema D, Tjan-Heijnen V, et al. Loneliness in patients with cancer: the first year after cancer diagnosis. Psychooncology 2015;24(11):1521-8

12) Deckx L, van Abbema D, Nelissen K, Daniels L, Stinissen P, Bulens P, et al. Study protocol of KLIMOP: a cohort study on the wellbeing of older cancer patients in Belgium and the Netherlands. BMC Public Health 2011;11: 825.

13) Andersen, R. (1995). Revisiting the behavioral model and access to medical care: Does it matter? *Journal of Health and Social Behavior*, 36(1), 1-10.

14) Katz S, Ford AB, Moskowitz RW, Jackson BA, Jaffe MW. Studies of Illness in the Aged - the Index of Adl - A Standardized Measure of Biological and Psychosocial Function. Jama-Journal of the American Medical Association 1963;185(12):914-9.

15) Lawton MP, Brody EM. Assessment of Older People - Self-Maintaining and Instrumental Activities of Daily Living. Gerontologist 1969;9(3P1):179-&.

16) Yesavage JA, Brink TL, Rose TL, Lum O, Huang V, Adey M, Leirer VO. Development and validation of a geriatric depression screening scale : a preliminary report. Journal of psychiatric research 1982;17:37-49.

17) De Jong-Gierveld J, Kamphuis F. The development of a Rasch-Type Loneliness Scale. Applied Psychological Measurement 1985;9:289-299.

18) Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. J Psychiatr Res 1975;12(3):189-98.

19) Charlson ME, Pompei P, Ales KL, Mackenzie CR. A New Method of Classifying Prognostic Co-Morbidity in Longitudinal-Studies - Development and Validation. Journal of Chronic Diseases 1987;40(5):373-83.

20) Rubenstein LZ, Harker JO, Salvà A, et al. (2001) Screening for undernutrition in geriatric practice: Developing the short-form mini-nutritional assessment (MNA-SF). J Gerontol A Biol Sci Med Sci 56:M366–M372.

21) Lyons KS, Zarit SH. Formal and informal support: the great divide. Int J Geriatr Psychiatry 1999;14(3):183-92.

22) Wildiers H, Heeren P, Puts M, Topinkova E, Janssen-Heijnen ML, Extermann M, et al. International Society of Geriatric Oncology Consensus on Geriatric Assessment in older patients with cancer. J Clin Oncol 2014;32(24):2595-603.

23) Baitar A, Kenis C, Moor R, Decoster L, Luce S, Bron D, et al. Implementation of geriatric assessment-based recommendations in older patients with cancer: a multicentre prospective study. J Geriatr Oncol 2015;6(5):401-10.

24) Buntinx F, Falez F, Swine C, Ylieff M, Closon MC, De Lepeleire J. Interface: Rapport 2003. De organisatie en de financiering van een grensoverschrijdend zorgaanbod in de ouderenzorg/Organisation

et financement offre de soins transinstitutionnelle secteur soins aux personnes agée. Leuven-Brussel; 2003.

25) De lepeleire J, Falez F, Swine C, Ylieff M, Pepersack T, Buntinx F. Interface: Rapport 2005. De organisatie en de financiering van een grensoverschrijdend zorgaanbod in de ouderenzorg/Organisation et financement offre de soins transinstitutionnelle secteur soins aux personnes agées. Leuven-Brussel; 2005.

26) Vanneste D, Vermeulen B, Declerq A. Healthcare professionals' acceptance of BelRAI, a webbased system enabling person-centred recording and data sharing across care settings with interRAI instruments: a UTAUT analysis. BMC Med Inform Decis Mak 2013;13-129. Chapter 8.

The influence of coping strategies on subsequent well-being in older patients with cancer: a comparison with two control groups.

Baitar A, Buntinx F, De Burghgraeve T, Deckx L, Schrijvers D, Wildiers H, van den Akker M

This chapter has been published in the Journal of Psycho-Oncology 2017;1-7

The online version of the published paper contains supplementary material which have not been included in this dissertation.

Abstract

Objective

To evaluate dispositional coping strategies as predictors for changes in well-being after one year in older patients with cancer (OCP) and two control groups.

Methods

OCP were compared with two control groups: middle-aged patients with cancer (MCP) (aging effect) and older patients without cancer (ONC) (cancer effect). Patients were interviewed shortly after a cancer diagnosis and one year later. Dispositional coping was measured with the Short Utrecht Coping List. For well-being, we considered psychological well-being (depression, loneliness, distress) and physical health (fatigue, ADL, IADL). Logistic regression analyses were performed to study baseline coping as predictor for subsequent well-being while controlling for important baseline covariates.

Results

A total of 1245 patients were included in the analysis at baseline: 263 OCP, 590 ONC, and 392 MCP. Overall, active tackling was employed most often. With the exception of palliative reacting, OCP utilized each coping strategy less frequently than MCP. At one-year follow-up, 833 patients (66.9%) were interviewed. Active coping strategies (active tackling and seeking social support) predicted subsequent well-being only in MCP. Avoidance coping strategies did not predict well-being in any of the patient groups. Palliative reacting predicted distress in OCP; depression and dependency for ADL in MCP.

Conclusions

Coping strategies influence subsequent well-being in patients with cancer, but the impact is different in the age groups. Palliative reacting was the only coping strategy that predicted well-being (i.e. distress) in OCP and is therefore, especially in this population, a target for coping skills interventions.

Background

Knowledge of effective and maladaptive coping strategies in relation to a cancer diagnosis is important in the development of interventions. Many studies have focused on the association between coping and well-being in patients dealing with cancer. The large majority of these studies have a cross-sectional design using different measures of coping. Although findings are not entirely consistent, active coping (e.g. active tackling, seeking social support) in contrast to passive coping (e.g. avoidance, palliative reacting) is thought to have a beneficial effect on both psychological and physical outcomes (1-2). However, reciprocal relationships between coping and well-being should be considered (3-4). A small number of longitudinal studies have evaluated coping following a cancer diagnosis as a predictor for subsequent well-being (2,5-10). The findings of these studies are difficult to summarize as they vary for example in study aim and design, in the classification of coping, in health-related outcomes, and in follow-up time. Nonetheless, they demonstrated an influence of coping on subsequent well-being. One study in patients with breast cancer even showed that avoidance coping used within six months after diagnosis predicted psychological well-being three years later (6).

The current paper on coping and subsequent well-being is part of the larger 'Klimop' study. The primary aims of this study are to assess the impact of cancer and ageing on subsequent wellbeing and to identify factors that predict well-being in older patients with cancer. (11). For this purpose, older patients with cancer (OCP) are compared with two control groups: middle-aged patients with cancer (MCP) and older primary care patients without cancer (ONC). From two previous analyses based on the Klimop study, we know that OCP cope differently than MCP shortly after a cancer diagnosis (baseline) and that increasing levels of psychosocial problems are observed one year later in both patient groups (12-13). These findings might suggest that if coping predicts subsequent well-being that this relationship will be different in OCP and MCP. Furthermore, the relationship between dispositional coping and subsequent well-being might be different between OCP and ONC considering changes in well-being related to the cancer.

A better understanding of coping strategies and a potential influence on subsequent health is important because interventions targeting coping skills might improve outcomes. The aim of the present paper is to investigate if baseline coping strategies predict changes in different dimensions of psychological well-being and physical health after one year in OCP while disentangling ageing effects (OCP versus MCP) and cancer effects (OCP versus ONC). In addition, we compared the frequencies of baseline coping strategies in each patient group and compared coping and well-being between OCP and the two control groups at baseline.

Patients and methods

Study design and population

This analysis was performed on baseline and 1-year follow-up data from the Klimop study, which is an ongoing study in Belgium and the Netherlands on the impact of cancer, ageing, and their interaction on well-being. Full methodological details have been described elsewhere (11). In short, OCP (\geq 70 years) are longitudinally compared with MCP (50-69 years) and ONC (\geq 70 years) for different measures of well-being. ONC consisted of patients without known cancer, seen by their general practitioner. The group of patients with cancer consisted of patients with breast, gastro-intestinal, prostate and lung cancer. Patients had to be recruited within three months after cancer diagnosis. All participants had to have an estimated life expectancy of more than six months and no formal diagnosis of dementia. Data have been collected through personal interviews at baseline, after one year, and subsequently every two years.

Coping

Coping was assessed with the short version of the Utrecht Coping List (UCL) (14-15). The UCL evaluates dispositional coping by asking participants about the frequency of coping strategies they use in response to problems or unpleasant situations in general. In this approach, coping strategies are viewed as relatively stable over the course of time and across different situations which allows greater comparison of coping strategies across different samples.

The short UCL consists of 15 items and covers four subscales (Appendix 1). Each question was answered on a four-point Likert scale. For each subscale, sum scores were divided by the number of questions. Total scores for each subscale range from 1 to 4. A higher score indicates that a certain coping strategy is used more often.

Active coping

Active tackling (five questions): this refers to behaviour directed at confronting or solving the problem or situation.

Seeking social support (five questions): this refers to efforts to actively pursue informational, physical, and/or emotional support.

Passive coping

Avoidance and awaiting (three questions): this refers to behaviour to avoid dealing with a situation like seeking distraction.

Palliative reacting (two questions): this refers to behaviour that involves giving up any effort to deal with the situation.

Well-being

For well-being we studied psychological well-being and physical health with tools that are widely used in the literature. A more in-depth description of the tools can also be found in previous publications from the Klimop study (12-13,16).

For psychological well-being, we considered depression, loneliness, and distress measured with the Geriatric Depression Scale (range:0-15, cut-off \geq 5) (17), the loneliness scale of De Jong-Gierveld (range: 0-11, cut-off \geq 3) (18), and the distress barometer (range:0-10, cut-off \geq 4) (19) respectively.

For physical health, we considered fatigue measured with the Visual Analogue Scale (range: 0-10, cut-off \geq 4) (20-22); Activities of Daily Living (ADL) measured with the Katz index (range: 0-6) (23); and Instrumental ADL (IADL) measured with the Lawton IADL scale (range: 0-8 for woman, range:0-5 for men) (24). Dependence in one or more domains of ADL and IADL was defined as having an impairment.

Statistical analysis

We compared OCP with the two control groups for patient characteristics, comorbidity, cancerrelated factors, and well-being with the chi-square test; and coping strategies with the Mann-Whitney U test. We used the Wilcoxon signed rank test to evaluate within-group differences in coping strategies. To study the predictive value of baseline coping strategies for changes in well-being, we used logistic regression and not linear regression considering advantages towards interpretability of results and relevance for clinicians. We first performed univariate logistic regression analyses with patient characteristics (age, gender, living situation, marital status, educational level), comorbidity (Charlson comorbidity index), and cancer-related factors (tumour type, stage, treatment) as predictors for every dimension of psychological well-being and physical health at 1-year follow-up. With the exception of age, all predictors were analysed as categorized variables. Secondly, multiple logistic regression analyses were performed with baseline coping as predictor for well-being at 1-year follow-up adjusted for the studied baseline values of well-being as well as for covariates that were found significant in univariate analysis. Sample sizes varied somewhat for each measure of well-being due to missing data. To test the robustness of the analysis for presence of missing data, worst-case best-case sensitivity analyses were performed. The significance threshold was set at 0.05 for every analysis. All analyses were performed using SPSS 23 software (Chicago, IL).

Ethics

The study protocol was approved by the Ethical Review Board of KU Leuven and UZ Leuven (S52097-ML6279) (Belgium) and the Maastricht University Medical Centre (Nl31414.068.10) (the Netherlands). All patients signed informed consent.

Results

Participants

A total of 1490 patients were included at baseline. For 245 patients (16.4%), there was no or incomplete information on coping, resulting in 1245 patients (83.6%) eligible for analyses: 263 OCP, 590 MCP, and 392 ONC. From these patients 833 (66.9%) could be interviewed at 1-year follow-up. Missing follow-up data were due to death (n= 59, 14.3%) or to loss of follow-up/refusal (n= 353, 85.7%). Patient characteristics of OCP and the two control groups at baseline are shown in Table 1. The most frequent types of cancer were breast and gastrointestinal cancer. Advanced cancer (stage III-IV) was present in 41.2% of OCP and in 34.6% of MCP. Comparative analyses showed that OCP had a lower comorbidity index than ONC and several differences were observed between OCP and MCP. For example, less OCP were female and they received less intensive cancer treatment than MCP (Table 1).

Well-being and coping at baseline

Considering baseline psychological well-being, OCP were more often depressed and distressed but less often lonely than ONC. Compared to MCP, they had more often feelings of loneliness. OCP were more often dependent for ADL and IADL than MCP. There was no difference in physical health compared to ONC (Table 2).

Within-group analyses showed that in each patient group, patients utilized active tackling more frequently than the other three coping strategies (p < 0.05). The comparison between OCP and the two control groups for coping strategies are shown in Table 2. The frequency of every coping strategy was different between OCP and MCP. OCP used less active tackling, seeking social support, and avoidance coping than MCP, while palliative reacting was more frequent in OCP. No differences were observed between OCP and ONC

	0	NC		0	СР		М	СР
	n	%	p-value ^a	n	%	p-value ^b	n	%
Total N° pts	392	100,0		263	100,0		590	100,0
Age								
Mean (SD)	78,2 (5,31)			76,2	(4,54)		60,48	(5,46)
Gender			0,30			0,001		
Female	235	59,9		147	55,9		402	68,1
Male	157	40,1		116	44,1		188	31,9
Living situation	n=	388	0,92			< 0,001	n=	587
Alone	112	28,9		75	28,5		92	15,7
Not alone	276	71,1		188	71,5		495	84,3
Marital status	n=	388	0,88			< 0,001	n=	587
Married/living together	253	65,2		170	64,6		466	79,4
Unmarried/widow/divorced	135	34,8		93	35,4		121	20,6
Educational level	n=	383	0,17	n=	254	< 0,001	n=	575
≤14 years	113	29,5		60	23,6		63	11,0
15-19 years	175	45,7		134	52,8		279	48,5
≥19 years	95	24,8		60	23,6		233	40,5
Comorbidity	n=	334	0,01	n=	241	< 0,001	n=	566
CCI 0-1	229	68,6		189	78,4		504	89,0
CCI ≥2	105	31,4		52	21,6		62	11,0
Tumour type				n=	229	0,15	n=	541
Breast				112	48,9		312	57,7
Gastrointestinal				70	30,6		137	25,3
Lung				28	12,2		50	9,2
Prostate				19	8,3		42	7,8
Stage				n=	233	0,08	n=	552
1-11				137	58,8		361	65,4
III-IV				96	41,2		191	34,6
Treatment *								
Surgery**				184	75,1	0,001	485	84,5
chemotherapy**				94	38,7	< 0,001	314	54,7
radiotherapy**				115	47,5	< 0,001	370	64,5
hormonal**				91	37,4	0,12	248	43,4

Table 1. Patient characteristics of the three patient groups at baseline

OCP were compared to the two control groups with the chi-square test; ^a OCP versus ONC; ^b OCP versus MCP CCI: Charlson comorbidity index

* More than one possibility.

** Percentages were calculated on valid cases

	ONC			ОСР			Μ	СР
	n	%	p-value ^a	n	%	p-value ^b	n	%
	392	100,0		263	100,0		590	100,0
			Ps	ychologic	al well-bei	ng		
Depression			0,03			0,52		
normal	353	90,1		222	84,4		508	86,1
impaired	39	9,9		41	15,6		82	13,9
Loneliness	n=	391	0,02			0,01	n=	587
normal	240	61,4		184	70,0		457	77,9
impaired	151	38,6		79	30,0		130	22,1
Distress	n=	381	0,002	n=	251	0,27	n=	558
normal	302	79,3		171	68,1		358	64,2
impaired	79	20,7		80	31,9		200	35,8
			Physical health					
Fatigue			0,09	n=	255	0,47	n=	576
no fatigue	174	44,4		96	37,6		232	40,3
Fatigue	218	55,6		159	62,4		344	59 <i>,</i> 7
ADL			0,40			0,001		
independent	244	62,2		155	58,9		414	70,2
dependent	148	37,8		108	41,1		176	29,8
IADL	n=	391	0,27	n=	261	0,001	n=	574
independent	249	63,7		155	59 <i>,</i> 4		407	70,9
dependent	142	36,3		106	40,6		167	29,1
				Со	ping			
Coping strategy	Mea	n (SD)		Mea	n (SD)		Mea	n (SD)
active tackling	2,69	(0,58)	0,38	2,66	(0,55)	<0,001	2,87 (0,51)	
social support	2,08	(0,57)	0,63	2,06 (0,61)		<0,001	2,23	(0,62)
avoidance	2,05	(0,55)	0,31	2,10	(0,55)	0,004	2,19 (0,52)	
palliative reacting	2,17	(0,69)	0,07	2,06	(0,66)	0,003	1,92	(0,65)

Table 2. Well-being and coping at baseline: comparison between OCP and control group

Well-being in OCP were compared to the two control groups with the chi-square test.

Coping strategies in OCP were compared with the control groups by using the Mann-Whitney test. ^a OCP versus ONC; ^b OCP versus MCP

The relation between baseline coping and changes after 1 year in well-being

The predictive value of baseline coping for well-being after one year is shown in Table 3. Active tackling predicted only less often distress in MCP. Social support seeking predicted only less often loneliness in MCP. Avoidance coping did not predict any of the measures of well-being in the three patient groups. Palliative reacting predicted distress in OCP, and depression and dependency for ADL in MCP.

Table 3. Adjusted analyses of baseline coping as predictor for subsequent well-being

	Psycholog	gical well-being af	ter 1 year	Phys	ical health after 1	Physical health after 1 year				
Coping at baseline	Depression	Loneliness	Distress	Fatigue	ADL	IADL				
			0	СР						
	n= 153/263 (58,2%)	n= 150/263 (57,0%)	n= 142/263 (54,0%)	n= 131/263 (49,8%)	n= 172/263 (65,4%)	n= 139/263 (52,9%)				
Active tackling	1,15 (0,45-2,96)	1,19 (0,56-2,51)	0,50 (0,19-1,33)	0,63 (0,28-1,44)	0,93 (0,48-1,78)	1,00 (0,48-2,08)				
Social support	0,77 (0,32-1,87)	0,55 (0,25-1,19)	1,74 (0,74-4,07)	0,69 (0,33-1,45)	1,30 (0,70-2,44)	1,28 (0,64-2,57)				
Avoidance	0,97 (0,36-2,63)	1,00 (0,43-2,32)	0,93 (0,37-2,34)	2,34 (0,91-5,97)	1,53 (0,75-3,12)	0,83 (0,37-1,86)				
Palliative reacting	1,86 (0,87-3,99)	1,31 (0,70-2,48)	2,61 (1,24-5,51)	0,99 (0,52-1,85)	0,89 (0,52-1,51)	1,76 (0,94-3,29)				
			M	СР						
	n= 367/590 (62,2%)	n= 367/590 (62,2%)	n= 292/590 (49,5%)	n= 323/590 (54,7%)	n= 354/590 (60,0%)	n= 305/590 (51,7%)				
Active tackling	0,71 (0,37-1,36)	0,93 (0,57-1,54)	0,50 (0,27-0,96)	1,10 (0,67-1,79)	1,12 (0,67-1,89)	1,02 (0,55-1,90)				
Social support	0,81 (0,49-1,34)	0,64 (0,41-0,98)	1,15 (0,71-1,87)	0,71 (0,48-1,07)	0,77 (0,51-1,17)	0,60 (0,35-1,01)				
Avoidance	1,28 (0,64-2,54)	1,52 (0,87-2,65)	0,99 (0,52-1,89)	0,95 (0,56-1,60)	0,97 (0,57-1,63)	1,19 (0,63-2,28)				
Palliative reacting	2,29 (1,31-3,99)	1,35 (0,87-2,11)	1,07 (0,62-1,84)	1,54 (0,99-2,39)	1,81 (1,16-2,83)	1,22 (0,71-2,11)				
			0	NC						
	n= 204/392 (52,0%)	n= 246/392 (62,8%)	n= 187/392 (47,7%)	n= 202/392 (51,5%)	n= 252/392 (64,3%)	n= 198/392 (50,5%)				
Active tackling	0,64 (0,27-1,47)	1,26 (0,72-2,21)	1,62 (0,76-3,46)	0,69 (0,38-1,25)	0,83 (0,46-1,49)	0,98 (0,48-2,02)				
Social support	1,12 (0,50-2,48)	0,76 (0,44-1,31)	1,40 (0,71-2,74)	0,82 (0,46-1,46)	1,53 (0,86-2,73)	1,02 (0,48-2,19)				
Avoidance	0,74 (0,29-1,89)	0,90 (0,51-1,60)	1,14 (0,54-2,39)	0,91 (0,48-1,69)	1,01 (0,54-1,86)	1,18 (0,52-2,64)				
Palliative reacting	1,41 (0,72-2,76)	1,51 (0,96-2,37)	1,38 (0,74-2,55)	1,24 (0,75-2,06)	1,50 (0,93-2,42)	1,12 (0,60-2,09)				

OR (95% CI) are shown. Values with p<0,05 are in bold.

Abrreviations. OCP: older patients with cancer; MCP: middle-aged patients with cancer; ONC: older patients without cancer

Logistic regression analyses were adjusted for covariates significant in univariate analyses and for baseline well-being status

Depression was adjusted in MCP for age; in ONC for age and comorbidity

Loneliness was adjusted in OCP for living situation and marital status; in MCP for living situation; in ONC for gender and educationl level

Distress was adjusted in MCP for marital status, tumour type, and radiotherapy; in ONC for educational level and comorbidity

Fatigue was adjusted in OCP for age, gender, and tumour type; in MCP for comorbidity, tumour type, chemotherapy, and radiotherapy;

in ONC for educational level and comorbidity

ADL was adjusted in OCP for gender; in MCP for educational level, tumour type, and comorbidity; in ONC for age, gender, marital status, and educational level

IADL was adjusted in OCP for age and educational level; in MCP for tumour type, stage, surgery, chemotherapy, and radiotherapy; in ONC for age, educational level, and comorbidity

Discussion

Main findings

The results of our main analysis showed that some baseline coping strategies influenced psychological well-being and physical health one year later in patients with cancer, even after controlling for important covariates including cancer-related factors. This relation between coping and well-being differs between OCP and the two control groups. However, there were few significant associations overall, no associations were observed in ONC, and most associations were observed in MCP.

The analysis of the individual coping strategies showed that active tackling and seeking social support did not predict well-being in OCP and ONC. Avoidance coping did not predict any of the outcomes in the three patient groups. Three out of a total of five significant associations were observed for palliative reacting. It was the only coping strategy that predicted physical health, though only in MCP. Furthermore, palliative reacting was the only coping strategy that predicted physical predicted any of the outcomes (i.e. distress) in OCP.

Our additional analyses at baseline described differences in well-being and showed that active tackling was used most often in the three patient groups. Furthermore, coping strategies did not differ between older patients with and without cancer. On the other hand, coping patterns differed between the age groups in patients with cancer.

Comparison with literature and discussion

Older patients with cancer often experience less psychological morbidity compared to younger patients (5,25). One possible explanation for this difference in well-being suggests that older people are more skilled at matching coping strategies to situational demands (coping flexibility) compared to their younger counterparts (26). However, in this study we did not study younger patients (often < 50 years in studies) but MCP (50-70 years) which might explain why we did not find a difference in distress or depression between OCP and MCP at baseline.

While we did not see an ageing-related effect for psychological well-being (except for loneliness) in cancer patients, we did observe such an effect for dispositional coping at baseline. Available evidence suggests that in later life there is a decline in the use of most coping strategies in the general population (27). We also observed this decline in the current study when comparing OCP with MCP, with the exception of palliative reacting. Different explanations can be given for the age difference in coping patterns (27-29). One general interpretation suggests that older patients are more likely to devote less energy to coping when

faced with stressful situations due to psychological and social changes that often accompany ageing.

The main goal of the current paper was to assess the predictive value of baseline coping for subsequent well-being after one year in OCP while disentangling ageing and diagnosis effects. We found no diagnosis effect (OCP versus ONC) on dispositional coping at baseline and, not considering palliative reacting, there was also no diagnosis effect in the relationship between coping and well-being. Perhaps also not entirely unexpected since we did not specifically measure coping in response to cancer. Still, a cancer diagnosis in patients who generally utilize palliative reacting more often predicted distress in OCP after one year while such an association was not observed in ONC. This might reflect the difficulty of these patients to return to normal life a year after a cancer diagnosis.

The differential relationship that we observed for coping and subsequent well-being between OCP and MCP is not unexpected given the ageing effect on dispositional coping described above. Our findings indicate that the association between palliative reacting and negative outcomes seems to persist with ageing (both present in MCP and OCP) while the association between active coping (active tackling, seeking social support) and positive outcomes does not (only present in MCP). The lack of any association with avoidance coping in this study is unexpected since it is linked to decreased psychological well-being in previous cross-sectional and prospective research (6,30). With four observed associations in MCP and only one in OCP, it seems that the ageing effect attenuates the association between coping shortly after a cancer diagnosis and well-being after one year. One possible interpretation of these results suggests that older patients are more likely to trivialize a cancer diagnosis, hence the lower utilization of active coping and avoidance coping in OCP compared to MCP, which could mute the impact of the diagnosis on well-being (29). Our results are an illustration of the inappropriateness of the extrapolation of evidence obtained from younger patients to the older population. Similarly, to decisions related to the medical treatment of OCP, different aspects of ageing should be taken into account in the psychosocial care of OCP. Our observation that age does not seem to affect the association between palliative reacting and decreased psychological well-being one year after a cancer diagnosis is important for clinical purposes and is discussed below.

Strengths and limitations

Major strengths of this study include the relatively large study population, the possibility for longitudinal analyses, and the availability of two control groups for OCP which allowed us to distinguish between age-linked effects and cancer (-diagnosis) effects. Furthermore, we were

able to study simultaneously multiple dimensions of well-being. A short-coming of this study is the high percentage of patients that were lost to follow-up after one year. However, many associations were confirmed in sensitivity analyses (Appendix S1 available in the online publication). Additional analyses showed that lost to follow-up patients were not fundamentally different in terms of baseline coping compared to patients that were included in the analysis. The former utilized active tackling less frequently while no differences were observed for the other coping strategies (data not shown). Another point to consider is that although we focused on four key coping strategies in this study, other strategies that are not covered by the short UCL might provide additional information on the association with well-being.

Clinical implications and future research

A comprehensive geriatric assessment is recommended in OCP to guide treatment decisionmaking considering the variability of health status in the older population (31). This entails the assessment of multiple geriatric domains like functional status but also psychological status prior to cancer treatment. This is also a good moment to identify early on OCP with inadequate coping tendencies and offers the opportunity for prevention programs. It is already well established that distress should be assessed routinely in patients with cancer. Particularly in OCP, future intervention studies should focus on cognitive behavioural therapy aimed at palliative reacting and its impact on distress on the short- and long-term given that psychosocial interventions might only have short-term effects (32). Future research should also focus on the validity and reliability of the short version of the UCL.

Conclusions

Coping strategies influence subsequent well-being in patients with cancer, but the impact is different in the age groups. Palliative reacting was the only coping strategy that predicted well-being (i.e. distress) in OCP and is therefore, especially in this population, a target for coping skills interventions.

Author Contributions

This study was designed by AB, FB, MvdA, whereas the umbrella study KLIMOP was designed and set up by FB, MvdA, LD, and HW. Data collection was performed by AB and LD. AB, FB, TdB, MvdA drafted the manuscript, which was commented by all authors. Statistical analysis was performed by AB and discussed with FB, MvdA, and TdB. All authors read and approved the final manuscript.

Appendix 1. The short UCL (15 questions)

		Zelden of nooit	Soms	Vaak	Zeer vaak
1.	Toegeven om moeilijke situaties te vermijden (A)				
2.	U neerleggen bij de gang van zaken (PR)				
3.	Uw zorgen met iemand delen (SS)				
4.	Direct ingrijpen als er moeilijkheden zijn (AT)				
5.	Afleiding zoeken (A)				
6.	Een probleem van alle kanten bekijken (AT)				
7.	Moeilijke situaties zoveel mogelijk uit de weg gaan (PR)				
8.	Verschillende mogelijkheden bedenken om een probleem op te lossen (AT)				
9.	Doelgericht te werk gaan om een probleem op te lossen (AT)				
10.	lemand om hulp vragen (SS)				
11.	De zaken eerst op een rij zetten (AT)				
12.	Aan andere dingen denken die niet met het probleem te maken hebben (A)				
13.	Uw gevoelens tonen (SS)				
14.	Troost en begrip zoeken (SS)				
15.	Laten merken dat u ergens mee zit (SS)				

A formal translation of this version of the short UCL is not available. We added the following in the questionnaire for clarity: Active tackling (AT), seeking social support (SS), Avoidance (A), and Palliative reaction (PR).

References

1. Roesch SC, Adams L, Hines A, Palmores A, Vyas P, Tran C, et al. Coping with prostate cancer: a metaanalytic review. J Behav Medicine 2005;28(3):281-93.

2. Kvillemo P, Bränström R. Coping with breast cancer: A meta-analysis. Plos One 2014:9(11):e1127733

3. Danhauer SC, Crawford SL, Farmer DF, Avis NE. A longitudinal investigation of coping strategies and quality of life among younger women with breast cancer. J Behav Med. 2009;32(4):371-9.

4. Paek MS, Ip EH, Levine B, Avis NE. Longitudinal reciprocal relationships between quality of life and coping strategies among women with breast cancer. Ann Behav Med 2016;50(5):775-783.

5. Stanton AL, Danoff-Burg S, Huggins ME. The first year after breast cancer diagnosis: hope and coping strategies as predictors of adjustment. Psychooncology 2002;11(2):93-102.

6. Hack TF, Degner LF. Coping responses following breast cancer diagnosis predict psychological adjustment three years later. Psycho-Oncology 2004;13(4) :235-247.

7. Culver JL, Arena PL, Wimberly SR, Antoni MH, Carver CS. Coping among African-American, Hispanic, and non-Hispanic white women recently treated for early stage breast cancer. Psychology and Health 2004;19 (2):157-166.

8. Richardson AE, Morton RP, Broadbent EA. Coping strategies predict post-traumatic stress in patients with head and neck cancer. Eur Arch Otorhinolaryngol 2016;273(10):3385-91.

9 Schlegel RL, Talley AE, Molix LA, Bettencourt BA. Rural breast cancer patients, coping and depressive symptoms: a prospective comparison study. Psychol Health. 2009;24(8):933-48.

10. de Leeuw JR, de Graeff A, Ros WJ, Blijham GH, Hordijk GJ, Winnubst JA. Prediction of depressive symptomatology after treatment of head and neck cancer: the influence of pre-treatment physical and depressive symptoms, coping, and social support. Head Neck. 2000;22(8):799-807.

11. Deckx L, van Abbema D, Nelissen K, Daniels L, Stinissen P, Bulens P, et al. Study protocol of KLIMOP: a cohort study on the wellbeing of older cancer patients in Belgium and the Netherlands. BMC Public Health 2011; 11: 825. Doi:10.1186/1471-2458-11-825

12. Aarts JW, Deckx L, van Abbema DL, Tjan-Heijnen VC, van den Akker M, Buntinx F. The relation between depression, coping and health locus of control: differences between older and younger patients, with and without cancer. Psychooncology 2015;24(8):950-7

13. Deckx L, van Abbema DL, van den Akker M, van den Broeke C, van Driel M, Bulens P, et al. A cohort study on the evolution of psychosocial problems in older patients with breast or colorectal cancer: comparison with younger cancer patients and older primary care patients without cancer. BMC Geriatr 2015;15:79.

14. Sanderman R, Ormel J. De Utrechtse Coping Lijst (UCL): validiteit en betrouwbaarheid. Gedrag en Gezondheid 1992;20-7.
15. Van den Akker M, Buntinx F, Metsemakers J, Knottnerus J. Marginal impact of psychosocial factors on multimorbidity: results of an explorative nested case-control study. Soc Sci Med 2000,50:1679-1693.

16. Deckx L, van den Akker M, van Driel M, Bulens P, van Abbema D, Tjan-Heijnen V, et al. Loneliness in patients with cancer: the first year after cancer diagnosis. Psychooncology 2015;24(11):1521-8.

17. Yesavage JA, Brink TL, Rose TL, Lum O, Huang V, Adey M, Leirer VO. Development and validation of a geriatric depression screening scale: a preliminary report. Journal of psychiatric research 1982;17:37-49.

18. De Jong-Gierveld J, Kamphuis F. The development of a Rasch-Type Loneliness Scale. Applied Psychological Measurement 1985;9:289-299.

19. Bauwens S, Baillon C, Distemans W, Theuns P: The 'Distress Barometer': validation of method of combining the Distress Thermometer with a rated complaint scale. Psychooncology 2009,18:534-542.

20. Lawrence DP, Kupelnick B, Miller K, Devine D, Lau J. Evidence report on the occurrence, assessment, and treatment of fatigue in cancer patients. J Natl Cancer Inst Monogr 2004: 40-50.

21. Jean-Pierre P, Figueroa-Moseley CD, Kohli S, Fiscella K, Palesh OG, Morrow GR. Assessment of cancer-related fatigue: implications for clinical diagnosis and treatment. Oncologist 2007;12:11-21.

22. Glaus A, Assesment of fatigue in cancer patients and non-cancer patients and in healthy individuals. Support Care Cancer 1993;1:305-315.

23. Katz S, Ford AB, Moskowitz RW, Jackson BA, Jaffe MW. Studies of Illness in the Aged - the Index of ADL - A Standardized Measure of Biological and Psychosocial Function. Jama-Journal of the American Medical Association 1963;185(12):914-9.

24. Lawton MP, Brody EM. Assessment of Older People - Self-Maintaining and Instrumental Activities of Daily Living. Gerontologist 1969;9(3P1):179-186.

25. Avis NE, Deimling GT. Cancer survivorship and aging. Cancer 2008;113(12Suppl):3519-29.

26. Folkman S, Moskowits JT. Coping: pitfalls and promise. Annu Rev Psychol. 2004;55:745-74.

27. Brennan PL, Holland JM, Schutte KK, Moos RH. Coping trajectories in later life: a 20-year predictive study. Aging Ment health 2012;16(3):305-16

28. Aldwin CM, Sutton KJ, Chiara G, Spiro A. Age differences in stress, coping, and appraisal: findings from the Normative Aging Study. J Gerontol B Psychol Sci Soc Sci 1996;51(4):P179-88.

29. Blank TO, Bellizzi KM. A gerontologic perspective on cancer and aging. Cancer 2008;112(11 Suppl):2569-76.

30. Suls J, Fletcher B. The relative efficacy of avoidant and nonavoidant coping strategies: a metaanalysis. Health Psychol 1985;4(3):249-88. 31. Wildiers H, Heeren P, Puts M, Topinkova E, Janssen-Heijnen ML, Extermann M, et al. International Society of Geriatric Oncology Consensus on Geriatric Assessment in older patients with cancer. J Clin Oncol 2014;32(24):2595-603.

32. Preyde M, Synott E. Psychosocial intervention for adults with cancer: a meta-analysis. J Evid Based Soc Work. 2009;6(4):321-47.

Chapter 9. General discussion

Discussion

The implementation of geriatric assessment in oncology

Geriatric screening and geriatric recommendations

A geriatric care program was put in place in Belgium since 2007 in response to the ageing population (1). Mobile internal geriatric consultation teams were implemented to increase the availability of geriatric expertise throughout the hospital. However, their focus is primarily on hospitalized patients older than 75 years. Between 2009 and 2015, multiple Belgian hospitals implemented geriatric assessment (GA) in routine oncology practice with the financial support of the federal government. Although the implementation varied somewhat across hospitals, a GA-based model was followed that allowed screening and assessment of both inpatients and outpatients on a large scale, which is important given that the majority of patients receive cancer treatment ambulatory. The most cited barriers for implementation, according to a Belgian survey, were a high workload, lack of time or financial/staffing problems (2). However, research has shown repeatedly that GA detects unknown, possibly modifiable, health problems that could interfere with cancer treatment. Furthermore, GA influences treatment decisionmaking by facilitating better informed decisions which in a minority of patients leads to an adjustment of the cancer treatment plan (e.g. dose modifications, less aggressive treatment), primarily in patients who are considered for chemotherapy (3,4). In this regard, it is important to know that less than half of older patients with cancer are estimated to be fit (5). On the other hand, surely there are oncologists and others involved in the care of the older patient who believe they do not generally need the GA to treat this population. We believe that a better understanding of the GA-based approach will allow improving its implementation in order to optimize its effectiveness. Therefore, the evidence base for the GA-based approach should be improved. In the time between the start and the finalization of this dissertation, a lot of progress has been made in the geriatric oncology field. During this time for example, the first systematic reviews were published on two topics covered in this dissertation: the performance of geriatric screening tools (6-7) and the prognostic and predictive value of GA for clinical outcomes (8-12).

A GA is not necessary for all older patients. The need for a short and accurate screening tool to quickly identify vulnerable patients who could benefit from a GA is illustrated by the number

of publications on this subject in recent years. Several tools have been studied for this purpose. In chapter 2, we validated and directly compared the G8 screening tool and the Groningen Frailty Indicator (GFI) in reference to a GA (13). Overall, there was no difference in the diagnostic accuracy of both screening tools (AUC= 0.87 for both screening tools). However, an adequate sensitivity and negative predictive value (NPV) are the most important characteristics for a screening tool while a lower specificity can be tolerated. The G8 demonstrated an adequate sensitivity (92%) and NPV (78%), however at the expense of the specificity (52%). In contrast, the GFI had a better specificity (87%) compared to the G8, but the more important sensitivity (66%) and NPV (59%) were inadequate. In contrast to other research, we also focused on other cut-off values for the two screening tools. We suggested an alternative cut-off value (GFI score \geq 3) for the GFI corresponding with similar diagnostic characteristics than the G8. An external analysis on data of the Klimop study supported the alternative cut-off value for the GFI (unpublished data). Whilst no best screening tool can convincingly be selected based on our work and that of others, the G8 was specifically developed for older patients with cancer and is one of the most studied tools in geriatric oncology. With this tool about 30% of the screened patients are exempted from a full GA. More recently, two studies developed a modified G8, a different version in each study, that improved the diagnostic performance properties of the original tool (14,15). One study reported a more stable diagnostic accuracy across tumor sites with the modified G8 (see below). As discussed in chapter 2, an important limitation of studies evaluating screening tools is the lack of a gold standard with differences between studies in the selection of scales, cut-off values for impairment, and in the definition of an 'abnormal' GA. Furthermore, others have reported significant differences in the performance of the G8 across different tumor sites and metastatic disease status (15, 16). Although not studied, this is probably an issue for other screening tools as well. More data on the performance of screening tools are expected to follow in the coming years.

In **chapter 3**, we gained more insight on the frequency and type of geriatric recommendations following a GA in routine practice and the implementation of these recommendations (17). Our aim was to have a view on clinical practice over nine hospitals spread all over Belgium and not to study differences between hospitals. Geriatric screening was performed in 1550 patients over a time period of one year. Our study revealed that geriatric recommendations were given in about three fourths of patients. Most referrals were made to the dietician, social worker and psychologist. After one month, at least one geriatric recommendation was performed in

approximately half of these patients and about one third of all geriatric recommendations were performed. Referrals that were not often performed included the geriatric day clinic and the fall clinic. Our results indicated that there are opportunities for improvement and we discussed several possible reasons for (non) adherence to geriatric recommendations. To our knowledge, this is the first study on geriatric recommendations in the oncology setting on a large scale.

As discussed in chapter 3, the benefit of CGA and associated geriatric interventions has been shown for different clinical outcomes (e.g. functional status, institutionalization, mortality) in patients with non-malignant disease. We also mentioned that the poor implementation of geriatric recommendations has been suggested as a possible reason for the lack of CGA effectiveness in some randomized controlled trials (RCT's). This would not be different in geriatric oncology. The few RCT's in patients with cancer that focus on the impact of geriatric interventions show conflicting results (18). However, RCT's evaluating a GA-based approach in oncology similarly to that as implemented in Belgium have only recently been started (11,18,19). Our study illustrates that a learning curve should be considered in the implementation of the GA in routine oncology practice where continued efforts towards efficient collaboration, coordination, and communication are imperative. Future research, preferably RCT's, on the effectiveness of the GA-based approach in oncology should take the adherence of geriatric recommendations into account next to the experience of the participating hospitals with this approach.

The predictive value of geriatric screening tools for treatment toxicity

In chapter 2, we evaluated the diagnostic performance of two screening tools to select patients who should receive a GA. A distinction should be made between the ability of a screening tool to discriminate patients with a normal and abnormal GA and its ability to predict oncological outcomes. Several studies have demonstrated the prognostic value of geriatric screening tools for clinical outcomes in patients with cancer (6-7,20). We also demonstrated the prognostic value of the G8 and the Flemish version of the Triage Risk Screening Tool (fTRST) in univariate and multivariate analyses for overall survival in chapter 5. While there is evidence for the predictive value of screening tools for mortality, less data is available for treatment toxicity. In **chapter 4**, we evaluated the predictive value of the G8 and the GFI for treatment toxicity.

In an interim analysis, previously published as an abstract, we reported that the G8 and the GFI predict severe treatment toxicity in patients who received one cycle of (radio)chemotherapy

(21). However, our analysis with the full data revealed that both screening tools did not predict short-term treatment toxicity (22). These results need to be confirmed in a larger sample. However, other recent studies with the G8 and the GFI could also not demonstrate a predictive value for treatment toxicity (including postoperative complications). To our knowledge, no other studies have evaluated the association between the GFI and chemotherapy toxicity. However, other research could not demonstrate an association between the GFI and postoperative complications in patients with cancer (23-25). For the G8, we found one study (abstract) that reported an association with severe chemotherapy toxicity (26). However, this was not the case in a smaller study (abstract) for patients with hematological malignancies (27). An association with postoperative complications could not be shown in a recent multivariate analysis with colorectal cancer patients (28). Of note, two recent risk prediction models, containing GA components, have been developed specifically to predict chemotherapy toxicity and have received great interest but still need further external validation (29-30).

The prognostic value of geriatric information and laboratory parameters for overall survival

In chapter 5, we studied the prognostic value of GA for mortality. The predictive value of GA for treatment toxicity was beyond the scope of this dissertation (see ref 3, 10-12). One of the important questions clinicians have to consider in the treatment decision-making process is whether a particular older patient is going to die of or with cancer. For example, a patient with a limited life expectancy due to comorbidity might have no benefit from adjuvant treatment. In our study, we showed that geriatric screening and GA improved the estimation of life expectancy beyond clinical information (age, stage, tumor type) in two separate cohorts (31). Most individual GA components predicted mortality in univariate and adjusted analyses, especially nutritional status and functional status (i.e. instrumental activities of daily living). The prognostic value of the 'Leuven Oncogeriatric Frailty Score' (LOFS) was demonstrated for the first time in this study. The LOFS summarizes the results of the GA in a single score which has the advantage of simplicity towards interpretation. Furthermore, our results showed that GA summaries, i.e. the GA as a whole and the LOFS, provide more prognostic information than individual GA components. It would be reasonable to assume that GA summaries are more able to capture the multidimensional process of aging than individual GA components. However, mortality was largely explained by clinical information and the incremental value of GA was only moderate.

The prognostic value of GA for mortality is studied using different analysis techniques (8-12). Some of these analyses have not been adjusted appropriately for age, stage, or tumor type to account for the heterogeneity of the populations. This might in part explain why little consistency is found between studies as to which GA components predict mortality. Studies often select GA components and other variables that are found statistically significant in univariate analyses for inclusion in multivariate analysis. Variable selection regression methods (e.g. backward stepwise) are also sometimes used to select the best set of variables or GA components. These studies show that certain individual GA components or a subset of GA components predict mortality, but they do not evaluate the prognostic value of the GA as a whole nor do they evaluate the incremental value of the GA in addition to clinical information. Although it is important to have a view on the most prognostic GA components, information from every GA component is needed to evaluate the global health status of the patient, to identify unknown health problems that could interfere with treatment and to implement geriatric interventions as previously discussed. Therefore, we quantified the prognostic value of the GA as a whole by comparing the performance of a basic model with clinical information with a model extended with all individual GA components, which no prior study has done before. We did not perform an analysis with variable selection methods. These types of analyses are for example more appropriate for the improvement of existing prognostic calculators with additional variables to estimate life expectancy. Such online tools have been shown not to function well in older patients with cancer (32). Other studies have shown that a classification of patients, based on the results of the GA, is able to differentiate risk groups for overall survival (e.g. fit, vulnerable, frail). However, the definition of each risk group is based on consensus and different approaches are used to classify patients. A continuous GA summary score might be more precise in quantifying the patient's global health status than a classification in three risk groups. For this reason, we included the recently developed LOFS in our analysis which proved to be an interesting tool.

Since our study showed that geriatric information only moderately improved the estimation of life expectancy, further research is needed in this area. The use of biomarkers might hold promise in the assessment of prognosis beyond clinical information (33-34). **In chapter 6**, we focused on three commonly measured laboratory parameters CRP, albumin, hemoglobin and three inflammation-based composite scores that combine CRP and albumin values: the Glasgow Prognostic Score (GPS), the modified GPS (mGPS), and the CRP/albumin ratio. We performed a subgroup analysis of the study described in chapter 5 and compared the performance of biomarkers with that of the GA and evaluated if biomarkers have an added prognostic value beyond clinical and geriatric information. Our analysis showed that the GA

adds slightly more prognostic information than the laboratory parameters which are known to be strong prognostic factors (35). It also showed that the extended model with GA could be further optimized with biomarkers. To the best of our knowledge, this is the first study to compare the incremental value of different biomarkers beyond the GA as a whole.

The impact of aging and cancer on well-being: the Klimop study

The utilization of formal and informal care

The Belgian healthcare system covers almost the entire population and has developed a wide range of home care services (36). The Belgian government also provides a number of financial incentives to families to promote informal care (37). In **chapter 7**, we evaluated the use of formal and informal care shortly after a cancer diagnosis (baseline) in older patients with cancer (OCP) and two control groups: older primary care patients without cancer (ONC) and middle-aged patients with cancer (MCP) (38). At baseline, about half of older adults (OCP and ONC) and 18% of MCP used formal care, while about 85% of cancer patients (OCP and MCP) and 57% of ONC relied on informal care. Formal care increased for all groups after one year though not significantly in OCP. The amount of informal care only changed in MCP which decreased after one year.

Differences in care at baseline and changes in patterns of care after one year were further discussed in more detail. For formal care, this was done at the level of individual services. A link was made between the disease trajectory, dynamics in informal care, and feelings of loneliness. The influence of a cancer diagnosis on the relationship between formal care and informal care was discussed as well. We provided possible explanations for differences in care between the patient groups. Furthermore, we determined that cancer-related factors and associated changes in need factors explain a transition towards formal care in newly-diagnosed patients with cancer. Finally, we discussed policy implications in the approach of OCP while considering the recommended implementation of GA in routine oncology practice. This study provides a better insight into the current utilization of home care in Flanders after a cancer diagnosis and concludes that the first year after a cancer diagnosis is an important time to follow-up on the patient's needs for home care.

In this study, we focused more on formal care than on informal care given the available data. There is considerable variation in the definition of formal care in the literature. Physiotherapy is sometimes not included in this definition for example. One of the advantages of our study is that we studied the utilization of formal care in general but also individual home care services were examined. This showed for example interesting trends for physiotherapy. Although the intention of our study was not to study specific tasks performed by caregivers, we referred to certain tasks to provide possible explanations for differences in formal care between cancer and non-cancer patients (e.g. the need for physiotherapy due to lymphedema of the arm after breast cancer surgery). Increases in formal care could be explained in this way for MCP. Except for physiotherapy, these increases in care (i.e. home nursing and cleaning help) were not observed in OCP. As explained in chapter 8, the use of formal care was already high at baseline in this cohort compared to MCP and new tasks were likely covered by the already available formal care. A better understanding of these suggested 'new' tasks for OCP would require further study, however the trends that we described in MCP provide some hints.

In the Belgian context, a series of recent policy reforms aims at more patient-centered, continuous, and integrated care or 'transmural care' which entails a close collaboration between primary and specialized care providers. This will shift hospital care further towards primary care and home care organizations. In our study, we focused on the first year after a cancer diagnosis, an important time frame in the cancer trajectory. Newly diagnosed patients generally start and complete their treatment (e.g. chemotherapy, surgery, radiotherapy) during this year. The majority of these patients are treated on an outpatient basis and need to visit the clinic frequently. New forms of care are explored in recent pilot projects for patients with cancer. Several projects are evaluating the feasibility of the administration of certain cancer treatments at home (sometimes referred to as 'hospital at home') (39,40). Others are exploring approaches that would improve the chemotherapy process and service by collaborating with home care organizations. In a pilot project, prechemotherapy blood collections were performed at home instead of in the hospital avoiding extra visits to the clinic and/or long waiting times (41). Another recent pilot project developed and evaluated a transmural care trajectory for patients with glioma and their informal caregivers (42). This involved the collaboration of a coordinator of the hospital and a coordinator of the home care organization, a care model that could be extended to patients with other types of cancer.

If deemed feasible and safe, these types of initiatives have the potential to improve patient satisfaction and well-being (and that of the informal caregiver). It might also reduce the burden on busy oncology day care units and perhaps reduce costs. However, a good collaboration and coordination of all stake holders involved will be required. The advantages for patients would apply for young and old but would be perhaps even more important for OCP with a geriatric profile. If these new forms of care arrangements are implemented on a larger scale, it will

always be important to evaluate the global health and physiological reserves of the OCP for which we refer to the first part of this dissertation.

Coping and subsequent well-being

A cancer diagnosis and treatment affect the lives of patients and their environment in many ways. During the first year of a cancer diagnosis, patients have to deal with the news of the diagnosis and face many uncertainties. They might get surgery and other cancer treatments with significant toxicities besides dealing with symptoms related to cancer; they usually have to visit the clinic frequently, they might need to visit the emergency department and they might need to be admitted to the hospital; they have to undergo repeated surveillance CT-scans and deal with the fear that the treatment might not be working. Psychological distress is common in patients with cancer and it is sometimes referred to as the sixth vital sign to assess next to body temperature, blood pressure, pulse, respiratory rate, and pain (43). During this time, patients use multiple coping strategies simultaneously influencing their effects and what works effectively will depend on the context and the patients' individual characteristics (e.g. personality). The goodness of fit hypothesis is worth mentioning here. According to this hypothesis, emotion-focused coping is more adaptive for uncontrollable situations and unsolvable situations while problem-focused coping is more adaptive in controllable situations (44). In this regard, we mentioned in chapter 9 that older people might be more skilled at matching coping strategies to situational demands (coping flexibility) when discussing differences in well-being between older patients and their younger counterparts shortly after a cancer diagnosis. Situational coping should not be confused with the focus of our study: dispositional coping (45). Dispositional coping refers to coping tendencies in response to problems or unpleasant situations in general and not to specific situations (e.g. a cancer diagnosis). These are viewed as relatively stable over time and situations which allows a better comparison between different patient groups.

There is a large body of literature on coping with stressful life events, however coping strategies have not been studied often as predictor variables in longitudinal studies. In **chapter 9**, we described that there were few significant associations overall between dispositional coping and subsequent well-being (46). From the 4 coping strategies that were studied, only palliative reacting predicted subsequent well-being in OCP. No diagnosis effect (OCP vs ONC) was observed in the relation between baseline coping and subsequent well-being. However, an aging effect (OCP vs MCP) was observed. Out of 5 significant associations, 1 was observed in OCP and 4 in MCP. We discussed that our results support the notion that the use of most coping strategies declines in later life. Moreover, we suggested that older patients are more likely to

trivialize a cancer diagnosis which could mute the impact of a cancer diagnosis on well-being. Furthermore, most associations were observed with palliative reacting and this coping strategy was the only strategy that predicted well-being (i.e. distress) in OCP. Palliative reacting refers to behaviour that involves giving up any effort to deal with the situation and we concluded that this type of passive coping should be studied further in future intervention studies in OCP. In discussing the clinical implications of our results, we emphasized the need for the evaluation of psychological status as part of the geriatric assessment prior to cancer treatment as psychosocial care is an essential part of cancer treatment (47).

Future perspectives

The implementation of the GA-based approach in Belgium has made much progress over recent years. This was made possible by the national cancer plan and the financing of the federal government of several projects in geriatric oncology. Unfortunately, financing has been stopped from 2015 and this has had a negative impact on the systematic screening and assessment of older patients with cancer in clinical practice. New initiatives and efforts are needed to continue and expand the developed expertise in Belgium. Other countries have also made a lot of progress with the development of geriatric oncology programs and for example the development of teaching programs. In this section, we discuss some future perspectives on the topics covered in this dissertation.

Our research and other work could not clearly identify the most accurate geriatric screening tool to select patients for a full GA. Given results from more recent research, future research should explore the accuracy of screening tools in different tumor types. While further research is ongoing, many experts agree it is better to screen patients with an imperfect tool than not to screen patients at all.

The role of geriatric interventions following a GA in patients with cancer was emphasized in this dissertation. We gained more insight into geriatric recommendation and their implementation in routine oncology practice. This information could inform future policymaking regarding the care of the older patient with cancer. Further (qualitative) research is needed to better understand why certain referrals are performed and others are not and for example which specific actions were undertaken. The specific process of care in each hospital should be taken into account when trying to understand the GA-based approach beyond the first two steps of screening and GA. Efforts to improve the collaboration between different disciplines and efforts to improve communication will undoubtedly be important for the future. Although this was not the focus of our research, we also note that more research is needed to evaluate the impact of geriatric interventions on cancer. Future research on geriatric interventions or hospitals planning to integrate the GA-based approach in oncology for the first time should always be aware of a learning curve.

Several studies on the predictive and prognostic value of geriatric information for oncological outcomes have been published over the past recent years. More research has focused on mortality than on treatment toxicity. More research is especially needed for the latter. In accordance with our work, future research should not only focus on the question whether GA

is predictive and prognostic for clinical outcomes but also how much it adds to clinical information. In accordance with our work on routinely measured laboratory parameters, future research should also focus on the combination of GA and biomarkers, given the complexity of the processes related to aging, cancer, and their interaction. More research is warranted to assess the incremental prognostic value of other potential biomarkers (including imaging biomarkers for sarcopenia) to improve the estimation of life expectancy and other relevant clinical outcomes. In this regard, the European Organisation for Research and Treatment of Cancer (EORTC) has initiated an aging biomarker program to examine the value of potential aging biomarkers with the aim to optimize treatment decision-making and care in general in older patients with cancer.

The future of geriatric oncology in Belgium will likely be influenced by current and planned changes in the organisation and payment system of hospitals. Furthermore, primary care and home care organisations are expected to play a greater role in our future healthcare system. Our analysis based on the Klimop study provided a better insight into the current utilization of formal and informal home care by older patients shortly after a cancer diagnosis and after one year in Flanders. Future analyses from this ongoing study will be able to evaluate the use of home care at later time points. Other future research should collect more detailed information on informal care as our society is under constant change. It also remains to see how a webbased comprehensive assessment system (BelRAI) in Belgium will influence the use of home care in older patients recently diagnosed with cancer.

Finally, our analysis on coping could be a basis for future intervention studies that focus on cognitive behavioural therapy aimed at palliative reacting and its impact on distress in older patients with cancer. And although knowledge about the development of coping over the life span has increased, research on coping theory should give more attention to OCP. After examining the literature, we also suggest that future research should focus on the validity and reliability of the short version of the Utrecht Coping List.

References

1. Deschodt M, Claes V, Van Grootven B, Milisen K, Boland B, Flamaing J, et al. Comprehensive geriatric care in hospitals: the role of inpatient geriatric consultation teams. Health Services Research (HSR) Brussels: Belgian Health Care Knowledge Centre (KCE). 2015. KCE Reports 245. D/2015/10.273/48.

2. Kenis C, Heeren P, Decoster L, Van Puyvelde K, Conings G, Cornelis F, et al. A Belgian survey on geriatric assessment in oncology focusing on large-scale implementation and related barriers and facilitators. J Nutr Health Aging 2016;20(1):60-70.

3. Wildiers H, Heeren P, Puts M, Topinkova E, Janssen-Heijnen ML, Extermann M, et al. International Society of Geriatric Oncology Consensus on Geriatric Assessment in older patients with cancer. J Clin Oncol. 2014;32(24):2595-603.

4. Decoster L, Kenis C, Van Puyvelde K, Flamaing J, Conings G, De Grève J, et al. The influence of clinical assessment (including age) and geriatric assessment on treatment decisions in older patients with cancer. J Geriatr Oncol. 2013;4(3):235-41.

5. Handforth C, Clegg A, Young C, Simpkins S, Seymour MT, Selby PJ, et al. The prevalence and outcomes of frailty in older cancer patients: a systematic review. Ann Oncol. 2015;26(6):1091-101.

6. Hamaker ME, Jonker JM, de Rooij SE, Vos AG, Smorenburg fCH, van Munster BC. Frailty screening methods for predicting outcome of a comprehensive geriatric assessment in elderly patients with cancer: a systematic review. Lancet Oncol. 2012;13(10).

7. Decoster L, Van Puyvelde K, Mohile S et al. Screening tools for multidimensional health problems warranting a geriatric assessment in older cancer patients: an update on SIOG recommendations. Ann Oncol. 2015;26(2):288-300.

8. Puts et al. Use of geriatric assessment for older adults in the oncology setting: a systematic review. J Natl Cancer Inst. 2012:104(15):1134-1164.

9. Hamaker et al. The value of geriatric assessment in predicting treatment tolerance and all-cause mortality in older patients with cancer. Oncologist 2012;17(11):1439-1449.

10. Puts et al. An update on a systematic review on the use of geriatric assessment for older adults in oncology. Ann Oncol. 2014 (2):307-315

11. Caillet et al. Optimal management of elderly cancer patients: usefulness of the comprehensive geriatric assessment. Clin Interv Aging 2014;9:1645-1660.

12. Versteeg KS, Konings IR, Lagaay AM, van den Loosdrecht AA, Verheul HM. Prediction of treatment-related toxicity and outcome with geriatric assessment in elderly patients with solid malignancies treated with chemotherapy: a systematic review. Ann Oncol. 2014;25(10):1914-8.

13. Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D. Evaluation of the Groningen Frailty Indicator and the G8 questionnaire as screening tools for frailty in older patients with cancer. J Geriatr Oncol. 2013; 4(1):32-38.

14. Martinez-Tapia C, Canoui-Poitrine F, Bastuji-Garin S, Soubeyran P, Mathoulin-Pelissier S, Tournigand C, et al. Optimizing the G8 screening tool for older patients with cancer: diagnostic performance and validation of a six-item version. Oncologist 2016;21(2):188-95.

15. Petit-Monéger A, Rainfray M, Soubeyran P, Bellera CA, Mathoulin-Pélissier S. Detection of frailty in elderly cancer patients: improvement of the GA screening test. J Geriatric Oncol. 2016;7(2):99-107.

16. Liuu E, Canouï-Poitrine F, Tournigand C et al. Accuracy of the G-8 geriatric-oncology screening tool for identifying vulnerable elderly patients with cancer according to tumour site: The ELCAPA-02 study. J Geriatr Oncol. 2014;5(1):11–19.

17. Baitar A, Kenis C, Moor R, Decoster L, Luce S, Bron D, et al. Implementation of geriatric assessment-based recommendations in older patients with cancer: a multicenter prospective study. J Geriatr Oncol. 2015;6(5):401-10.

18. Soubeyran P, Terret C, Bellera C, Bonnetain F, Jean OS, Galvin A, et al. Role of geriatric intervention in the treatment of older patients with cancer: rationale and design of a phase III multicenter trial. BMC Cancer 2016,16(1):932.

19. Brugel L, Laurent M, Caillet P, Radenne A, Duran-Zaleski I, Martin M, et al. Impact of comprehensive geriatric assessment on survival, function, and nutritional status in elderly patients with head and neck cancer: protocol for a multicentre randomised controlled trial (EGeSOR). BMC Cancer 2014;14:427.

20. Martinez-Tapia C, Paillaud E, Liuu E, Tornigand C, Ibrahim R, Fossey-Diaz V, et al. Prognostic value of the G8 and modified-G8 screening tools for multidimensional health problems in older patients with cancer. Eur J Cancer 2017;83:211-219.

21. Van Fraeyenhove F, Baitar A, De Vos M, Galdermans D, De Droogh E, D'hooghe A, et al. Prediction of chemotherapy toxicity by the Groningen Frailty index (GFI) and the comprehensive geriatric assessment (CGA) in elderly cancer patients (pts): an interim analysis. Ann Oncol. 2010;21(suppl 8):viii185-viii188 (abstr 568 PD).

22. Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D. Geriatric screening results and the association with severe treatment toxicity after the first cycle of (radio)chemotherapy J Geriatr Oncol. 2014;5(2):179-84.

23. Papis D, Lembo R, O'Reilly D, Ammori B, Sherlock D, Deshpande R, et al. The predictive ability of timed "Up & Go", VES-13 and GFI in hepato-pancreato-biliary onco-geriatric surgical patients. J Geriatr Oncol. 2012; volume 3, Supplement 1, S100-S101 (abstr P130)

24. Huisman MG, Van Leeuwen BL, Ugolini G, Montroni I, Stabilini C, Vigano A, et al. Predicting outcome in onco-geriatric surgical patients: screening tools versus the comprehensive geriatric assessment. EJC. 2013;49,S337-338.

25. Bras L, Peters TT, Wedman J, Plaat BE, Witjes MJ, van Leeuwen BL, et al. Predictive value of the Groningen Frailty Indicator for treatment outcomes in elderly patients after head and neck, or skin cancer surgery in a retrospective cohort. Clin Otolaryngol. 2015,40(5):474-82.

26. Fagard K, Casaer J, Wolthuis A, Flamaing J, Milisen K, Lobelle JP, et al. Value of geriatric screening and assessment in predicting postoperative complications in patients older than 70 years undergoing surgery for colorectal cancer. J Geriatr Oncol. 2017;8(5):320-327.

27. Stokoe JM, Pearce J, Sinha R, Ring A. G8 and VES-13 scores predict chemotherapy toxicity in older patients with cancer. J Geriatr Oncol. 2012; volume 3, Suppl 1, S81 (abstr P95)

28. Dubruille S, Bron D, Roos M, Vandenbossche S, Meuleman N, Maerevoet M, et al. The respective usefulness of the G8 and a comprehensive geriatric assessment (CGA) to predict intolerance to chemotherapy and survival of fit and vulnerable older patients with hematological malignancies. J Geriatr Oncol. 2013; volume 4, suppl 1 S56 (abstr P073)

29. Hurria A, Togawa K, Mohile SG, Owusu C, Klepin HD, Gross CP, et al. Predicting chemotherapy toxicity in older adults with cancer: a prospective multicenter study. J Clin Oncol. 2011;29(25):3457-65.

30. Extermann M, Boler I, Reich RR, Lyman GH, Brown RH, Defelice J, et al. Predicting the risk of chemotherapy toxicity in older patients: The Chemotherapy Risk Assessment Scale for High-Age Patients (CRASH) score. Cancer 2012;118(13):3377-86.

31. Kenis C*, Baitar A*, Decoster L, De Grève J, Lobelle JP, Flamaing J, et al. Geriatric screening and geriatric assessment add independent prognostic information to a basic model with clinical data. Cancer. Accepted for publication.

32. de Glas NA, van de Water W, Engelhardt EG, Bastiaannet E, de Craen AJ, Kroep JR, et al. Validity of Adjuvant! Online program in older patients with breast cancer: a population-based study. Lancet Oncol. 2014;15(7):722-9.

33. Pallis AG, Hatse S, Brouwers B, et al. Evaluating the physiological reserves of older patients with cancer: the value of potential biomarkers of aging? J Geriatr Oncol. 2014;5(2):204-18.

34. Hubbard JM, Cohen HJ, Muss HB. Incorporating biomarkers into cancer and aging research. J Clin Oncol. 2014;32(24):2611-6.

35. Baitar A*, Kenis C*, Decoster L, De Grève J, Lobelle JP, Flamaing J, et al. The prognostic value of three commonly measured blood parameters and geriatric assessment to predict overall survival in addition to clinical information in older patients with cancer. Cancer. Accepted for publication.

36. Arnaert A, Heuvel BV, Windey T. Health and social care policy for elderly in Belgium. Geriatr Nurs. 2005;26(6):366-71.

37. www.allesoverkanker.be

38. Baitar A, Buntinx F, De Burghgraeve T, Deckx L, Bulens P, Wildiers H, et al. The utilization of formal and informal home care by older patients with cancer: a Belgian cohort study with two control groups. BMC Health Serv Res. 2017;17(1):644

39. Farfan-Portet M-I, Denis A, Mergaert L, Daue F, Mistiaen P, Gerkens S. Implementation of hospital at home: orientations for Belgium. Health Services Research (HSR) Brussels: Belgian Health Care Knowledge Centre (KCE). 2015. KCE Reports 250. D/2015/10.273/68.

40. Schrijvers D, Van Asch L, Van Loon H, Sarens S. Transmurale zorg voor oncologische patiënten. Tijdsch Gen 2017;73:1527-1530. 41. Schrijvers D, Van Asch L, Van Goethem A, Heylen E, De Schouwer P, Van Eeckhout S, et al. Bloedname in de thuiszorg: een pilootproject in de oncologie. In preparation.

42. Transmuraal zorgtraject voor patiënten met een glioma en hun mantelzorgers. Uitbouw en evaluatie 2014-2016.<u>www.komoptegenkanker.be/transmurale-zorg-hersentumorpatienten-en-</u> mantelzorgers

43. Vitek L, Rosenzweig MQ, Stollings S. Distress in patients with cancer: definition, assessment, and suggested interventions. Clin J Oncol Nurs. 2007;11(3):413-8.

44. Howell D, Olsen K. Distress - the 6th vital sign. Current oncology 2011;18(5):208-210.

Folkman S, Moskowits JT. Coping: pitfalls and promise. Annu Rev Psychol. 2004;55:745-74.

45. Carver CS, Scheier MF. Situational coping and coping dispositions in a stressful transaction. J Pers Soc Psychol. 1994;66(1):184-95.

46. Baitar A, Buntinx F, De Burghgraeve T, Deckx L, Schrijvers D, Wildiers H, et al. The influence of coping strategies on subsequent well-being in older patients with cancer: a comparison with two control groups. Psychooncology. 2017;1-7.

47. Jacobsen PB, Holland JC, Steensma DP. Caring for the whole patient: the science of psychosocial care. J Clin Oncol. 2012;30(11):1151-3.

Chapter 10. Summary (English & Nederlands)

Summary

The number of older patients with cancer will further increase due to the increasing life expectancy and ageing population. Health care providers and policy makers should be aware that older patients with cancer require special attention regarding treatment decisions and care. Treatment in older patients with cancer is complex due to the heterogeneity in global health status, physiological reserve capacity and the lack of evidence-based data for this population. In this dissertation, we focus on some clinical, societal, and psychosocial aspects of ageing and cancer.

International guidelines recommend the implementation of geriatric assessment (GA) in routine oncology practice to detect previously unknown health problems, implement geriatric interventions, improve the estimation of life expectancy and the risk of treatment toxicity. In other words, the aim of GA is to guide treatment decision-making. Multiple Belgian hospitals implemented GA in routine oncology practice between 2009 and 2015 with the financial support of the federal government. Given that the majority of patients with cancer are treated ambulatory, a GA-based model was followed that allowed the approach of both inpatients and outpatients. Based on the data from these implementation studies, we validated two geriatric recommendations following a GA. A better understanding of the GA-based approach will allow improving its implementation in order to optimize its effectiveness. Our study on the implementation of geriatric recommendations indicated that there are opportunities for improvement and we discussed several possible reasons for (non) adherence to these recommendations.

In order to help physicians guide treatment decision-making, geriatric screening and GA should ideally have a predictive and prognostic value for oncological outcomes. We could not demonstrate a predictive value of two geriatric screening tools for short-term treatment toxicity in patients who received (radio)chemotherapy. On the other hand, we showed that geriatric screening and GA improved the estimation of life expectancy beyond clinical information, though the improvement was moderate. Especially nutritional status and functional status (i.e. instrumental activities of daily living) add more prognostic information compared to the other GA components. However, our results showed that GA summaries, i.e. the GA as a whole and a recently developed tool summarizing the results of a GA in a single score (the LOFS), provide

more prognostic information than individual GA components. GA summaries are more able to capture the multidimensional process of ageing. In a separate analysis, we showed that GA as a whole adds slightly more prognostic information to clinical information than known prognostic biomarkers (e.g. albumin, CRP, Glasgow Prognostic Score). Furthermore, we showed that with routinely measured laboratory parameters life expectancy is estimated more precisely than with only clinical and geriatric information.

The second part of this dissertation comprises of two analyses based on the Klimop study, an ongoing longitudinal cohort study on the impact of ageing and cancer on well-being. In a first analysis, we provided a better insight into the current utilization of formal and informal care in Flanders shortly after a cancer diagnosis and showed that the first year after a cancer diagnosis is an important time to follow-up on the patient's needs for home care. We expect to see some significant changes in the utilization of home care by older patients with cancer in the coming years in light of recent policy reforms with a focus on more transmural care.

In the second analysis, we showed that ageing influences the association between coping strategies shortly after a cancer diagnosis and subsequent well-being in patients with cancer. Different aspects of ageing should be taken into account in the psychosocial care of older patients with cancer. Based on our results, we suggested that this population is more likely to trivialize a cancer diagnosis which could mute the impact of the diagnosis on well-being. Palliative reacting, which involves giving up any effort to deal with the situation, was the only coping strategy that predicted well-being (i.e. distress) in older patients with cancer after one year. This type of passive coping should be a target for future intervention studies, especially in this population.

Samenvatting

Het aantal oudere patiënten met kanker zal verder toenemen als gevolg van de stijgende levensverwachting en de vergrijzing van de bevolking. Zorgverleners en beleidsmakers moeten zich ervan bewust zijn dat oudere patiënten met kanker speciale aandacht nodig hebben bij het nemen van beslissingen rond hun behandeling en hun zorg in het algemeen. De behandeling van oudere patiënten met kanker is complex vanwege de heterogeniteit in hun globale gezondheidsstatus, fysiologische reserve capaciteit en het gebrek aan evidence-based data voor deze populatie. In dit proefschrift richten we ons op enkele klinische, maatschappelijke en psychosociale aspecten van veroudering en kanker.

Internationale richtlijnen bevelen aan om een geriatrische 'assessment' (GA) te implementeren in de dagelijkse oncologie praktijk om eerder onbekende gezondheidsproblemen te detecteren, geriatrische interventies te implementeren, de levensverwachting van de patiënt beter te kunnen inschatten en om beter het risico op toxiciteit ten gevolge van een kankerbehandeling te kunnen inschatten. Kortom, de GA dient als leidraad bij het nemen van beslissingen over de behandeling van de oudere patiënt. Met de financiële steun van de federale overheid, hebben meerdere Belgische ziekenhuizen een GA geïmplementeerd in de dagelijkse oncologie praktijk tussen 2009 en 2015. Aangezien de meerderheid van de patiënten met kanker ambulant wordt behandeld, werd een model gevolgd waarbij zowel ambulante als gehospitaliseerde patiënten konden bereikt worden. Op basis van de gegevens van deze implementatiestudies, valideerden we twee geriatrische screening tools om patiënten te selecteren voor GA in een tweestap benadering en bestudeerden we geriatrische aanbevelingen die volgden na een GA. Meer kennis rond deze aanpak in de oncologie zal het mogelijk maken om de implementatie van GA te verbeteren en om de effectiviteit ervan te optimaliseren. Ons onderzoek naar de uitvoering van geriatrische aanbevelingen wees uit dat er mogelijkheden voor verbetering zijn en we bespraken verschillende mogelijke redenen voor het al dan niet naleven van deze aanbevelingen.

Geriatrische screening en GA zouden idealiter een voorspellende en prognostische waarde moeten hebben voor oncologische uitkomsten opdat de GA als leidraad kan dienen bij de behandeling van de oudere patiënt. We konden geen voorspellende waarde aantonen van twee geriatrische screeningsinstrumenten voor toxiciteit op korte termijn bij patiënten die (radio) chemotherapie kregen. Anderzijds toonden we aan dat geriatrische screening en GA de inschatting van de levensverwachting verbeterden ten opzichte van een inschatting met alleen klinische informatie. Deze verbetering was echter matig. Vooral voedingsstatus en functionele status (instrumentele activiteiten van het dagelijks leven) voegen meer prognostische informatie toe in vergelijking met de andere GA componenten. Onze resultaten toonden echter aan dat 'GA-samenvattingen', hier de GA als geheel en een recent ontwikkeld instrument dat de resultaten van een GA in een enkele score (de LOFS) samenvat, meer prognostische informatie verschaffen dan individuele GA componenten. GA-samenvattingen zijn beter in staat om het multidimensionale proces van veroudering vast te leggen. In een afzonderlijke analyse hebben we aangetoond dat de GA als geheel iets meer prognostische informatie toevoegt aan klinische informatie dan gekende prognostische biomerkers (e.g. albumine, CRP, Glasgow Prognostic Score). Bovendien toonden we aan dat met routine laboratoriumparameters de levensverwachting nauwkeuriger wordt ingeschat dan met alleen klinische en geriatrische informatie.

Het tweede deel van dit proefschrift bestaat uit twee analyses op basis van de Klimop-studie, een lopend longitudinaal cohortonderzoek naar de invloed van veroudering en kanker op welzijn. In een eerste analyse hebben we het huidige gebruik van formele en informele zorg in Vlaanderen kort na een diagnose van kanker beschreven en hebben we aangetoond dat het eerste jaar na een diagnose van kanker een belangrijke periode is om de nood voor thuiszorg op te volgen. We verwachten de komende jaren een aantal belangrijke veranderingen rond thuiszorg bij oudere patiënten met kanker gezien de recente beleidshervormingen die zich meer richten op transmurale zorg.

In de tweede analyse hebben we aangetoond dat veroudering een invloed heeft op de associatie tussen coping strategieën kort na een diagnose van kanker en welzijn na 1 jaar. Bij de psychosociale zorg van oudere patiënten met kanker moet rekening worden gehouden met verschillende aspecten van veroudering. Op basis van onze resultaten suggereerden we dat de kans groter is dat oudere patiënten een diagnose van kanker zullen trivialiseren. Dit zou de impact van de diagnose op het welzijn kunnen beperken. 'Palliatief reageren', wat inhoudt dat elke poging om de situatie aan te pakken werd opgegeven, was de enige coping strategie die welzijn (meer bepaald distress) na één jaar bij oudere patiënten met kanker voorspelde. Vooral in deze populatie, is deze vorm van passieve coping een doelwit voor toekomstige interventie studies.

Scientífic acknowledgement

For the authors contributions we refer to the end of each chapter.

Personal contribution

Study 1

Geriatric assessment (GA) was implemented in ZNA Middelheim and Jessa ziekenhuis between October 2009 and December 2011. ZNA Middelheim coordinated this study. Chapter 2 and Chapter 4 were based on this study.

The doctoral candidate tasks included the coordination of the study, literature study, geriatric screening and GA of patients in ZNA Middelheim, data management and analysis, and manuscript writing.

Study 2

GA was implemented in nine hospitals across Belgium between August 2011 and July 2012. 'De universitaire ziekenhuizen Leuven' coordinated this project. Chapter 3 and Chapter 5 (cohort B) were based on this study.

The doctoral candidate tasks included the discussion of the analysis plan, literature study, data management and analysis, and manuscript writing.

Study 3

GA was implemented in 'de universitaire ziekenhuizen Leuven' and in 'het universitair ziekenhuis Brussel' in Belgium between August 2009 and July 2011. 'De universitaire ziekenhuizen Leuven' coordinated this project. Chapter 5 (cohort A) and chapter 6 were based on this study.

The doctoral candidate tasks included the discussion of the analysis plan, literature study, data management and analysis, and manuscript writing.

Study 4

The Klimop study is an ongoing longitudinal cohort study in Belgium and the Netherlands. The department of public health and primary care of the KU Leuven coordinated this study in Belgium. Chapter 7 and chapter 8 were based on this study.

The doctoral candidate tasks included the discussion of the analysis plan, literature study, interviewing patients at home during follow-up visits, data analysis, and manuscript writing.

Conflict of interest statement

The authors indicated no potential conflicts of interest.

Professional career

Abdelbari Baitar received his Masters in Biomedical sciences at the K.U. Leuven in 2006. After working about 2 years in a chemistry laboratory, he started as a part-time researcher in ZNA Middelheim in Antwerp. There he mainly worked on a geriatric oncology project between 2009-2012 (National Cancer plan action 24). His responsibilities included the coordination of the project, which was in collaboration with Virga Jesse, Hasselt. A similar project was at that time ongoing in UZ Leuven which lead to the start of his PhD in 2012 at the K.U. Leuven. The project in ZNA Middelheim was extended between 2012-2015. The extended project, including 22 participating hospitals, was coordinated by UZ Leuven. Meanwhile, he also joined the Klimop study linked to the department of public health and primary care, K.U. Leuven. He continued to work part-time in the department of oncology in ZNA Middelheim after the finalization of these projects.

While doing his work focused on cancer, he also worked from 2009 until 2018 part-time for vzw Farmaka, an evidence-based medicine organization. For their academic detailing project financed by the government, he visited and informed primary care physicians by presenting three different topics a year with the aim to promote the rational use of drugs. From 2018, he started to work as a part-time scientific collaborator for 'the BCFI' (the Belgian Centre for Pharmacotherapeutic Information), a non-profit, non-governmental organization. The BCFI provides independent information on all drugs prescribed in Belgium and promotes rational prescribing of drugs.

Publications: Full Articles

Quinten C, Kenis C, Decoster L, Debruyne PR, De Groof I, Focan C,...,Baitar A et al. Determining clinically important changes in health-related quality of life in unfit elderly patients undergoing chemotherapy or surgery as cancer treatment. Under review.

Baitar A*, Kenis C*, Decoster L, De Grève J, Lobelle JP, Flamaing J, et al. The prognostic value of three commonly measured blood parameters and geriatric assessment to predict overall survival in addition to clinical information in older patients with cancer. Cancer. Accepted for publication

Kenis C*, Baitar A*, Decoster L, De Grève J, Lobelle JP, Flamaing J, et al. The added value of geriatric screening and assessment to predict overall survival in older patients with cancer. Cancer. Accepted for publication

Baitar A, Buntinx F, De Burghgraeve T, Deckx L, Schrijvers D, Wildiers H, et al. The influence of coping strategies on subsequent well-being in older patients with cancer: a comparison with two control groups. Psychooncology. 2017;1-7.

Baitar A, Buntinx F, De Burghgraeve T, Deckx L, Bulens, Wildiers H, et al. The utilization of formal and informal home care by older patients with cancer: a Belgian cohort study with two control groups. BMC Health Serv Res. 2017;17(1):644.

Kenis C, Heeren P, Decoster L, Van Puyvelde K, Conings G, Cornelis F,...,Baitar A, et al. A Belgian survey on geriatric assessment in oncology focusing on large-scale implementation and related barriers and facilitators. J Nutr Health Aging 2016;20(1):60-70.

Baitar A, Kenis C, Moor R, Decoster L, Luce S, Bron D, et al. Implementation of geriatric assessment - based recommendations in older patients with cancer: a multicenter prospective study. J Geriatr Oncol 2015;6(5):401-10.

Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D. Geriatric screening results and the association with severe treatment toxicity after the first cycle of (radio)chemotherapy J Geriatr Oncol 2014;5(2):179-84.

Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D. Evaluation of the Groningen Frailty Indicator and the G8 questionnaire as screening tools for frailty in older patients with cancer. J Geriatr Oncol 2013; 4(1):32-38.

Baitar A., Deckers L, Messens E., Van Fraeyenhove F, Schrijvers D. Fall prevention in a Palliative Care Unit: evaluation of an infrared sensor device. EJPC 2012;19(6):285-287.

Baitar A, De Vos M, Vandebroek A, Schrijvers D. Carboplatin and gemcitabine in patients with advanced and/or metastatic urothelial cancers: a phase II study. J Geriatr Oncol 2011;2(1):31-35.

Casteels C, Lauwers E, Baitar A, Bormans G, Baekelandt V, Van Laere K. In vivo type 1 cannabinoid receptor mapping in the 6-hydroxydopamine lesion rat model of Parkinson's disease. Brain Res. 2010;1316:153-62.

Publications: Abstracts

Kenis C*, Baitar A*, Decoster L, De Grève J, Lobelle JP, Flamaing J, et al. Geriatric screening and geriatric assessment add independent prognostic information for survival to basic clinical data in older patients with cancer. J Geriatr Oncol. 2017; volume 8, Supplement 1, S81

Decoster L, Kenis C, Flamaing J, DeBruyne P, DeGroof I, et al. Adherence to geriatric assessment (GA)-based recommendations in older patients (pts) with cancer. Journal of Clinical Oncology 2017;35, 15_suppl.

Baitar A, Van Fraeyenhove F, Vandebroek A, De Droogh E, Galdermans D, Mebis J, Schrijvers D. Will 'fit' older patients with cancer as assessed by frailty screening tools tolerate the first cycle of (radio)chemotherapy without serious adverse events? Ann Oncol (2012) 23 (suppl 9): ix447-ix461. [Abstr 1386]

Baitar A, Deckers L, Messens E, Schrijvers D. Fall prevention in a Palliative Care Unit: the use of bed exit alarms in practice. Palliat Med 2012;384-674 (P45).

Baitar A, Vandebroek A, De Droogh E, Mebis J, Schrijvers D. Evaluation of the G8 questionnaire as a screening tool for frailty in older cancer patients. EJC 2011 Vol. 47 Suppl 1, page S282 [abstr 4006] (Oral presentation, 2011 ECCO/ESMO congress)

Baitar A, Van Fraeyenhove F, Galdermans D, Mebis J, Schrijvers D. Evaluation of the Groningen Frailty Indicator as a screening tool for frailty in older cancer patients. EJC 2011 Vol. 47 Suppl 1, page S275 [Abstr 4031]

Van Fraeyenhove F, Baitar A, De Vos M, Galdermans D, De Droogh E, D'Hooghe A, Schrijvers D. Prediction of chemotherapy toxicity by the Groningen Frailty Index (GFI) and the Comprehensive Geriatric Assessment (CGA) in elderly cancer patients (pts): an interim analysis. Ann Oncol (2010) 21 (suppl 8): viii185-viii188. [abstr 568PD]

Schrijvers D, Baitar A, De Vos M, Mebis J, Vroman P. Evaluation of the Groningen Frailty Index (GFI) as a screening tool in elderly patients (pts) : an interim analysis. Ann Oncol (2010) 21 (suppl 8): viii185-viii188. [abstr 566PD]

Baitar A, De Vos M, Vandebroek A, Schrijvers S. Carboplatin and gemcitabine in elderly and/or frail patients with advanced transitional cell carcinoma (TCC): a phase II study. Ann Oncol (2010) 21 (suppl 8): viii271-viii303

Studies I worked on as a data manager/study coordinator in ZNA Middelheim:

SPECTA: Screening cancer patients for efficient clinical trial access (EORTC-1553)

A Phase 1b/2, Open-Label, Multicentre Study Assessing the Safety, Tolerability, Pharmacokinetics, and Preliminary Anti-tumor Activity of MEDI4736 in Combination With AZD9150 or AZD5069 in Patients With Advanced Solid Malignancies and Subsequently Comparing AZD9150 and AZD5069 Both as Monotherapy and in Combination With MEDI4736 as Second Line Treatment in Patients With Recurrent and/or Metastatic Squamous Cell Carcinoma of the Head and Neck.

BT-414 alone or ABT-414 plus temozolomide versus lomustine or temozolomide for recurrent glioblastoma: a randomized phase II study of the EORTC brain tumor group.

A phase 3 randomized, placebo-controlled trial of carboplatin and paclitaxel with or without veliparib (ABT-888) in HER2-negative metastatic or locally advanced unresectable BRCA-associated breast cancer.

A Phase 3 Randomized, Multicenter, Placebo-controlled Double-blind Study of JNJ-56021927 in Combination with Abiraterone Acetate and Prednisone Versus Abiraterone Acetate and Prednisone in Subjects with Chemotherapy-naive Metastatic Castration-resistant Prostate Cancer (mCRPC).

A phase III, open-label, multicenter, randomized study to investigate the efficacy and safety of MPDL3280A (anti-PD-L1 antibody) compared with chemotherapy in pts with locally advanced or metastatic urothelial bladder cancer after failure with platinum-containing chemotherapy.

Reassure study: Radium-223 alpha emitter agent safety study in mCRPC population for long-term evaluation.

A phase II study of IV vinvlunine in combination with methotrexate versus methotrexate alone in pts with recurrent or metastatic squamous cell carcinoma of the head and neck previously treated with platinum-based chemotherapy.

Real life observations on Innohep thrombosis management in oncology patients: an observational study of treatment of DVT and/or PE in cancer patients with Innohep.

Observational study to obtain information on the efficacy and safety of Erbitux (cetuximab) in combination with standard chemotherapy in daily clinical practice in Belgium in 1st line treatment of mCRC patients with KRAS wild-type tumor.

Prospective non-interventional multicenter observational phase IV trial to describe first line anti-cancer treatment in patients with metastatic renal cell cancer in Belgium.

Prospective non-interventional study to collect data on the use of Avastin and conventional chemotherapy for the treatment of previously untreated metastatic colorectal cancer in pts \geq 70 years.

European survey of oncology patient's experience of breakthrough pain.

Observational study to obtain information on the safety and efficacy of Erbitux (cetuximab) in daily clinical practice in Belgium in 2e line treatment of mCRC patients with KRAS wild-type tumor.

Academic detailing for vzw Farmaka (www.farmaka.be)

Topics presented to family physicians individually and in group from the year 2009 until 2018:

2009: Hulp bij rookstop

2010: Aanpak acute COPD exacerbatie; Aanpak van neuropathische pijn; Vitaminen en mineralen

- 2011: Insomnia; Hormonale anticonceptie I; Urine incontinentie
- 2012: Hormonale anticonceptie II; Majeure depressie; Osteoporose
- 2013: Aanpak van Artrose; Polyfarmacie I; Hypercholesterolemie in primaire preventie
- 2014: Preventie van thrombo-embolie bij VKF; Polyfarmacie II; Acute hoest bij kinderen
- 2015: Angststoornissen; Diabetes type II; Nierinsufficiëntie
- 2016: Inhalatiemedicatie bij COPD; CV preventie bij diabetes; Bariatrische heelkunde-opvolging
- 2017: Langdurig PPI gebruik; Probleemgedrag bij dementie; Antibiotica bij hoest?